**IMISCOE Research Series** 

# Mastoureh Fathi Caitríona Ní Laoire

# Migration and Home IMISCOE Short Reader







**IMISCOE Research Series** 

This series is the official book series of IMISCOE, the largest network of excellence on migration and diversity in the world. It comprises publications which present empirical and theoretical research on different aspects of international migration. The authors are all specialists, and the publications a rich source of information for researchers and others involved in international migration studies. The series is published under the editorial supervision of the IMISCOE Editorial Committee which includes leading scholars from all over Europe. The series, which contains more than eighty titles already, is internationally peer reviewed which ensures that the book published in this series continue to present excellent academic standards and scholarly quality. Most of the books are available open access. Mastoureh Fathi • Caitríona Ní Laoire

# Migration and Home

**IMISCOE** Short Reader



Mastoureh Fathi Department of Sociology and Criminology University College Cork Cork, Ireland Caitríona Ní Laoire School of Applied Social Studies University College Cork Cork, Ireland



ISSN 2364-4087 ISSN 2364-4095 (electronic) IMISCOE Research Series ISBN 978-3-031-51314-5 ISBN 978-3-031-51315-2 (eBook) https://doi.org/10.1007/978-3-031-51315-2

© The Editor(s) (if applicable) and The Author(s) 2024. This book is an open access publication.

**Open Access** This book is licensed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license and indicate if changes were made.

The images or other third party material in this book are included in the book's Creative Commons license, unless indicated otherwise in a credit line to the material. If material is not included in the book's Creative Commons license and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder.

The use of general descriptive names, registered names, trademarks, service marks, etc. in this publication does not imply, even in the absence of a specific statement, that such names are exempt from the relevant protective laws and regulations and therefore free for general use.

The publisher, the authors, and the editors are safe to assume that the advice and information in this book are believed to be true and accurate at the date of publication. Neither the publisher nor the authors or the editors give a warranty, expressed or implied, with respect to the material contained herein or for any errors or omissions that may have been made. The publisher remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

This Springer imprint is published by the registered company Springer Nature Switzerland AG The registered company address is: Gewerbestrasse 11, 6330 Cham, Switzerland

Paper in this product is recyclable.

## Contents

1	Conceptualising Home in Migration: An Introduction				
	1.1	Introduction	1		
	1.2	Key Tenets of Home in Migration	3		
	1.3	Structural Im/Possibilities: A Theoretical Framework			
		for Understanding Home in Migration	7		
		1.3.1 Intersectionality, Identity and Home	8		
		1.3.2 Migration/Citizenship Regimes and Bordering	10		
		1.3.3 Home as Intersectional and Bordered	12		
	1.4	Positionality, Reflexivity and Our Homes-in-Migration			
		Biographies	14		
	1.5	Overview of the Book	15		
	Refe	erences	17		
2	Gender and Home in Migration				
	2.1	Introduction	21		
	2.2	Feminist Thinking on Home	22		
	2.3	Women's Mobilities and Home/Land	24		
	2.4	Spatial Boundaries and Gender in Migration.			
	2.5	Migrant Women, Home-Making and the Domestic Sphere			
	2.6	Changing Gender Relations and Masculinities			
		in Migrant Home-Making	30		
	2.7	Gender and Future Homes in Migration	34		
	2.8	Conclusion	35		
	Refe	References			
3	Age and Home in Migration				
	3.1	Introduction	41		
	3.2	Conceptualising Age, Generation and Life-Course	42		
	3.3	Ageing and Home in Migration	43		
		3.3.1 Temporality and Home-Making Among			
		Older Migrants	44		

		3.3.2	Deepening Home Roots in Place Among	
			Older Migrants	45
		3.3.3	Transnational Home and Older Migrants.	47
		3.3.4	Attachments to 'Homeland' in Older Age	48
		3.3.5	Summary: Structural and Intersectional	
			Im/Possibilities of Home for Older Migrants	50
	3.4	Home	e, Migration and Young Adulthood	51
		3.4.1	Youth, Mobility and Home	51
		3.4.2	Young Adulthood and 'Leaving Home'	52
		3.4.3	Young Migrants and Home in Mobility	54
		3.4.4	Summary	57
	3.5	Childl	hood, Migration and Home	58
		3.5.1	Children and Home-Making in Migration:	
			Agency and the Everyday	58
		3.5.2	Multi-scalarity and the Politics of Home	
			in Migrant Childhoods	60
		3.5.3	Summary	62
	3.6	Concl	lusions	62
	Refe	erences		63
4	Mad		as of Home in Mismetian	69
4	4.1		es of Home in Migration	69 69
			luction	
	4.2		rtance of Objects in Migrants' Lives	70
	4.3		tts, Memories and Nostalgia	71
	4.4		mentality and Everyday Life	75
	4.5		ts and Social Identity	78
	4.6		lusion	80
	Refe	erences		81
5	Mig	ration	Status, Class, Race and Home	85
	5.1	Introd	luction	85
	5.2	Privile	ege and Home in Migration	86
		5.2.1	Feeling at Home in the Host Society as Privilege	88
		5.2.2	Temporary/Mobile Home in Privileged Migration	89
		5.2.3	Privileged Migration and Homeland	91
		5.2.4	Summary	92
	5.3	Preca	rity and Home	93
		5.3.1	Refugee Accommodation and the Structural	
			Impossibilities of Home	93
		5.3.2	Protracted Displacement and Precarity	95
		5.3.3	Intersectionality, Selectivity and Everyday	
			Bordering of Home	96
		5.3.4	Making Home in Displacement and Precarity	98
		5.3.5	Summary	99
	5.4	Concl	lusion	99
	Refe	erences		101

	Conclusion			
	6.1	Introduction	105	
	6.2	Home in Migration at a Glance	106	
	Reference			

## **Chapter 1 Conceptualising Home in Migration: An Introduction**



#### 1.1 Introduction

Migration and home-making are both integral to the history of human settlement. In order to secure livelihoods, find new partners, form families, educate ourselves, flee from war, violence, climate or freedom of expression, as humans, the quest to find another place in which to live is as old as human settlement itself. However, the extent to which the multiple locations in which we live can be referred to as *home* varies and is context-dependent. Home can be regarded as a core part of the human condition. Indeed, according to the United Nations, shelter, alongside adequate nutrition and education, is considered a basic human need (King, 1998). However, we know that home is about much more than simply a place of shelter: 'home' encompasses deeply personal feelings about the places and spaces of the intimate spheres of our lives.

As such, home is a significance source of identification for migrants. Mobility does not erode processes of identifying with or attaching to place (Cuba & Hummon, 1993) and in fact can bring questions of 'home' into sharper focus than in contexts of immobility. In an era of unprecedented global mobility, the idea of 'home' has particular resonance in debates surrounding questions of migration and border-crossing. Public and political concerns have crystallized around the figure of the migrant, imagined as a person 'out of place', triggering international policy concerns about migration management, increased securitisation of borders, as well as rising xenophobia in many contexts, reflecting anxieties that are, at their core, about 'who belongs where' and 'who *can* belong where'. At the same time, migrants and diasporic communities develop individual and collective strategies for remembering past homelands, shaping new homes or making home in mobility itself. In other words, migration triggers questions, narratives and discourses about belonging/notbelonging in and through places and spaces, provoking reflections on the notion of home and what it means.

Hence the idea of 'home' is a powerful, yet contested, one in scholarship on migration. Geographers, sociologists and anthropologists have grappled with this idea of 'home', with its deeply emotional, yet ambiguous, resonance for migrants and those close to them (Ahmed et al., 2003; Blunt & Dowling, 2006; Boccagni, 2017; O'Connor & Crowley-Henry, 2020; Pink, 2004; Ralph & Staeheli, 2011). Migration highlights the complexities inherent in the idea of home as a place or space of belonging in contexts of mobility and border-crossing where home can be a place that is lost or mourned, that is constantly being made, that is imagined, or that is found in the in-between spaces of here and there (Ahmed et al., 2003; Brah, 1996). Most migration scholarship on home views it as a phenomenon that has a much wider and deeper meaning than simply a domestic place of dwelling (Boccagni, 2023; Mallett, 2004; Fox, 2016; Miranda Nieto et al., 2021) and instead looks at structure, biographical significance, and feelings associated with home in contexts of migration (Mitchell, 1971; Myers, 1989; Cuba & Hummon, 1993; Altman & Werner, 1985; Ahmed et al., 2003; Mallett, 2004; Fox, 2016). Home, as a space that is associated with strong emotions (Svašek, 2012), is not just a space to which one resorts from the outside world; it becomes a site from which people understand their place in the world.

As such, home has been associated with a range of (mostly) positive feelings and experiences of belonging, such as security, familiarity, control, comfort, familial and caring relationships, intimacy, place embeddedness and hope (Ahmed, 1999; Boccagni, 2017; Boccagni & Duyvendak, 2019; Hage, 1997; Ralph & Staeheli, 2011). For example, home is usually associated with meaningful relationships such as those with one's children, spouse/partner, parents, siblings, extended family, flatmates, friends, neighbours and wider (imagined) communities. However, it is important to recognise that sites deemed to be 'home' can feel unhomely and be associated with negative and ambiguous feelings, as 'home' is always saturated with power relations (Ahmed, 1999; Brickell, 2012). Discussions revolving around ideas such as non-home (Boccagni & Miranda Nieto, 2022), critical geographies of marginal home (Brickell, 2012) and unhome (Fathi, 2022a), refer to a range of negative feelings that challenge the positivity so often associated with the concept of home. Sometimes, notions of homeliness or home-likeness are more relevant than the idea of a singular complete home. The idea of feelings of homeliness can capture the ambiguities of home by recognising how homes can be more or less home-like, shifting with changing circumstances. In this way, the idea of 'a home' can be distinguished from the more complex and ambiguous qualities associated with 'making home' (Fathi, 2021). In other words, it is recognised that home is in some ways an ideal, something that is worked towards, rather than a fixed place.

However, the vast and rich social-scientific literature that exists on home and migration from various perspectives needs a thorough theorisation that could be of use for scholars across several disciplines. In this book, we delve into the existing scholarship on home and migration, in doing so drawing on our own experiences as migration researchers, to present this Short Reader. However, we do so through a theoretical lens that incorporates the concepts of intersectionality and structural im/ possibility in order to ground migrant homing and home-in-migration in debates on power, privilege and precarity.

In this opening chapter, first we outline some of the key conceptual principles, or tenets, that have emerged from contemporary social-scientific literature on home and migration, then moving on to build on these key principles to present an intersectional and structural conceptual framework for reading migration-and-home scholarship. In the final part of the chapter, we reflect on the implications of our own home-in-migration biographies and positions for our reading of this literature and present an overview of the structure of the rest of the book.

#### **1.2** Key Tenets of Home in Migration

We present here some key tenets of the existing social-scientific scholarship on home in migration – some conceptual principles around which much of the literature has evolved in recent decades and which provide underpinning core principles to the subsequent discussions in the rest of this book.

- (a) *Home in migration as multi-scalar:* Home is an inherently spatial phenomenon as it reflects how feelings of belonging/not-belonging are experienced in relation to particular places and spaces. While conventionally 'home' might be understood simply in terms of a domestic dwelling space, or as a series of nested spatial belongings (for example, domestic space - local community nation), contemporary scholarship questions this neat characterisation and highlights the complexities of multi-scalarity in constructions of home in migration (Blunt & Varley, 2004; Staeheli & Nagel, 2006). Migrant home, it is argued, can span or traverse household, community, city and nation (Blunt & Dowling, 2006) and can encompass spaces from the domestic to transnational spheres (Al-Ali & Koser, 2003; Pérez Murcia & Boccagni, 2022). In other words, home is often experienced and imagined at scales and frames of reference beyond the domestic. However, as Miranda Nieto (2021, p. 17) points out, 'scales are not concrete spaces, places or locations, but ways of framing social activity'. For example, discussions of how urban space can be experienced as home (or unhome) are not just about the space of a city; they are also about the meaningful activities taking place within that space that associate it with home (Fathi & Ní Laoire, 2023). Similar arguments can be made about nation or homeland (see for example, Nash, 2003).
- (b) Home as multilocational: Furthermore, much literature points to the possibilities of home as a multi-locational, multi-focal, scattered or even 'stretched' phenomenon, that is, not tied to one fixed location but understood more as a sphere of meaningful connections and attachments at different scales and involving different places (Brettell, 2006; Ralph & Staeheli, 2011). Thus, home is on one hand about emplacement, localisation and attachments to places, and on the other about connections, relations and attachments between places

(Ahmed et al., 2003). Indeed, in this way, home is understood as intimately bound up with mobility and border-crossings, or, as being shaped through the experience of migration, but also in the sense that home itself is always changing and characterised by porous boundaries (Ahmed, 1999; Ahmed et al., 2003).

(c) Home in migration as process: As a counterpoint to traditional understandings of home as a fixed and bounded place that one either leaves behind or stays in, current thinking, located in a more constructivist ontology, emphasises the processual, fluid and dynamic nature of home (Boccagni, 2017; Ralph & Staeheli, 2011). Home is understood through a range of concepts such as homing desire, homing and home-making, denoting home as a process encapsulating those practices, performances, desires and acts through which home is lived, felt and done (Ahmed et al., 2003; Boccagni, 2017; Brah, 1996; Tolia-Kelly, 2004; Walsh, 2011). While the three concepts have certain aspects in common, they are not exactly the same.

The concept of homing desire is usually attributed to the work of Black feminist writer, Avtar Brah, for whom the desire for home is inherent to diasporic identities, and which she distinguishes from the more essentialist idea of desire for a homeland (Brah, 1996). This idea of homing desire suggests an 'almost out of reach' quality to home, acknowledging the ways in which 'homes' (domestic dwellings, homelands) have not always been places of safety, comfort and belonging for those on the margins and challenging nativist and essentialist constructions of place and belonging. Brah (1996, p. 192) makes a distinction between feeling at home and claiming home, recognising that the idea of home is integrally bound up with 'the social regulation of belonging', wherein for some, the feeling and claiming of home overlap easily and for others, they are out of step. Hence, embedded in the idea of homing desire is a rejection of the idea of fixed origins and recognition of the multi-locality and contingency of home in diaspora (Brah, 1996).

More recently, from a sociological perspective, Boccagni (2017) has developed the concept of homing which similarly captures the processual and unfinished nature of home. He uses the concept to denote a lifelong process of 'attaching a sense of home to [...] life circumstances'. In Boccagni's (2017, p. 26) theorisation of the term:

Homing [...] is a range of *spatialized social practices through which migrants* – as exemplary of people who went through extended detachment from their earlier homes – *try to reproduce, reconstruct and possibly rebuild meaningful home-like settings, feelings and relationships.* (italics in the original)

In this sense, homing is viewed as an individual-level biographical process, though with a political aspect, involving an ongoing search for home, and with normative, emotional and practical dimensions, in which home in migration is more an ideal than a reality.

Finally, the concept of home-making emphasises that home is made up of practices of inhabitance at the micro level. There is not a fixed thing that is called home. Home is composed of repetitive, routine, quotidian practices, performances and acts that are done, with or without a purpose of making a home, but that function as makers of a home. This is what can be called home-making – a process that through habitual acts, produces particular meanings, places and identities (Muñoz, 2018). The concept of home-making has been adopted widely in recent years in the home and migration literature as a valuable way of connecting the micro-level, material and everyday practices of doing home to wider social and cultural processes (Boccagni, 2014; Meijering & Lager, 2014; Walsh, 2011; Wilkins, 2019).

These processual aspects of home, as homing desire, homing and homemaking, together point to three further important observations: (1) That home is a *yearning for something that is out of reach*. Even when we think we have achieved home, there is always something that reminds us that the feeling of comfort or safety is only temporary. (2) Home lies in the *constant and ongoing attempts* towards those desires for home. It is shaped within the desire for home and through the efforts and practices of working towards making home. (3) Finally, the processual aspect of home shows us that achieving home is a *process of the imagination* in that it comes into being through our longings and desires. The spaces and places of our lives become home through the meanings that we give them. These insights remind us that home is an ongoing process that is constantly under construction through everyday practices as well as enmeshed in longer-term projects of desire.

(d) Home and the boundaries between outside and inside: Home is understood as an oft insecure phenomenon, incorporating strangeness and difference within it (Ahmed, 1999). The stranger is always in proximity, and as Ahmed (1999, pp. 21–22) argues:

Strangers are not simply those who are not known in this dwelling, but those who are, in their very proximity, already recognized as not belonging, as being out of place. Such a recognition of those who are out of place allows both the demarcation and enforcement of the boundaries of 'this place', as where 'we' dwell.

What Sara Ahmed means here is that by recognising the stranger as a character who does not belong to this place, we are facilitating the (re)production of social, political and geographical boundaries that demarcate places as homely or unhomely. The one who belongs may find the former to be valid whilst the alien, the stranger and the one who is placed outside of these boundaries, is the one who feels the latter. In fact, the idea of home is often deployed in popular or political discourses of 'community' and 'nation' that seek to draw boundaries between those who belong and those who do not. Thus, home is integrally bound up with the politics of bordering at different scales.

(e) Home as temporal: Understood in the sense of an ongoing process, home can be seen as composed of a set of practices and feelings that tie the past, present and future together. Home in migration is shaped in relation to memories of the past, practices in the present and hopes for the future (Boccagni, 2017; Fathi, 2021). Migrations can result in ruptures in life histories, separating migrants from their familiar social worlds, from a feeling of home that is in the past. For some, this is experienced as loss, but home may be re-created in the present through remembering, maintaining transnational ties or carrying elements of past homes into the present. For others, who might have felt like strangers in their past homes, home can be re-constituted through migration, creating a new home in the present or working towards a home in the future (Fortier, 2003). The temporality of home is also bound up with biographical temporality, whereby certain life stages are normatively associated with aspects of home-making/homing, such as leaving a parental home or setting up a new home. However, the ruptures produced by migration can complicate this association and challenge homing expectations (Fathi, 2022b; Sirriyeh, 2016).

- (f) *Home as everyday experience*. If home is not just about a house, a geographical location, a country or nation, then what is it? Much of the recent research focuses on the everyday as a valuable lens through which to grasp less visible aspects of homing in migration (Boccagni, 2017; Hatfield, 2010; Fathi & Ní Laoire, 2023). Removed from its fixed elements, home is about the everyday lived and felt experience. Home is what we do on a daily basis to feel embedded in our personal lives and the social world. What we do make sense to us. It is done with an aim and it serves a purpose. This could be mundane and not reflected upon, such as turning on the kettle, cleaning the toilet or watering the plants, or playing, but these are important in how we make connections to the worlds around us. So, when we talk about home as experience, in a way, we are talking about the sense of fulfilment we achieve through placing ourselves within the dynamics of our home space (whatever that is). For example, one important aspect of everyday experiences of home-making is that of personalisation, which is addressed in Chap. 4, which looks at how personal touches can be added to even the most unhomely environments.
- (g) Home as sensorial. Our last tenet of home refers to embodied experiences. Senses are important in the construction of the space we call home (Hamilton, 2017; Mata-Codesal, 2023). The phrase 'feeling at home' is directly related to this, that after all, home is about how our bodily senses, such as smell, colour or sounds, relate the dynamics of a space to our core sense of self (Fathi, 2021). Our emotions about a place of home are always entangled in what it reminds us of (past), what is meaningful to us (present) and what new opportunities of connection it will bring (future). Thus, embodied and sensorial elements of home are also connected to temporality. Placing body at the heart of discussions on home in migration is also about how bodily presence in social and public spaces can evoke feelings of homeliness or unhomeliness. In our earlier study, we found that young migrant men in Cork, Ireland, who were mostly visibly different to the Irish white population, were aware of their bodily presence in public and felt less at home compared to when they were in their domestic spaces (Fathi & Ní Laoire, 2023). Not feeling at home in public spaces can be directly tied to experiences of racialisation and exclusion (Lloyd & Vasta, 2017), which points to one of many ways in which home is bound up with wider social structures and processes.

## **1.3** Structural Im/Possibilities: A Theoretical Framework for Understanding Home in Migration

The above seven tenets of home in migration are well-recognised in the existing scholarship. However, we suggest that a stronger social and political framing is needed to contextualise home within the wider power relations and structures that shape where, when and for whom home in migration is more or less possible. There is an understandable tendency for studies on home and migration to focus on the small scale, intimate and personal level at which home is experienced. Many of these valuable studies (which we discuss throughout this book) do reveal the ways in which these small-scale experiences are underpinned by, and constituted through, wider structural and political processes of migration and settlement. However, there is a need for a coherent and comprehensive theoretical framework of home in migration that recognises this structural and political underpinning and puts power at its heart. Here inspiration can be taken from scholars of home in migration such as Ahmed et al. (2003), Blunt and Dowling (2006) and Brun and Fábos (2015), who have made significant contributions in this respect. Drawing on existing scholarship and wider literature, we propose here a conceptual framework for understanding home in migration that focuses on what we call the structural possibilities and impossibilities of home.

In conceptualising how migrants imagine and do home/homing, it is important to acknowledge the wider structural contexts in which desires and imaginations are formed and the possibilities of home are opened up or closed off. Such an approach would integrate an understanding of the everyday lived and felt aspects of home-inmigration with a critical analysis of the structural contexts in which 'living and feeling home' is made possible. Home is inherently about the politics of belonging and bordering (Mitzen 2018) - about who can belong where and the role of boundary construction, maintenance and contestation in lived experiences of home and homing (Brah, 1996; Yuval-Davis, 2006; Yuval-Davis et al., 2018). As Moore (2000) argues, home is as much about exclusion as inclusion. While home-in-migration is about the everyday and personal lived experiences of home-making and homing, it cannot be separated from questions about the resources available to migrants with which to do home. As Ralph and Staeheli (2011, p. 520) point out, it is necessary to 'be attuned to the different ways in which migrants, with different resources and different social locations, negotiate the extensibility and fixity of home'. They suggest that it is likely that migrants in different legal and social positions, with different levels of material or cultural resources, will imagine and 'do' home differently.

We understand this to have two key co-constitutive dimensions. First, social location/position is about the *intersectionality* of experiences of home, recognising that home is experienced differently by people in different social positions. Home is gendered, classed, racialised and marked by axes of social difference and power. Secondly, *transnational migration regimes* play a key role in how migrants imagine and do home, not only in terms of the influence of laws and regulations but also as mediated by economic and political regimes of migration. Together these two

dimensions structure the possibilities and impossibilities of home in migration – via the boundaries and borders of belonging and the access to resources with which to make and imagine home. If home is a site of safety, security and belonging, then it is essential to think about it in relation to the power relations that circumscribe and delineate who can feel safe, secure and that they belong, and where and how these feelings are more or less possible. We refer to this as the *structural im/possibilities of home in migration*.

These two dimensions of intersectionality and migration regime cannot be easily disentangled from one another as migration regimes are themselves inherently racialised, gendered and classed (as well as intersecting with other categories that for the sake of concision, are not mentioned here). However, we foreground these as two core concepts that together aid an understanding of the social and political nature of home in migration. The following sections detail how each concept can contribute to understanding the structural im/possibilities of home in migration.

#### 1.3.1 Intersectionality, Identity and Home

Home is 'situated' and should be analysed with an awareness of, and identification with, the interrelated dynamics of gendered, classed, aged and racialised relations and systems (Crenshaw, 1989, 1991). Home as a social and individual process is always underpinned by circuits of power and networks of people. We argue that the ways in which migrants understand home and identify with a home are always linked with the intersectionality of their identities: gender, social class, the method/ route of migration, their migration status, their language skills, sexuality and age among others. As such, home is intersectional.

Intersectionality has been taken as 'the most important theoretical contribution that women's studies, in conjunction with other fields, has made so far' (McCall, 2005, p. 1771). Indeed, intersectional analysis can be traced to multiple movements oriented to equality, human rights, feminism, and development (Cho et al., 2013). Given its history, intersectionality is not a unified body of theory, but it has spanned different disciplines that are concerned with multiple axes of power relations and how systems of oppression and marginalisation reinforce and constitute each other (Crenshaw, 1991). Each of the social categories, although related to each other, ontologically is independent and has distinctive meanings. However, they are co-constitutive, that is, their meanings are always constructed in relation to others in their ontological bases.

It is this aspect of intersectionality (the co-constitution of systems of power and marginalisation) that makes it a useful tool to understand how home-making is rooted in differential positionings across axes of power, that is, how some people are able to make a home and other are not. Whilst categories of class, gender, race, age, sexuality, ability and others are important for analysing the degree to which a migrant feels at home, intersectionality allows for an understanding of how the underlying power relations intersect each other to constitute one another and once different systems of marginalisation are compounded, they impact individuals' abilities and chances to feel at home, belonged and wanted in a society. For example, intersectionality reveals how racism constitutes classism (and vice versa) and how patriarchy operates hand-in-hand with racism and ableism. In other words, by focusing on power relations that define and value certain individuals, groups and ways of living and thinking (Yuval-Davis, 2006), we can see how one form of homemaking in migration is valued and enabled over another.

Intersectionality then offers an understanding of knowledge as situated, which challenges the positivist view of knowledge which sees everything from the position of nowhere (Stoetzler & Yuval-Davis, 2002). Yuval-Davis (2015, p. 95) argues that:

Situated intersectionality analysis, therefore, in all its facets, is highly sensitive to the geographical, social and temporal locations of the particular individual or collective social actors examined by it, contested, shifting and multiple as they usually are.

So in this view of intersectionality, we are always positioned along an axis of power, and have a standpoint that we do not share with others. For example, as an Iranian migrant woman, or as an Irish return-migrant woman, we share some but not all understandings of home with other people with the same intersectional identity categories. While intersectionality and migration regimes structure the possibilities of migrant home, it is imagination that gives lived experiences their particular meaning. We understand who we are and we make sense of the world through the power of imagination. Imaginations are the bedrock of identities and practices and play an important role in how home is experienced and its meanings are developed. Indeed, everyone has an 'ideal home' in mind that for the present time is only achieved through imagining it. But the extent to which this ideal is understandable by others comes from the situated positioning of the person. The concept of 'situated imagination' is useful here (Stoetzler & Yuval-Davis, 2002). It recognises that no one imagines from a position of nowhere. We are always situated in how we understand the world around us and this situation has a determining effect in how we construct imagination about self and others. According to Stoetzler and Yuval-Davis (2002), imagination is the corporeal, impulsive and creative force within us that makes sense of lived experience but is always grounded in intersectional social positioning. Through our situated imagination, we construct an image of an 'ideal' (possible) home and a 'non-home' (as the two extremes) and direct our attempts towards achieving the former and distancing ourselves from the latter, that is, we imagine home and act on this basis in doing home.

These perspectives are beneficial for thinking about how contemporary migrants understand and imagine home (in ways that span temporal and spatial imaginaries), what home means to them, and how they put this understanding into practice (at multiple geographical scales from the domestic and local to the translocal). For example, how do feelings of at-homeness or out-of-homeness articulate emotionally and materially in migrants' lives? And how do ideas about past, present and future home/s form part of migrants' imaginings and how these imaginings are articulated and put into practice in their everyday lives? These are questions about home-making processes and homing desires. In order to understand home in migration intersectionally, we must consider how a person's social location, which is intersectionally experienced, shapes the possibilities and impossibilities of home creation, maintenance and feelings. For example, home is inherently a gendered phenomenon and may be experienced very differently by migrant men and migrant women. Additionally, social class, which intersects closely with migrant status, shapes material possibilities of home-making in terms of access to housing, ability to establish comfortable surroundings and lifestyles and ease of transnational mobility. In the rest of this book, we tease out these structural im/possibilities of home in migration and we draw particular attention to the ways in which different axes of power intersect with each other.

#### 1.3.2 Migration/Citizenship Regimes and Bordering

The concept of migration regime is increasingly being discussed and adopted in migration studies (see Cvajner et al., 2018, for an overview). While the concept is understood in a number of different ways, with different theoretical inspirations (Horvath et al., 2017), the term is used here to refer to a constellation of regulatory influences, practices and power relations that impact on the formation of migration and migration-related social processes in any particular migration context (Bernt, 2019; Horvath et al., 2017; Williams, 2012). These influences can include immigration and citizenship policies, regulations, histories of migration, discourses of migration, labour market dynamics, care cultures and family cultures in societies of destination and/or origin (Amelina, 2017; Lutz, 2017; Williams, 2012). This understanding of migration regime recognises the role of (powerful) non-state actors, such as business interests, in shaping migration processes, and also acknowledges that while the nation-state is a very powerful player in migration regimes, many constellations of influence also operate across national borders.

Regimes of citizenship are closely intertwined with migration regimes. Erdal et al. (2018) argue that citizenship (understood in a formal legal sense) matters for migrants and non-migrants in relation to feeling secure and recognised in society. In effect, citizenship, as Yuval-Davis (2006) argues, is a method of governance that regulates 'the politics of belonging' in a given nation state. The politics of belonging is regulated through a series of policies used to govern and control populations, demarcating boundaries between majority and minorities and defining the boundaries of who belongs and who does not. Erdal and Sagmo (2017) argue that citizenship, as a vertical relationship between the state and individual, grants membership of this collectivity in a way that allows members to define themselves in relation to their belonging to it. We argue in this book that citizenship and the right to stay in a host country for a migrant is an important part of feeling at home. It is not the only aspect, but is structurally very important for the foundations of making a home. The legal right to stay is a key part of what we call *spatial security*, or the feeling of secure inhabitance in a place, which comes about when the right to stay is

accompanied by meaningful place attachment (Fathi & Ní Laoire, 2023). Spatial security is an essential prerequisite for home in migration.

Structures such as international and national migration and citizenship policies as well as the dynamics of global labour markets and globalised imaginaries produce different migration regimes, which differentially facilitate or restrict migration and settlement, or possibilities of spatial security, according to migrants' social positions. These structures include forces such as the demand for cheap labour in the global North as well as the politics of immigration and legal frameworks that define who can move where and who belongs where. These forces determine material aspects of home-making of migrants, such as their access to housing, work and social security. Beyond the immediate material aspects, how might the macro-level structures of global migration make it possible, or impossible, at the micro-level, to feel at home, to make home and to imagine home? For example, to what extent are European migrants in EU member states made to feel like migrants compared to migrants from Yemen, Algeria or Afghanistan? While some of the bordering mechanisms at work are subtle, others are less so. In very obvious ways, asylum regimes in many destination countries deliberately seek to restrict migrants' possibilities of home-making (Howlett-Southgate, 2021). Migration policy is replete with the symbolism of home/not-home, evident for example in the language of 'home affairs' and 'homeland security' that reflects a view of the state and its territory as a 'home' to be protected (Walters, 2004) or diaspora policies that seek to encourage return migration of those deemed to be 'coming home' (for example, the Irish government's #hometowork campaign in 2015-2016). The powerful forces that shape regimes of home and migration also encompass the role of media, for example, in re-producing constructions of home, host and newcomer, as one element of the apparatus of bordering that works to define belonging/not-belonging in contemporary migration and citizenship regimes. For example, the use of ambiguous language in inflating migrant identity and the 'skilled or non-skilled' distinction has been a tactic used in showing the UK as not a homely place for migrants in general in the UK government's Brexit campaign (Parnell, 2023).

In the twenty-first century, we are witnessing regimes of bordering becoming ever more present in our lives, as everyday bordering introduces rationalities and regulation of inclusion/exclusion into spheres of everyday life (Yuval-Davis et al., 2018). This means that structures of im/possibility can permeate everyday life very quickly and easily, limiting the possibilities of feeling at home. Recent immigration and refugee policies (such as Dublin Regulation III, 2017, new deportation practices and relocation of borders to outside the EU borders) are central to migration regimes within the European contexts. These policies pose a potential reconfiguration of what constitutes home as part of a broader shift towards migration/citizenship regimes that are more complex than in the past. The growth in 'proliferation of migrant statuses' that means that immigrants may find themselves constantly moving in and out of more-or-less 'legal' or documented situations over time, points to the increased fragility and temporariness of immigrant statuses (Gonzales & Sigona, 2017). As a result, migrants may become immobile in migration for long periods of time, stuck in legal/policy limbos and separated from family members, unable to return to a past home, move on to a new home or build a home in the place where one is stuck.

This situation can be contrasted with the EU policy agenda of 'free movement' which allows EU citizens<sup>2</sup> to live and work in other EU countries with little restriction or regulation (though in ways that are contingent – see Parker & Catalán, 2014), thus distinguishing between EU and non-EU citizens in terms of their structural possibilities of home in migration. Lulle et al. (2018) refer to the privilege of EU citizenship in enabling fluid mobilities among EU citizens in pre-Brexit UK. Correspondingly, the rupture of Brexit, in removing the right to free movement between the UK and the EU, has been experienced as an affective event by EU citizens living in the UK, threatening their future stability (Lulle et al., 2018) and disrupting their dynamics of belonging (Ranta & Nancheva, 2019). Similarly, Miller (2019) refers to the 'unsettling' effect of Brexit on the sense of home among older British migrants living in Spain, which she relates to the uncertainties and questions raised by Brexit about where home is now and where it will be in the future. What this research on Brexit highlights is the unsettling effects of shifting borders of citizenship, but more importantly, the emotional effects of such changes, which can be related to how people feel about where they live and have made home/s (Guma & Dafydd Jones, 2019; Lulle et al., 2018; Miller, 2019; Tyrell et al., 2019). Thus migration/citizenship regimes not only shape the im/possibilities of home through rights of residence and citizenship but also through how they make people feel about where they live.

#### 1.3.3 Home as Intersectional and Bordered

Therefore it is important to pay attention to the migration regimes within which migrants' lives and movements are regulated – the laws, policies, labour markets, citizenship regulations, institutions and borders that prescribe and proscribe the possibilities of emplacement, belonging, autonomy and social connection – in short, that frame the socio-political-economic context in which home is imagined and practiced. However, it is important to emphasise here that migration regimes, as underlying power relations that regulate migrants' lives, do not in themselves make or unmake home, but it is only as places and spaces are endowed with meaning by migrants that they can become homely or home-like. Emotions can be seen as a 'constitutive part of transnational family experience' (Skrbiš, 2008, p. 236), part of the belonging process (Yuval-Davis, 2006) but also, importantly, linked to imaginations about what home may look like in future. We argue that home and emotional attachments in migration need to be understood intersectionally in order to tease out the multiple experiences of migration within which people feel excluded, included, privileged or marginalised simultaneously.

It is also important to recognise that imaginaries of home are bound up with wider discourses and norms relating to gender, social class, ethnicity or sexuality. Gray (2000) shows how gendered discourses of nation and family shaped the

contours of the landscape within which Irish migrant women in England in the 1980s navigated expectations and im/possibilities of home and belonging. Therefore, it is important to pay attention to the wider collective discourses that migrants engage with, such as the nostalgic discourses that construct myths of return to a remembered homeland (Christou 2006), or the discourses of cosmopolitanism that drive mobilities, or the gendered heterosexual norms that dominate imaginaries of home. Thus, home as a site of belonging is about personal feelings and attachments to spaces, people and objects, although at the same time these feelings are embedded in wider collective discourses and regimes of inclusion and exclusion. Home is integrally tied up with the personal and intimate lived experiences of relationality, emotion and materiality (dimensions discussed throughout this book) while being permeated by power relations. Therefore, migrant homing and home-making can be viewed as negotiated and practiced in the context of intersectional power relations and powerful transnational migration regimes. This raises a key question that is at the heart of this book:

How are intersectionalities and migration regimes understood, felt, experienced, lived and navigated by migrants, who are differently positioned, in various contexts, in the making and imagining of home?

Foregrounding this question illuminates the role of intersectional power relations at both macro and micro levels in shaping the possibilities of everyday homing practices and of dreams of imagined home. Thus, home in migration can be explored through the lens of migrants' own imaginings and practices of home and homing. However, this analysis can be deepened by paying particular attention to the power relations that shape the possibilities of imagination and practice and the ways in which migrants navigate these possibilities. Such epistemology of home allows us to think critically about how mobilities, settlement, practices and ruptures are shaped in current debates on home and migration, focusing more on intersectional elements of home and how migration regimes limit and expand the boundaries of these imaginations and practices.

Studies of home in migration draw our attention to everyday acts of homemaking, practices to belong, feelings and emotional attachments to places, the daily routines and ruptures of life as a migrant. The role of mobility as well as fixity, of past as well as present and future, in making sense of these practices and feelings, is emphasised. However, it is important to acknowledge the ways in which wider power relations make those practices and feelings possible, understandable, reproduceable, and transmittable. Intersectional power relations frame emerging, continuing, and reiterating imaginations of home. This is the space in which conflict between movement and settlement can be teased out and that shapes the meaning of home as, for example, a gendered identity, as a memory of the past, as a generational practice, as practices of the present, or as a racialised space. An analytical framework of imaginations interconnected with im/possibilities allows us to examine the role played by rules and regulations, policies and polities, discourses and memories, in migrants' different forms of identity construction, practices and expressions of feelings in relation to home. Such an approach would foreground migrants' own narratives and imaginations of home, homing and home-making and, importantly, would set these in the context of structural possibilities that frame and condition imagination. Looking at home through this framework, this book provides a reading of existing scholarship that seeks to disentangle multiple layers of power relations that are in place when we discuss home and home-making in migration, presenting a uniquely intersectional and transnational theoretical perspective on migrant homing and home-in-migration.

#### 1.4 Positionality, Reflexivity and Our Homes-in-Migration Biographies

We started our conversation about writing this book as an endeavour to discuss our own understandings of home, in Cork, Ireland, as two women working in academia, one a sociologist, the other a social geographer by background. One of us, Fathi, is a woman of colour, a new migrant to Ireland, and the other, Ní Laoire, is a returned Irish migrant. Although in principle, our lines of thinking are very similar in relation to questions of belonging and the politics of belonging, and how Cork and in general Ireland, as a new destination country, is positioned in our theorisation of migration, we realised that we come to these questions from different situated understandings. We thought it important to lay out our situated positioning here to set in context some of the decisions we have made in relation to what to include in this book and what to leave out.

Our biographical histories, coming from two different worlds, Iran and Ireland, and both subsequently living as migrants (though with very different migrant statuses) in Britain, inform quite a lot of our political understanding of migration. Whilst Fathi, now a British citizen, came to Ireland with rights of entry, living, even voting when she settled in her new home, feelings of estrangement from the surroundings marked her experiences of home-making in Cork. Contrasting to this, on Ní Laoire's much earlier return migration to Ireland from Britain, she found that even though she had to re-settle, find new accommodation, navigate different administrative systems, and so on, doors were quite suddenly opened to her after initial feelings of being a complete newcomer, once people realised she was 'one of them'.

More recently, we were collaborating on a project called Youth Home (2019–2020) on young male migrants' home-making practices in Cork. Our extended conversations about what it feels like to interview young (male) migrants about home and being a (female) migrant or resident, shaped much of this understanding: home is intersectional, it is temporal, and it is bound by spatial and legal power relations. Fathi wrote in one of her fieldnotes:

I tend to forget that, I, after all have the right to live in Ireland, for an unlimited time due to my Britishness (second citizenship), that puts me on a different scale of migrancy compared to my participants who are refugees and international students on short term visas.

Amongst us, we also experienced Cork differently in relation to our biographical histories. Both of us had arrived from somewhere else, settled in Cork and started working in the university space. Our understanding of Irish society however was shaped by our personal histories of connections to the place, one having a foreign name and accent, one an Irish name and accent, one being seen as a woman of colour, the other a white Irish woman. Ní Laoire's emotional and biographical connection to Cork was rooted in a personal history of growing up in 1970s and 1980s rural county Cork, with strong family connections and memories of being a university student in the city who went 'home-home'1 at weekends. Fathi's shorter history of inhabiting Cork shaped a different understanding and lived experience of homemaking that was informed by her 'new gaze' at home. Structural possibilities and impossibilities that enabled and prohibited us in making, imagining and feeling Cork as a home were bound by these intersectional biographies. These are the intersectional factors that shape how, as researchers, we understand home as well. We started thinking about personal histories of places, the fresh and accumulated lived experiences of specific places throughout our biographies which shape how we make sense of our own positioning in place and our narratives of homes. This biographical approach to understanding places is related to the notion of 'emplaced intersectionality' proposed by Sircar (2022), which offers a geographical approach to emplacement that can serve to enrich intersectional analysis within feminist geography and entails attending to fields of power that are contextual and placebased. To summarise, our shared and divergent situated positionalities bring us to understand home epistemologically as a fluid, constantly changing phenomenon that is geographically contingent, intersectional and embodied both for us and for the migrants we write about.

#### 1.5 Overview of the Book

The rest of the book provides an overview of the sizeable and rich body of scholarship in the social sciences that explores questions of home and migration and that sheds light on the questions raised above. This overview is framed by the conceptual framework outlined here and by our own situated positionalities as migration researchers. Hence, it is not necessarily a complete or comprehensive encyclopaedia of research on home and migration and we acknowledge that there is a wealth of valuable studies that we have been unable to include. However, the intention of this Short Reader is to provide a pathway through the vast body of work that exists.

Taking seriously the claim that home-in-migration is situated and should be analysed with an awareness of, and identification with, the dynamics of gendered, classed, aged and racialised relations, three of the chapters focus on these categories

<sup>&</sup>lt;sup>1</sup>Home-home is a colloquialism, commonly used in Ireland, that distinguishes between two different meanings of home – the home that is the domestic dwelling where I sleep at night (home) and the real home where my family is (home-home).

of power and identity – gender (Chap. 2), age (Chap. 3), migration status, class and race (Chap. 5). Chapter 4 focuses on materiality, as an illuminating lens through which an intersectional analysis of home in migration can be grounded. Within each chapter, particular attention is paid to the ways in which each axis of difference works intersectionally with others. Space does not permit us to devote as much attention to other axes of difference, such as sexuality (but see: Fortier, 2001, 2003; Waitt & Gorman-Murray, 2011). Chapter 6 is a short conclusion to summarise the contribution of this volume to scholarship on home in migration. Across all of these chapters, the discussions pay particular attention to the role of migration regimes and the ways in which they work (intersectionally) to shape structural possibilities and impossibilities of home.

Chapter 2 analyses gender in relation to home-making and home in migration. Gender is an essential factor that shapes home-making in any context. The chapter sets the scene with a brief discussion of the feminist scholarship on home and goes on to draw on existing literature to explore the relationships between gender, mobility and home. Although the discussion shows that traditional gender roles and normative practices can be challenged in migration, there is still much evidence showing that traditional gender roles are conserved and practised through processes of home-making in migration.

Chapter 3 focuses on the importance of age, generation and life-course transitions in the analysis of home in migration. Home is intimately bound up with lived experiences of age and life-course stage – hence the chapter focuses respectively on home in migration in older age, young adulthood and childhood. The existing literature reveals how meanings of home in migration shift across the life-course and how some age-related understandings, such as those associated with childhood, can be overlooked in mainstream migration/home literature.

Chapter 4 discusses the importance of material aspects of home in migration. The scholarship on materiality of migration, in its limited scope, tends to place the role of objects mostly within the context of memory, overlooking their other functions such as their instrumentality and use as objects to facilitate life. Others seem to locate material objects in relation to identity and processes of identification and their biographical importance. We draw on these differences to offer an analytical framework that focuses on objects in migrant home-making that are reminiscent of memories of old homes, are used as everyday objects or are of importance to one's identity. This chapter offers a fresh basis for further research into the migrant materialities field.

Chapter 5 places migration regimes in the foreground, focusing on the role of migrant status and relatedly, of race and social class, as powerful differentiating forces shaping the structural im/possibilities of home in migration. Migrant status is closely bound up with the racism and classism of contemporary migration regimes, and this entanglement is reflected in the literature. To attempt to disentangle these then seems counter-productive, so in order to navigate the literature, the chapter deploys the concepts of privilege and precarity, both of which are indicative of how migrant status, class and race intersect to position migrants differentially and to shape their im/possibilities of home.

Chapter 6 concludes that although the field of home in migration is expanding in recent decades, there is still much to be learned about it in the mobile world in which we are living. Home after all is what we desire to have, to achieve and to aim for, and the possibilities of achieving it seem to be ever more volatile in the current context of mass mobility. This is a rich and vibrant area of scholarship for migration studies and we hope that this book will offer another drop in the ocean of the research being conducted in relation to migration and home and open up new pathways into further focused research.

#### References

- Ahmed, S. (1999). Home and away: Narratives of migration and estrangement. *International Journal of Cultural Studies*, 2(3), 329–347.
- Ahmed, S., Castañeda, C., Fortier, A. M., & Sheller, M. (Eds.). (2003). Uprootings/regroundings: Questions of home and migration. Routledge.
- Al-Ali, N., & Koser, K. (2003). Transnationalism, international migration and home. In N. Al-Ali & K. Koser (Eds.), *New approaches to migration? Transnational communities and the transformation of home* (pp. 1–14). Palgrave.
- Altman, I., & Werner, C. M. (1985). Home environments. Plenum.
- Amelina, A. (2017). Transnationalizing inequalities in Europe: Sociocultural boundaries, assemblages and regimes of intersection. Routledge.
- Bernt, M. (2019). Urban regimes' and 'migration regimes': Contradictions, connections and pitfalls. *Erdkunde*, 73(1), 9–18.
- Blunt, A., & Dowling, R. (2006). Home. Routledge.
- Blunt, A., & Varley, A. (2004). Geographies of home. Cultural Geographies, 11(1), 3-6.
- Boccagni, P. (2014). What's in a (migrant) house? Changing domestic spaces, the negotiation of belonging and home-making in Ecuadorian migration. *Housing, Theory and Society*, 31(3), 277–293.
- Boccagni, P. (2017). Migration and the search for home: Mapping domestic space in migrants' everyday lives. Palgrave Macmillan.
- Boccagni, P. (Ed.). (2023). Handbook on home and migration. Edward Elgar Publishing.
- Boccagni, P., & Duyvendak, J. W. (2019). On the 'scaling up' of home in majority-minority relations. HOMInG WP 6. https://homing.soc.unitn.it/publications/working-papers/. Accessed 30 Aug 2023.
- Boccagni, P., & Miranda Nieto, A. (2022). Home in question: Uncovering meanings, desires and dilemmas of non-home. *European Journal of Cultural Studies*, 25(2), 515–532.
- Brah, A. (1996). Cartographies of diaspora: Contesting identities. Routledge.
- Brettell, C. (2006). Introduction: Global spaces/local places: Transnationalism, diaspora, and the meaning of home. *Identities: Global Studies in Culture and Power*, *13*(3), 327–334.
- Brickell, K. (2012). 'Mapping 'and 'doing' critical geographies of home. *Progress in Human Geography*, 36(2), 225–244.
- Brun, C., & Fábos, A. (2015). Making homes in limbo? Conceptual framework. *Refuge: Canada's Journal on Refugees*, 31(1), 5–18.
- Cho, S., Crenshaw, K. W., & McCall, L. (2013). Toward a field of intersectionality studies: Theory, applications, and praxis. *Signs: Journal of Women in Culture and Society*, 38(4), 785–810.
- Christou, A. (2006). Deciphering diaspora Translating transnationalism: Family dynamics, identity constructions and the legacy of 'home' in second-generation Greek-American return migration. *Ethnic and Racial Studies*, 29(6), 1040–1056.

- Crenshaw, K. (1989). Demarginalizing the intersection of race and sex: A Black feminist critique of antidiscrimination doctrine, feminist theory and antiracist politics. *University of Chicago Legal Forum*, 139–167.
- Crenshaw, K. (1991). Mapping the margins: Intersectionality, identity politics, and Vio-lence against women of color. *Stanford Law Review*, 43(6), 1241–1299.
- Cuba, L., & Hummon, D. M. (1993). Constructing a sense of home: Place affiliation and migration across the life cycle. *Sociological Forum*, 8(4), 547–572.
- Cvajner, M., Echeverría, G., & Sciortino, G. (2018). What do we talk about when we talk about migration regimes? The diverse theoretical roots of an increasingly popular concept. In A. Pott, C. Rass, & F. Wolff (Eds.), *Was ist ein migrationsregime? What is a migration regime?* (Migrationsgesellschaften). Springer VS.
- Erdal, M. B., & Sagmo, T. H. (2017). Descent, birthplace and residence: Aligning principles of citizenship with realities of migrant transnationalism. *Norsk Geografisk Tidsskrift - Norwegian Journal of Geography*, 71(4), 208–219.
- Erdal, M. B., Doeland, E. M., & Tellander, E. (2018). How citizenship matters (or not): The citizenship-belonging nexus explored among residents in Oslo, Norway. *Citizenship Studies*, 22(2), 705–724.
- Fathi, M. (2021). Home-in-migration: Some critical reflections on temporal, spatial and sensorial perspectives. *Ethnicities*, 21(5), 979–993.
- Fathi, M. (2022a). Constructing home through unhome: Narratives of resistance by an Iranian asylum seeker in Germany. *Social Sciences*, *12*(1), 16.
- Fathi, M. (2022b). 'My life is on hold': Examining home, belonging and temporality among migrant men in Ireland. *Gender, Place & Culture, 29*(8), 1097–1120.
- Fathi, M., & Ní Laoire, C. (2023). Urban home: Young male migrants constructing home in the city. *Journal of Ethnic and Migration Studies*, 49(3), 813–831.
- Fortier, A. M. (2001). 'Coming home': Queer migrations and multiple evocations of home. *European Journal of Cultural Studies*, 4(4), 405–424.
- Fortier, A. M. (2003). Queer migrations and motions of attachment. In S. Ahmed et al. (Eds.), Uprootings/Regroundings: Questions of home and migration (pp. 115–136). Routledge.
- Fox, M. A. (2016). Home: A very short introduction. Oxford University Press.
- Gonzales, R. G., & Sigona, N. (2017). Mapping the soft borders of citizenship: An introduction. In Within and beyond citizenship (pp. 1–16). Routledge.
- Gray, B. (2000). Gendering the Irish diaspora: Questions of enrichment, hybridization and return. Women's Studies International Forum, 23(2), 167–185.
- Guma, T., & Dafydd Jones, R. (2019). "Where are we going to go now?" European Union migrants' experiences of hostility, anxiety, and (non-) belonging during Brexit. *Population, Space and Place*, 25(1), e2198.
- Hage, G. (1997). At home in the entrails of the west: Multiculturalism, ethnic food and migrant home-building. In *Home/world: Space, community and marginality in Sydney's west* (pp. 99–153). Pluto Press.
- Hamilton, O. (2017). Senses of home. In J. Lloyd & E. Vasta (Eds.), *Reimagining home in the 21st century* (pp. 179–191). Edgar Elgar.
- Hatfield, M. E. (2010). Children moving 'home'? Everyday experiences of return migration in highly skilled households. *Childhood*, 17(2), 243–257.
- Horvath, K., Amelina, A., & Peters, K. (2017). Re-thinking the politics of migration. On the uses and challenges of regime perspectives for migration research. *Migration Studies*, 5(3), 301–314.
- Howlett-Southgate, R. (2021). *Life after Asylum: Investigating the precarious geographies of home with refugees leaving Direct Provision in Cork, Ireland.* Unpublished PhD thesis, University College Cork.
- King, L. A. (1998). Economic growth and basic human needs. *International Studies Quarterly*, 42, 385–400.
- Lloyd, J., & Vasta, E. (2017). Reimagining home in the 21st century. In J. Lloyd & E. Vasta (Eds.), *Reimagining home in the 21st century* (pp. 1–20). Edgar Elgar.

- Lulle, A., Moroşanu, L., & King, R. (2018). And then came Brexit: Experiences and future plans of young EU migrants in the London region. *Population, Space and Place, 24*(1), e2122.
- Lutz, H. (2017). Care as a fictitious commodity. Reflections on the intersections of migration, gender and care regimes. *Migration Studies*, 5(3), 356–368.
- Mallett, S. (2004). Understanding home: A critical review of the literature. *The Sociological Review*, 52(1), 62–89.
- Mata-Codesal, D. (2023). Feeling at home: Migrant homemaking through the senses. In P. Boccagni (Ed.), *Handbook on home and migration* (pp. 228–238). Edward Elgar Publishing.
- McCall, L. (2005). The complexity of intersectionality. *Signs: Journal of Women in Culture and Society*, *30*(3), 1771–1800.
- Meijering, L., & Lager, D. (2014). Home-making of older Antillean migrants in The Netherlands. *Ageing & Society*, 34(5), 859–875.
- Miller, R. G. (2019). (Un)settling home during the Brexit process. *Population, Space and Place,* 25(1), e2203.
- Miranda Nieto, A. (2021). Scales. In A. Miranda-Nieto et al. (Eds.), *Ethnographies of home and mobility: Shifting roofs* (pp. 15–38). Routledge.
- Miranda Nieto, A., Massa, A., & Bonfanti, S. (2021). Ethnographies of home and mobility: Shifting roofs. Routledge.
- Mitchell, R. E. (1971). Social implications of high density housing. *American Sociological Review*, 36, 18–29.
- Mitzen, J. (2018). Feeling at home in Europe: Migration, ontological security, and the political psychology of EU bordering. *Political Psychology*, *39*(6), 1373–1387.
- Moore, J. (2000). Placing home in context. Journal of Environmental Psychology, 20(3), 207–217.
- Muñoz, S. (2018). Precarious city: Home-making and eviction in Buenos Aires, Argentina. *Cultural Geographies*, 25(3), 411–424.
- Myers, D. T. (1989). Shared housing: An adaptable affordable housing option. *Journal of Housing*, 46(6), 297–305.
- Nash, C. (2003). 'They're family!' Cultural geographies of relatedness in popular genealogy. In S. Ahmed et al. (Eds.), *Uprootings/Regroundings* (pp. 179–206). Berg.
- O'Connor, E. P., & Crowley-Henry, M. (2020). From home to host: The instrumental kaleidoscopic careers of skilled migrants. *Human Relations*, 73(2), 262–287.
- Parker, O., & Catalán, Ó. L. (2014). Free movement for whom, where, when? Roma EU citizens in France and Spain. *International Political Sociology*, 8(4), 379–395.
- Parnell, T. (2023). The representation of migrant identities in UK Government documents about Brexit: A corpus-assisted analysis. *Journal of Language and Politics*, 22(1), 46–65.
- Pérez Murcia, L. E., & Boccagni, P. (2022). Of home-comings and home-scales: Reframing return migration through a multiscalar understanding of home. *Global Networks*, 22(3), 499–513.
- Pink, S. (2004). Home truths: Gender, domestic objects and everyday life. .
- Ralph, D., & Staeheli, L. A. (2011). Home and migration: Mobilities, belongings and identities. *Geography Compass*, 5(7), 517–530.
- Ranta, R., & Nancheva, N. (2019). Unsettled: Brexit and European Union nationals' sense of belonging. *Population, Space and Place*, 25(1), e2199.
- Sircar, S. (2022). Emplacing intersectionality: Autoethnographic reflections on intersectionality as geographic method. *Gender, Place & Culture, 29*(7), 903–922.
- Sirriyeh, A. (2016). Inhabiting borders, routes home: Youth, gender, asylum. Routledge.
- Skrbiš, Z. (2008). Transnational families: Theorising migration, emotions and belonging. *Journal of Intercultural Studies*, 29(3), 231–246.
- Staeheli, L. A., & Nagel, C. R. (2006). Topographies of home and citizenship: Arab-American activists in the United States. *Environment and Planning A: Economy and Space*, 38(9), 1599–1614.
- Stoetzler, M., & Yuval-Davis, N. (2002). Standpoint theory, situated knowledge and the situated imagination. *Feminist Theory*, 3(3), 315–333.
- Svašek, M. (Ed.). (2012). Moving subjects, moving objects: Transnationalism, cultural production and emotions (Vol. 1). Berghahn Books.

- Tolia-Kelly, D. P. (2004). Materialising post-colonial geographies: Examining the textural landscapes of migration in the South Asian home. *Geoforum*, 35(6), 675–688.
- Tyrrell, N., Sime, D., Kelly, C., & McMellon, C. (2019). Belonging in Brexit Britain: Central and Eastern European 1.5 generation young people's experiences. *Population, Space and Place,* 25(1), e2205.
- Waitt, G., & Gorman-Murray, A. (2011). Journeys and returns: Home, life narratives and remapping sexuality in a regional city. *International Journal of Urban and Regional Research*, 35(6), 1239–1255.
- Walsh, K. (2011). Migrant masculinities and domestic space: British home-making practices in Dubai. Transactions of the Institute of British Geographers, 36(4), 516–529.
- Walters, W. (2004). Secure borders, safe haven, domopolitics. Citizenship Studies, 8(3), 237-260.
- Wilkins, A. (2019). *Migration, work and home-making in the city: Dwelling and belonging among Vietnamese communities in London.* Routledge.
- Williams, F. (2012). Converging variations in migrant care work in Europe. Journal of European Social Policy, 22(4), 363–376.
- Yuval-Davis, N. (2006). Belonging and the politics of belonging. *Patterns of Prejudice*, 40(3), 197–214.
- Yuval-Davis, N. (2015). Situated intersectionality and social inequality. *Raisons politiques*, 2, 91–100.
- Yuval-Davis, N., Wemyss, G., & Cassidy, K. (2018). Everyday bordering, belonging and the reorientation of British immigration legislation. *Sociology*, 52(2), 228–244.

**Open Access** This chapter is licensed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license and indicate if changes were made.

The images or other third party material in this chapter are included in the chapter's Creative Commons license, unless indicated otherwise in a credit line to the material. If material is not included in the chapter's Creative Commons license and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder.



### Chapter 2 Gender and Home in Migration



#### 2.1 Introduction

It is now well established that gender and migration are inextricably interlinked (Cresswell & Uteng, 2012; Christou & Kofman, 2022; Yeoh & Ramdas, 2014). Gendered identities and power relations define and shape spaces and mobilities (Massey, 2013) and thus are pertinent in how processes of home-making and migration are both understood and experienced. Furthermore, gender is acknowledged as a social construct with meanings that can vary from context to context. For example, conventional understandings of migration have constructed it as normatively masculine, that is, emigration and return migration have been seen historically as naturally masculine endeavours. The archetypal conventional 'migration story' is one where men would leave their homes with the hope of finding success overseas and after a while return to their families, or their families would follow them (Hondagneu-Sotelo, 1994). Although this type of migration was and still is a common social phenomenon in different societies, this narrowly gendered idea of how migration, gender and family interrelate is one that denies the long history of female migration, child migration and family migration. Since the early 1970s, feminist voices have challenged traditional heteronormative assumptions about migration (Christou & Kofman, 2022; Kofman & Raghuram, 2015). The highly significant role played by women in global migration and global labour markets is now widely recognised; furthermore, scholarship agrees that migration is a highly gendered process and is shaped by the intersectionality of gender with race, ethnicity, nationality, class and sexuality (Mahler & Pessar, 2006).

At the same time, the idea of home-making as an unproblematically female practice has also been challenged, as feminist-inspired literature has highlighted the gender inequalities that are re-produced through dominant gendered practices of home-making in particular in the domestic sphere (Young, 2005). Feminist scholarship has highlighted how sites called 'home' are often places of oppression and violence against women (Blunt & Dowling, 2006; Hanmer & Itzin, 2000). In fact, gendered identities and power relations are so deeply embedded in how we understand and do home that they cannot be disentangled. Hence, in this chapter we focus on the formation of gendered identities and their impact on understandings and experiences of home in migration. To clarify, in this chapter we do not aim to impose a binary on gendered identities, but we focus in particular on the categories of masculinity and femininity, reflecting the existing literature on home-making and migration. Our aim is to offer a novel reading into gendered practices of homemaking in migration, by focusing on several dimensions of this relationship, given how central gender is to the imagining and doing of home.

Before delving into the literature on gender and home in migration, we present some insights from feminist scholarship that are particularly relevant to understanding the relationship between home and gender. This forms the backdrop to the subsequent section, which explores the shifting relationships between women's mobilities and constructions of home and homeland, before going on to explore how gendered spatial boundaries of home are re-worked and re-produced in contexts of migration. The next section focuses on migrant women's experiences of homemaking and this is followed by a section that highlights emerging literature on changing gender relations and domestic masculinities in migrant homes. Finally, we address temporality, exploring how gendered notions of home in migration are connected to imaginaries of future homes.

#### 2.2 Feminist Thinking on Home

Feminist scholarship has made a major contribution to how we understand both home and migration. Feminist scholars have been actively writing about home for a long time (Ahmed, 1999; Brah, 1996; Kaplan, 1987; Salih, 2003; Werbner, 1999; Young, 2005) as 'the home' has presented itself as the first space where gendered identities are shaped since childhood, learned through the process of socialisation and practised, and as a key site in the re/production of gendered power relations.

Feminist scholars (mainly within socialist and Marxist traditions) have criticised the division between the private sphere (associated with femininity and nurturing) and the public sphere (associated with masculinity), pointing to the role played by this division in unequal access to means of production (Fernández-Kelly, 2000). During industrialisation, private space became the sphere of so-called 'natural' practices such as reproduction and sustenance of the family, while the public arena was the site of social, economic and cultural production of capital. One of the most important assertions in feminist approaches is that home is *both* a private and public space where gendered, classed and racialised relationships are enacted and experienced (Longhurst, 2012) and where women are exploited through multiple systems of oppression (Kuhn & Wolpe, 1982; Vogel, 1983). Through socialist feminism the intersections of sexism, class oppression and racism have been illuminated (Joseph, 1981; Naples, 2003). Feminist scholarship across all its disciplines points to how, historically, women's presence in the domestic space not only has been an effect of

industrial capitalism but also a cause of their subjugation in relation to legal rights within the home sphere (Ardener, 1981; McDowell, 1999). As Fraser (2021, p. 150) puts it, the family provides 'the social-reproductive conditions for capitalist production'. As such the domestic space earns an important role in the divisions of power based on class, gender and race.

One of the reasons feminist thinking has flourished in research on home is because gender is crucial to lived experiences of home and the practices that help in constructing and maintaining a home. Young (2005) writes about how gender has been fundamental in ownership, claims, boundary reconstruction, and titles relating to land and housing. Whilst men have benefitted from the privilege of being involved in construction and ownership of domestic property, women have been seen as 'cultivating' that space and preserving it through domestic practices, in other words, the real home makers (Young, 2005). McDowell (1999) argues that this power-imbalanced spatial division has shaped and been shaped by other social relations involving indirect control of (female) bodies. In other words, classed and racialised divisions in relation to private and public space are co-constituted with gendered power relations. Marxist and socialist feminism in particular has focused on how classed and gendered divisions in society are reinforced through and within the family (Delphy, 1984; Fraser, 1994). After all, capitalism needs the free care labour to look after society's younger generations for its own survival.

These gendered divisions of labour at the level of the household are reflected in gendered international divisions of labour, including the expansion in recent decades of global migration of female labour to meet the demand for low-paid service work (often in sectors related to social reproduction) in countries of the Global North. Enloe (2007) argues that capital relations of gender have made women a docile and cheap workforce that can be manipulated through state, legal and physical coercion. It could be argued that the growth in migration of mostly female care workers for a global care system that attracts cheap labour from Global South to Global North countries, in effect is about making homes comfortable for some, at the expense of others being denied the possibility of having their families at home with them (Parreñas, 2008). These feminist insights have important implications for understandings of home in contexts of migration, drawing attention to the role of micro-level gender power relations within domestic (and transnational) migrant spaces and hence for the meanings of home for migrant households and families (as is explored later in the chapter).

From a different perspective, Black feminism has also been extremely influential in opening up important questions about family, home and migration. Hooks (1990) has argued that family and home have become an institution where women provide privatised care for children, older and male family members: home is the place where patriarchy and racism are experienced together. These views, alongside Crenshaw's (1991) introduction of the concept of intersectionality, based on the interrelation of race and gender on a macro scale, have forged fresh thinking about home and gender. Hill Collins (2022) argues that the perception of motherhood and women's housework as an unpaid 'duty' compared to male paid occupations outside the home has challenged the traditional view on family, but that this mainly white

model was not widespread among African American women. She argues that Black women have used a variety of strategies to undermine the oppressive power of patriarchy at home, and that the home can be seen as a site of empowerment as well as oppression. In particular, Black feminist scholarship has emphasised that for Black women, the home can be a place of refuge from a racist society (hooks, 1990). Furthermore, the recognition that experiences of home and migration are intersectional means acknowledging that migrants experience gendered aspects of home and mobility differently depending on their intersectional social positions in relation to race, class and migration history.

Thus, gender relations of home in migration have to be understood in the context of how they emerge from different histories of migration, colonialism and dis/placement and hence their politics of belonging. Ahmed (1999) for example is critical of the erasure of different histories of migration in narratives of home and migration. Boyce Davies (1994), by looking at the writings of Afro-Caribbean women in the USA, marks out a 'politics of location' in relation to Black women's experiences of displacement and argues that mythical representations of 'home' as a singular origin are undermined by their experiences. She points to the contradictions of the notion of home for those whose 'homes' have been colonised and for those who are regularly asked where they are from even as they are 'at home'. Hence, the meaning of home as an unproblematic place of origin and belonging is brought into question. Similarly, Brah (1996) questions notions of home in migration that assume a simple association between place of origin, identity and belonging, arguing instead that there is no essential connection between these. According to Ahmed et al. (2020) [2003]), there is a need to move beyond notions of home as something one leaves behind or that is fixed prior to migration. Instead, they argue that 'homes are always remade as grounds and conditions of family/work etc.' (p. 9) and home and mobility are seen as intertwined processes. The same geographical space can have different histories and meanings for different groups or individuals, linked to histories of colonisation, displacement and patriarchy (Brah, 1996).

Feminist scholarship from both traditions therefore has made a number of key contributions of value in analysing home and migration. First, it has highlighted how homes can be a site of both repression and empowerment for women. Second, it draws attention to micro-level intersectional gendered politics of home-making as well as situating home within the context of larger socio-political relations, economic structures and colonial histories. Third, it has undermined long-held assumptions about home as a fixed place of origin, and finally, it demonstrates how experiences of home are not equal for all.

#### 2.3 Women's Mobilities and Home/Land

Within discussions on mobility and home, there lies an important aspect of home in migration: Who has the power to move? Who is being left behind? How does the gender of mobile and sedentary family members determine such statuses and how

are gendered experiences changed by these movements? With the expansion of migratory journeys in the latter part of the twentieth century, the notion of homes for migrant families (both in relation to the home that is made after migration and the home that is left behind) has changed. Traditional perceptions of men being mobile and women being sedentary home-makers were to an extent exacerbated by imperialist impulses towards nation-building. For example, McClintock (1995), from a gendered angle, looks at the formation of the British Empire and particularly focuses on the 'functional' role of women in its formation (see also the classic text by Yuval-Davis, 1997). Both McClintock (1995) and Yuval-Davis (1997) argue that the role of women in building domestic space and reproduction of subsequent generations was paramount to the reproduction of masculinity and nation as a collectivity. Aitken (2009) focuses on the role of domestic space in facilitating colonial homes, arguing that the 'civilised' domestic space of home was an important symbol and means of imperialism. Others have highlighted the role of traditional constructions of womanhood, often linked to motherhood, in dominant ideas of nation as home (Gray, 2004). Women's reproductive ability has been constructed as vital in the process of creating the nation and nurturing younger generations (Yuval-Davis, 1997), an expectation that can work to limit women's mobility.

However, research on female migration has challenged traditional constructs of 'men who migrate, women who wait' (Brettell, 2014) by showing how home, through the migration of women and border-crossings has been reconfigured and transformed (Wilkins, 2017). These contributions reveal how gendered constructions of space and place are challenged by the mobilities of women. For example, Giuffré (2017) has conducted research with Cape Verdean migrant women who leave their homes to go to Europe and North America, in order to support their families on Santa Antão island. The research shows that home for women who leave their children at the hands and mercy of spouses, extended family and even neighbours, at the minimum, contests traditional gendered roles in the construction of the space of home. Where the space of home, Terra Mamaizinha, was the known, domestic, and indoors closed off from the rest of the world as a feminine space, Terra Longe was the sea, the outdoors and outside world of home that was dangerous and where only men ventured to explore (Giuffré, 2017). The more permanent migration of women in these islands, as opposed to the cyclical and short-term migration of men, has made Terra Longe a feminised space. These gendered emigrations and the remittances sent home to pay for expenses of left-behind children bestow relative power to migrant women as heads of families, placing them in 'decision-maker' positions within households. These positions were held by men before the 1970s era of the emigration of women from the islands (Giuffré, 2017). This also meant that due to the involvement of other women as carers within the space of the immediate home or extended homes, women have gradually taken ownership of the process of receiving and managing remittances and nurturing of family members who have stayed put (Giuffré, 2017). What these shifting roles and understandings of gender in the context of spaces of Terra Mamaizinha and Terra Longe show is that within marketised liberal and global caring and migration regimes, gendered roles can change and, accordingly, the notion of home can change based on these shifting positionings. Even on a small island, notions of mobility, home and gender have been transformed drastically based on the power that women gain by becoming the main breadwinners of their families.

The shifts in gendered relations between migrant and homeland that are triggered by women migrants' greater independence as a result of migration are potentially far-reaching. For example, Constable's (1999) research on Filipina migrant women highlights their ambiguous attitudes towards returning to the place they call 'home' (back home). While on the one hand, they feel a strong attachment to this home, they are aware that their financial independence would disappear after returning; in other words, their imaginings of home change with migration. This relationship between migrant home-making, gender and homeland is complex. Studies on return migration show that the notion of homeland as a place of belonging depends considerably on the social conditions under which people originally left for other destinations. Sometimes structures such as patriarchal systems or ethnic and religious persecution reduce the tendency to return and minimise attachment to homelands. For example, Hondagneu-Sotelo (1994) finds that Mexican women in the USA are more likely to integrate faster compared to their husbands and have less tendency to return to their homelands. Similarly, Graham and Khosravi (1997) in a study with Iranian migrant men and women in Sweden, find that the interest towards returning to their homeland, Iran, is higher among men than women. They attribute this to the patriarchal norms that grant men more freedom and power and place women in inferior positions socially, legally and within families. Yeoh and Willis (2005) also consider return migration as a gendered translocation which improves men's status and reinforces notions of 'masculinity', distinguishing between the experiences of men and women.

Some studies which have looked at gendered relationships with homelands see them from an intersectional lens, analysing geographies of power and positionalities in terms of privilege and precarity. One of these is Wong's (2014) research on Ghanaian skilled migrant women who returned to Ghana at their peak of productivity. She argues that these women migrants' diverse resources and their privileged class positions provided them with networks for recurrent transnational mobility and granted them choice and agency. Therefore, women migrants' positions in relation to their homelands, when considered intersectionally, can be as autonomous as those of male return migrants (Wong, 2014). On the other hand, there are studies that look at migration and transnational gendered practices from the perspective of undocumented migrant women whose transnational mothering and home-making practices are impacted by their precarious and disadvantageous positions (Brandhorst et al., 2020; Carling, 2014; Fresnoza-Flot, 2009; Lourenço & Cachado, 2012; Madianou, 2012; Madianou & Miller, 2011; Pratt, 2012; Ramirez et al., 2007). These scholars explore how transnational families manage physical caring practices in the context of immobility, resorting to different forms of maintaining family relationships such as regular visiting and maintaining intimacy through the use of mobile phones, ICTs and gift giving and sending.

What can be seen from these experiences is that the relationship with home and homeland is a gendered process and needs to be located within the macro politics of home-making in the countries of origin and destination. The gendered relationship one has to a homeland is always impacted by other social locations; for example, to what extent does one have freedom of expression in the country of origin as a female activist or a transgender man? When considering these delicate nuances, we can better understand how making a home is (structurally) possible and impossible in certain contexts and for some people and not for others. It is through a relational treatment of gender that one can understand the complex web of meaning-making in relation to home and notions of homeland among migrants and the families of migrants who stay put in other transnational locations (Salih, 2003; Sandu, 2013; Väänänen et al., 2005).

#### 2.4 Spatial Boundaries and Gender in Migration

Blunt and Dowling (2006), in their book, *Home*, draw attention to the intimate spaces of home and their interconnection with power and they argue that power relations define and determine the meanings of home in migration. They differentiate between different contexts of home ('imperial homes and home-making; home-land, nation and nationalist politics; and the politics of indigeneity, home and belonging') and show how home on a domestic scale is bound up with imperial, national and indigenous politics that are articulated through geographies of home (Blunt & Dowling, 2006, p. 142). Their critical geographical perspective on home gives a central role to gendered and racialized constructions and seeks to challenge conventional public-private distinctions; they show how internal politics of home are intimately bound up with 'domestic/national' and imperial politics.

In Marxist feminist analysis of domestic home space, the person who has the economic means is the one who decides over the distribution and the use of that space (Fraser, 2021). Within the private space then, the composition of material entities and symbolic values attached to them set out how a home is structured in relation to ownership, belonging, the right to use the space and so on. One cannot separate this from how power relations in place give to and/or take away 'spatial rights' from residents in a house (Bonfanti, 2021, p. 116). These gendered (and aged) power relations are articulated spatially in terms of boundaries of and in home – boundaries between public and private, inside and outside, ownership and lack of ownership.

Research in migration studies shows how patriarchal relations can be maintained and reproduced through migration, even if they change form (Gilmartin & Migge, 2016). These power relations are manifest spatially through and in the spaces of home after migration – the spaces of migrant households and of cities and neighbourhoods inhabited by migrant groups. Much literature reveals how gendered public-private divisions are actually maintained and even reinforced after migration. For example, Gilmartin and Migge (2016) show how well-educated migrant mothers in Ireland find themselves struggling with lack of childcare in a context where family support is crucial and thus defining themselves primarily as mothers and becoming confined to the space of the home.

The maintenance of gendered public-private spatial boundaries is a common feature of migrant communities and households with a greater tendency for migrant women to be more restricted to private home and domestic work spaces than men, who are more likely to inhabit public spaces of work and leisure. This can be related to the gendered nature of migration regimes which reinforce gendered responsibilities of caring and breadwinning after migration. For example, Buckley (1997) writes about the invisibility of Irish women in Britain in the twentieth century, despite their vast numbers, which she relates to the tendency to confine them to 'sheltered' environments such as the private home and hidden sectors of the economy (cleaning, caring, catering). Recent research has pointed to the isolation and loneliness of refugee women in destination societies who are unable to access spaces of employment, education or leisure outside the home (Casimiro et al., 2007; Ghorashi, 2010). Even among more elite female migrants, Yeoh and Khoo (1998, p. 172) found that 'the lines that divide the public world of work from the private world of home in a new environment had hardened, resulting in their relegation to the latter'; so even though they are 'at home', they may not feel at home.

The gendered segregation of intimate private spaces has been particularly evident in many Muslim contexts. In countries where traditionally a strict code of conduct is observed in relation to gender and space, women's presence and mobility are closely guarded and controlled by the male family members, extended family members and even the morality police (Samuri & Hopkins, 2023). Purdah, or curtain, that also extends to bodily coverage by a veil, is a practice mostly aimed at protecting a woman's chastity by hiding her physical presence from male strangers' gaze (Mazumdar & Mazumdar, 2001). This is a deeply religious concept that has translated into how homes are compartmentalised into private and public spaces and even dividing the time of the day and night when staying at home is encouraged or enforced to 'protect' women and girls' honours (Guglielmi et al., 2021; Mim, 2020). Relatedly, in contexts of migration, practices that seek to control the mobilities of Muslim young women are evident, for example in the everyday patterns of use of public spaces by second and third-generation young Muslims in Brussels (De Backer, 2020). Macey (1999) shows how in Bradford, in the north of England, young Pakistani girls are banned from certain activities such as going to university, school trips, or sports, due to the community male members maintaining such gendered divisions within and outside their communities, mostly against females' will. Research also shows how young migrant women develop selective strategies to challenge the patriarchal boundaries that they face both within and outside their homes (De Backer, 2020; Ehrkamp, 2013).

# 2.5 Migrant Women, Home-Making and the Domestic Sphere

The domestic space of home plays a particularly important role in how gender manifests in any social arena and gendered power relations have historically shaped home-making practices and feelings at the domestic scale. Practices that make a domestic space meaningful and that denote it as 'home' are highly entangled with emotions (Belford & Lahiri-Roy, 2019; Boccagni & Baldassar, 2015; Dyck, 2018) both positive and negative (Brickell, 2012) and are historically embedded within gendered values and expectations. Hence, the domestic scale has been the focus of much research on how the micro-level politics relating to gender relations and the right to space are re-worked after migration, through a particular focus on migrant homes.

Some scholars have written about the importance of practices of cooking and eating as elements of home-making in migration and in the maintenance of transnational ties of belonging (Povrzanović Frykman & Humbracht, 2013). While foodwork and the related space of the kitchen have been recognised as oppressive spaces for women, Meah (2014a) argues that for many migrant and ethnic minority women, these are also spaces of creativity, solidarity, sharing and love, from which they build new homes and develop a sense of place. In this sense, the domestic space of home can be a safe space for migrant women away from the racism of wider society. Longhurst et al. (2009), based on their research with migrant women in New Zealand, argue that for these women, the domestic space was one in which they felt very comfortable and relaxed, and for them cooking was a way of staying connected with past homes while recreating a new home.

This research points to an important argument that relates to the intersectionality of gender, race and ethnicity in migrant home-making experiences, which is that for migrant women who experience racism and denigration of their identities in their host societies, the domestic home space can represent a haven from this, and furthermore, can be a site of resistance to wider processes of home unmaking. Wallace's (2012) research reveals how Muslim women who experience anti-Muslim racism in the UK create the (domestic) home as a safe site of identity affirmation and political engagement, through their religious practice and their role as family educators. Similarly, Walter (2002) writes about the role of the private space of the home as a safe space for expression of Irish identities for Irish migrant households in Britain in the 1970s when anti-Irish hostility was common. Irish women played a crucial role in the maintenance of these homes as sites of negotiation of diasporic identities, through their roles as home-makers. Migrant women also often play a crucial role in migrant communities in host societies, as informal, often unpaid, community workers and key players in the maintenance of social and cultural migrant organisations (Buckley, 1997; Vacchelli & Peyrefitte, 2018; Yeoh & Khoo, 1998). Even though such work is often invisible, it is valuable in creating a shared sense of home (beyond the domestic space) among migrant communities (for example, Nititham, 2016).

While much of the literature on reproduction of the politics of home in migration contexts focuses on the role of migrant women and on the domestic space of home, Datta (2008) looks at how Polish men working as builders in London view their work in building homes for others as a validation of their masculine Polish identities and linked to imagined future homes in Poland. Similar to migrant women who leave their homes and children to care for other people's children and make others' homes as migrant workers (Fresnoza-Flot, 2009; Ukwatta, 2010), migrant men also engage actively in building the physical homes of others whilst temporarily live in these building sites or experiencing temporary homelessness. Therefore, practices of home-making in migration are bound up with the reproduction and re-working of both masculine and feminine identities.

# 2.6 Changing Gender Relations and Masculinities in Migrant Home-Making

So far, we have discussed how the gendering of home is maintained after migration and how traditional gender roles are reproduced, whether within the space of the domestic home, in migrant communities or at the level of discourses of nation or empire. However, some recent developments in discussions of gender in homemaking processes tell us about how such traditional understandings of the space of a home are often challenged through the migration process (Gorman-Murray & Hopkins, 2014). An important line of enquiry here is scholarship on migrant masculinities, which points to some ways in which hegemonic masculine identities in relation to home are changing (Donaldson et al., 2009; Gallo & Scrinzi, 2016; Palenga-Möllenbeck, 2016; Walsh, 2011; Wojnicka & Nowicka, 2022). This body of scholarship opens up new ways of thinking about gender, home and migration; for example Ye's (2014) research on migrant masculinities and class is an interesting addition to this field.

Home is a material, spatial, temporal and symbolic process that rewrites itself within historical and cultural contexts, and in addition, social locations of gender, class, race, ethnicity, sexuality need to be understood as fluid and contingent in the process of transnational dislocation, or what Anthias (2008) calls 'translocation'. As such it is important not to presume that masculinity and femininity are fixed identities that only change when people move from one country to another; on the contrary, home as a gendered construct is constantly in process. But there are intricate differences when men and women move as opposed to when they do not, and this change of setting has important implications for their ability to make home. The questions that are raised here in relation to home and its gender, age and class power relations are amplified when migration experiences are added.

In our earlier research on home and belonging in Cork, the migrant men we interviewed discussed the act of cooking as a marker of their domesticated masculinities (Fathi, 2022). This led us to explore this new shift in relation to home and

gender and the idea of alternative masculinities. Recent studies of masculinities show that the practices and spaces of home cooking, previously considered a female domain, are arguably becoming more connected to men and masculine worlds (Cox, 2014; Gorman-Murray, 2014; Meah, 2014b). Szabo (2014) discusses how hege-monic masculinities become contested when men engage in home cooking as a feminine practice. Their study shows that the act of cooking, set in the kitchen space, is not an act devoid of any home politics. Gendered relationships within the space of home are complex but migration adds a new layer of complexity as new gendered relations can be formed after mobility.

When ethical and political values within a home do not correspond with the hegemonic values prevalent in the host society after migration, the space of home can become a site of contestation. Some studies with different migrant groups in family homes show that within the same family, gendered practices can change as a result of transformations in expectations of gender roles within the wider context of ethical and political values. For example, Shahidian (1999) in his research with Iranian migrants in Canada, argues that gender relations and sexuality change drastically as a result of migration when Iranian families move from Iran to Canada. He finds that in Iranian families in his study, leaving a misogynist regime, women find their public and private lives in many ways transformed in terms of their choices regarding gender roles, family life, occupation, dress code, and sexuality. Shahidian's (1999) findings are consistent with studies conducted with Iranian families in other countries (Graham & Khosravi, 1997; Fathi, 2017). These studies find that women in general can access the labour market more easily compared to men, which in many cases, leaves men to do more of the domestic chores, a position that contrasts with the traditional roles that men perceived for themselves (see also Hamada's 2017 research on Japanese house husbands). Shahidian (1999) argues that Iranian men in his study referred to themselves as matarsak (scarecrow), indicating the decline in patriarchal gendered divisions at home and undermining of their masculine identities. Hamada's (2017) research with Japanese migrant men who take on the role of house husbands finds that most men in her study expressed discomfort with taking up domestic chores that they would not do in Japan and they would not disclose their participation in them after migration.

Another similar study is Charsley's (2005) on Pakistani families living in the UK, where British Pakistani young girls marry young men in Pakistan who migrate to the UK subsequently. The reversal of the traditional roles within homes in such scenarios (the migrant groom being dependent on the wife and her family) is a traumatic experience for migrant men, as the new couple's idea of home is usually tied to traditional gender roles (the man being the sole breadwinner and the woman being a housewife). However, the new groom's dependence on his in-law family, as a result of his inability to find a job and thus to live separately from the in-laws, shows a reversal of this ideal home scenario (Charsley, 2005). This is mainly due to the fact that the purchasing of a house or affording to rent a place at the start of such transnational movements is not possible. The term *ghar damad* (meaning imported son-in-law who lives with the wife's family, or what is called house son-in-law) is then used as a term to define the reversed gender and social positioning roles and

specifically refers to the intersection of gender and home. It describes dependency and lack of individuality and in practice it is a derogatory term that participants in Charsley's (2005) study avoid using to refer to themselves. Yet they still see themselves as controlled heavily by their lack of financial and housing independence within restructured gendered household relations of power that redefine their gendered identities and bind them to the home sphere in a negative way (Charsley, 2005).

Hamada's (2017) research with Japanese migrant men in Australia focuses on transnational domestic masculinity and these migrant men use the domestic space, as house husbands, to perform practices such as home cooking that were/are seen as women's work in Japan. However, as Hamada (2017) argues, in post-migration contexts these traditional arrangements change mostly due to the positions that each spouse occupies outside the home and the necessary roles that need to be performed accordingly. Percot (2012) looks at the situation of migrant men from Kerala who emigrate as spouses dependent on their nurse wives or as 'followers' (no page number) to their wives who act as the main migrant applicant. She argues that these men find their positions disturbing, not knowing what careers they can pursue when arriving in a western country (in this case, Ireland). "This is not a man's job" explains Matthew, one of the participants who describes how as a family they decided he would look after their toddler child until she goes to school and Matthew can start a 'proper' job (Percot, 2012, no page number). Although we are mainly dealing with questions around socio-economic status and position of the migrant working spouses, at a deeper level, these testimonies tell a lot about how financial necessities are providing a new gendered reality to the home-making practices of migrant families.

Changing notions of masculinity and femininity, in relation to home-in-migration pathways are also intersected with global systems of racialisation and class. In the provision of care for families, cheap labour is provided by female migrants who travel to more developed countries to undertake caring positions, contributing to what Hochschild (2000, p. 131) calls 'the global care chain'. In affluent countries privileged women (and men) buy domestic help at low wages to raise their families, a form of employment that Gallo and Scrinzi (2016, p. 1) call 'outsourcing' of domestic/care tasks, mostly employing women in exploitative conditions who migrate transnationally to look after others' elderly, children, or the sick within families. They question the general tendency to view men as 'outsiders' to the domain of home and this matrix of the transnational caring system, and they argue that, despite more men taking up transnational domestic roles, this issue has not been taken up in the scholarship in home and migration.

In fact, migrant men working as domestic workers provide the focus for a notable line of research in relation to issues of gender, home and power. Whilst most of the literature that concentrates on 'hired help' or domestic workers focuses on women (Dodson, 2008; Fresnoza-Flot, 2009; Oishi, 2005), the provision of care for families using the cheap labour of migrant men who travel to more developed countries is also on the rise. Studies on migrant men working in internal spaces of homes as employees, workers, and helpers, reveal that they often do so as undocumented migrants, which facilitates their exploitation and sub-standard work conditions

(Gallo & Scrinzi, 2016). The implications for migrant masculinities are revealed in the constructions of gendered roles in relation to home-making. For example, Gallo (2006) in research with Malayali men in Italy finds that although practices such as cooking and laundry were seen as feminine, performing them was accepted with some unease. For this group of migrant men, who worked as domestic carers, performing jobs that usually belonged to realm of women meant that their gendered positions were seen as complex and ambiguous – on the one hand the men asserted their respectability due to their status as employees but on the other hand their Malayali masculine identities were threatened (Gallo, 2006).

Some have criticised the representation of the domestic space as essentially a feminine space and the view of men in domestic space as 'out of norm' (Gorman-Murray, 2008). It is important not to over-state the transformations in gender relations of the migrant home. Walsh (2011) argues that the site of home in the UK has become more gender egalitarian with a more even division of labour and that this shift is also visible among British 'expats' who live in Dubai. She shows that men in these households engage with housework as part of a practice of home-making, but importantly, even these activities remain limited and minimal.

Gorman-Murray (2008) argues that although feminist scholarship about home has helped a lot in understanding power relations at home, describing it in relation to femininity, it does little to help in understanding other gendered identities being shaped in the home sphere. This is an interesting and valuable reading of home and gender, as bringing masculinities (even as privileged subjects) into the home sphere helps to understand the processes through which marginalised positions are shaped. The shifts in gendered identities within households, especially migrant households, must be understood in the context of the larger social changes that have happened globally. These include women's increased participation in the labour market, their higher rate of solo migration (with families and children left behind), of migrating as the main applicants in migration processes or as sole breadwinners after migration (thus becoming integrated within the social spaces of host countries faster their male spouses – Shahidian, 1999). However, when considering the structural im/ possibilities of home, intersectional understandings of home and gender are important. For example, women in middle-class migrant families have different access to resources compared to those in economically fragile families. Peinhopf (2014) argues that when gender is intersected with class, different roles can be assumed and practised. We also need to be mindful about how other identity categories, such as sexuality, impact gendered experiences of migration (for example see Fortier, 2020 [2003]).

What we see here is that gendered roles are complicated particularly when we consider other identity categories and how they intersect with gender in migration contexts. In host countries, language and employment requirements, pre-migration histories of employment, sexual identities, ability and disability, number of children, language proficiency (among others) impact gendered identities and roles. After all, in the post-migration settings dominant and mainstream masculinities and femininities are continuing to exist in their traditional forms alongside new alternative forms because these intersectional identities are never fixed, even in the countries of origin

(see Anthias, 2008). As such, gendered practices still conform to 'accepted' and normalised roles for men and women, alongside alternative understandings of gender and home.

# 2.7 Gender and Future Homes in Migration

As discussed in Chap. 1, home in migration has a strong temporal element. Some research has pointed to the ways in which the temporality of home is related to certain gendered aspects of the experiences of home. For example, Datta's (2008) research with Polish migrant men working in construction in London reveals how their plans to build houses in the future in Poland are highly significant to their constructions of home in migration. Though they have no immediate plans to return to Poland, the planned house is symbolic of a secure future home for either themselves or other family members and is bound up with their masculine identities as builders (Datta, 2008).

In our study with migrant men (international students and refugees) in Cork, Ireland (see Fathi & Ní Laoire, 2023), we realised that these young men had many aspirations towards fulfilling their dreams as outcomes of their migrations. All of them had invested a lot financially, emotionally and temporally to be in Cork at the time of the study (2019–2020). Some of them had been in the asylum process for almost a decade, others had spent tens of thousands of euros to obtain a degree from a western university to expand their human capital. However, the circumstances of their lives in Cork meant that their quest for a home was not achieved at the present time; they also did not think much of their homelands as homes anymore (they expressed no intention to return) but resorted to presenting their plans for making a home as a future endeavour. As such their active participation in either university or the labour market was viewed as a pathway to obtaining 'reputable' degrees, and/or saving up money to invest in purchasing a house, and ultimately, making a home of their own in the future. The most important aspect of these efforts at making a home, however, was the (gendered and heteronormative) aspiration to form a family with a wife and children (all identified themselves as heterosexual). Their aspiration to form such a family within a material home (thinking about or attempting at purchasing a house) and their imaginings about how it would look, where it might be (often outside Ireland), could be seen as a bridge towards a future place of belonging. This was envisaged as somewhere that would be different to their experiences of unbelonging and seclusion in Cork in the present time. A considerable number of the participants in this study did not feel a strong sense of belonging to Cork. Although they did not find much hostility as migrants, Cork as a city was not seen for them as a place they would call home; it was not homeland, and neither was it seen as cosmopolitan enough to secure a smaller network of likeminded people around them. Instead, as Fathi (2022, p. 1113) argues, for these young migrant men in Cork, the home in the present was only 'a vehicle to achieve a home in future'. She cites from

a participant, Ranit, an Afghan refugee who had just become an Irish citizen and whose family lived in Manchester in the UK, as follows:

Ranit: I want to go to Manchester. Me and my brother we want to set up a business. Like I worked here for a few years in a mobile phone shop. I'm good with technical stuff, like computers and—he [his brother] said, 'If you come there, we might open up a shop'. Plus he said, 'I could find you a wife there! (Fathi, 2022, p. 1113)

Here Ranit's masculine identity makes a link to this imaginary future home that will revolve around a business venture with his brother but also the possibilities of an arranged marriage there. What is important about Ranit's account is that he did not find his present (at the time of the interview) homely, even though structurally he was able to make a home. He was an Irish citizen (with an Irish passport) at the time and had full-time employment in a factory. What was missing in his life was the presence of significant others. Their absence impacted his sense of self, but also his masculine identity. The presence of a wife, in Ranit's family and culture is an important indicator of establishing a masculine identity, being a breadwinner and being heterosexual. This example reveals how gendered identity, home and temporality are intimately interwoven in migrant men's aspirations for home in the future: even if some aspects of home exist, there are still other factors needed to make a place feel like a home.

### 2.8 Conclusion

When discussing the intersection of gender and home, the politics of home that produce structural im/possibilities of home need to be considered. Home can include a variety of positive and negative meanings, a variety of traditional and alternative practices, and the reinforcement and reformation of gendered identities. This chapter looks at studies that examine changes, continuities and re-workings in traditional ways of understanding home and mobility through a gender lens. We examine how these hegemonic gendered understandings have shifted through the mobilities of men and women and how mobility and exposure to new value systems or new life opportunities have transformed migrants' relationships to home. Furthermore, the literature on migration, home and gender helps us to understand that masculinity is not necessarily equated to mobility or femininity to sedentariness. The processes of displacement and translocation add to the complexities that shape new meanings attached to home, homeland and imaginations of home in future, because they become intersected with systems of patriarchy, racialisation, discrimination and transnational class systems.

To conclude, the home space, traditionally seen as a feminised space, is constantly shifting in response to changing economic and social power relations before and after migration. It is evident from the literature that the economic power of one spouse (gender) can change the meanings of home in light of changing positions within domestic and public space. This gender dynamic is tightly linked to economic power relations, underlying relationships of race and ethnicity, relationships between migrants and their significant others and the restrictions as well as opportunities of migration regimes. Thus the structural im/possibilities of home in migration always involve complex entanglements of gender with other power relations.

In examining the literature on home and gender, it is evident that there are several areas that are in need of further research. The first is how experiences of home among migrant men are impacted by intersections of class, race, gender and sexuality in migration. Our research suggests that single migrant men in their endeavours to make a home often link home-making to their future homes, but how these imaginaries are shaped by class, race and sexuality needs further research. Secondly, given that there is still a dearth of research in relation to LGBTQ+ identities and home in migration, this field is in desperate need of further research (Thongkrajai, 2022; Fortier, 2001).

The impact of divorce and spousal splitting on migrant men and women in their experiences of home-making is another area that merits further research. Some research suggests for example that there is a link between relationship breakdown, return migration and ideas about home (Ní Laoire, 2008). This area could be explored in relation to pre- and post-migration home-making, relationship break-ups and divorces given that many migration regimes necessitate a marriage or official civil partnership proof for granting or continuation of different forms of visa and such separations may contribute to structural impossibilities of making a home after migration.

Finally, given the rising number of migrant women and men who leave families behind with the aim of providing for them, the process of making home when one is aging while still separated from families is another area for further research.

# References

- Ahmed, S. (1999). Home and away: Narratives of migration and estrangement. *International Journal of Cultural Studies*, 2(3), 329–347.
- Ahmed, S., Castañeda, C., Fortier, A. M., & Sheller, M. (Eds.). (2020 [2003]). Uprootings/ regroundings: Questions of home and migration. Routledge.
- Aitken, S. C. (2009). The awkward spaces of fathering. Ashgate.
- Anthias, F. (2008). Thinking through the lens of translocational positionality: An intersectionality frame for understanding identity and belonging. *Translocations: Migration and social change*, 4(1), 5–20.
- Ardener, S. (1981). Women and space: Ground rules and social maps. Croom Helm.
- Belford, N., & Lahiri-Roy, R. (2019). (Re)negotiating transnational identities: Notions of 'home' and 'distanced intimacies'. *Emotion, Space and Society*, 31, 63–70.
- Blunt, A., & Dowling, R. (2006). Home. Routledge.
- Boccagni, P., & Baldassar, L. (2015). Emotions on the move: Mapping the emergent field of emotion and migration. *Emotion, Space and Society*, 16, 73–80.
- Bonfanti, S. (2021). Diversities. In M. Nieto, A. Massa, & S. Bonfanti (Eds.), *Shifting roofs: Ethnographies of home and mobility* (pp. 115–140). Routledge.

- Boyce Davies, C. (1994). Black women, writing and identity: Migrations of the subject. Routledge.
- Brah, A. (1996). Cartographies of diaspora: Contesting identities. Routledge.
- Brandhorst, R., Baldassar, L., & Wilding, R. (2020). Introduction to the special issue: "Transnational family care 'on hold'? Intergenerational relationships and obligations in the context of immobility regimes". *Journal of Intergenerational Relationships*, 18(3), 261–280.
- Brettell, C. B. (2014). *Men who migrate, women who wait: Population and history in a Portuguese parish* (Vol. 470). Princeton University Press.
- Brickell, K. (2012). 'Mapping' and 'doing' critical geographies of home. *Progress in Human Geography*, 36(2), 225–244.
- Buckley, M. (1997). Sitting on your politics: The Irish among the British and the women among the Irish. In J. Mac Laughlin (Ed.), *Location and dislocation in contemporary Irish society* (pp. 84–132). Cork University Press.
- Carling, J. (2014). Scripting remittances: Making sense of money transfers in transnational relationships. *International Migration Review*, 48, S218–S262.
- Casimiro, S., Hancock, P., & Northcote, J. (2007). Isolation and insecurity: Resettlement issues among Muslim refugee women in Perth, Western Australia. *Australian Journal of Social Issues*, 42(1), 55–69.
- Charsley, K. (2005). Unhappy husbands: Masculinity and migration in transnational Pakistani marriages. *Journal of the Royal Anthropological Institute*, 11(1), 85–105.
- Christou, A., & Kofman, E. (2022). *Gender and migration: IMISCOE short reader*. Springer Nature.
- Collins, P. H. (2022). Black feminist thought: Knowledge, consciousness, and the politics of empowerment. Routledge.
- Constable, N. (1999). At home but not at home: Filipina narratives of ambivalent returns. *Cultural Anthropology*, *14*(2), 203–228.
- Cox, R. (2014). Working on masculinity at home. In A. Gorman-Murray & P. Hopkins (Eds.), Masculinities and place (pp. 227–240). Routledge.
- Crenshaw, K. (1991). Mapping the margins: Identity politics, intersectionality, and violence against women. *Stanford Law Review*, 43(6), 1241–1299.
- Cresswell, T., & Uteng, P. (Eds.). (2012). Gendered mobilities. Ashgate.
- Datta, A. (2008). Building differences: Material geographies of home(s) among Polish builders in London. *Transactions of the Institute of British Geographers*, 33, 518–531.
- De Backer, M. (2020). Street harassment and social control of young Muslim women in Brussels: Destabilising the public/private binary. *Journal of Gender Based Violence*, 4(3), 343–358.
- Delphy, C. (1984). Close to home: A materialist analysis of women's oppression. University of Massachusetts Press.
- Dodson, B. (2008). Gender, migration and livelihoods: Migrant women in Southern Africa. In N. Piper (Ed.), *New perspective on gender and migration* (pp. 137–158). Routledge.
- Donaldson, M., Hibbins, R., Howson, R., & Pease, B. (Eds.). (2009). Migrant men: Critical studies of masculinities and the migration experience. Routledge.
- Dyck, I. (2018). Migrant mothers, home and emotional capital Hidden citizenship practices. *Ethnic and Racial Studies*, 41(1), 98–113.
- Ehrkamp, P. (2013). 'I've had it with them!' Younger migrant women's spatial practices of conformity and resistance. *Gender, Place & Culture, 20*(1), 19–36.
- Enloe, C. (2007). Globalization and militarism: Feminists make the link. Rowman and Littlefield.
- Fathi, M. (2017). Intersectionality, class and migration: Narratives of Iranian women migrants in the UK. Palgrave Macmillan.
- Fathi, M. (2022). 'My life is on hold': Examining home, belonging and temporality among migrant men in Ireland. *Gender, Place & Culture, 29*(8), 1097–1120.
- Fathi, M., & Ní Laoire, C. (2023). Urban home: Young male migrants constructing home in the city. *Journal of Ethnic and Migration Studies*, *49*(3), 813–831.
- Fernández-Kelly, P. (2000). Reading the signs: The economics of gender twenty-five years later. Signs: Journal of Women in Culture and Society, 25(4), 1107–1112.
- Fortier, A. M. (2001). 'Coming home': Queer migrations and multiple evocations of home. *European Journal of Cultural Studies*, 4(4), 405–424.

- Fortier, A. M. (2020 [2003]). Making home: Queer migrations and motions of attachments. In S. Ahmed, C. Castada, A. M. Fortier, & M. Sheller (Eds.), *Uprootings/regroundings: Questions* of home and migration (pp. 115–136). Routledge.
- Fraser, N. (1994). After the family wage: Gender equity and the welfare state. *Political Theory*, 22(4), 591–618.
- Fraser, N. (2021). Gender, capital and care. In J. Browne (Ed.), *Why gender?* (pp: 144–169). Cambridge University Press.
- Fresnoza-Flot, A. (2009). Migration status and transnational mothering: The case of Filipino migrants in France. *Global Networks*, 9(2), 252–270.
- Gallo, E. (2006). Italy is not a good place for men: Narratives of places, marriage and masculinity among Malayali migrants. *Global Networks*, 6(4), 357–372.
- Gallo, E., & Scrinzi, F. (2016). *Migration, masculinities and reproductive labour*. Palgrave Macmillan.
- Ghorashi, H. (2010). From absolute invisibility to extreme visibility: Emancipation trajectory of migrant women in the Netherlands. *Feminist Review*, 94(1), 75–92.
- Gilmartin, M., & Migge, B. (2016). Migrant mothers and the geographies of belonging. *Gender*, *Place & Culture*, 23(2), 147–161.
- Giuffré, M. (2017). The transnational matrifocal home among Cape Verdean migrant women: The case of Santo Antão Island. In J. Lloyd & E. Vasta (Eds.), *Reimagining home in the 21st century* (pp. 57–71). Edgar Elgar.
- Gorman-Murray, A. (2008). Masculinity and the home: A critical review and conceptual framework. Australian Geographer, 39(3), 367–379.
- Gorman-Murray, A. (2014). Materiality, masculinity and home: Men and interior design. In A. Gorman-Murray & P. Hopkins (Eds.), *Masculinities and place* (pp. 209–226). Routledge.
- Gorman-Murray, A., & Hopkins, P. (Eds.). (2014). Masculinities and place. Routledge.
- Graham, M., & Khosravi, S. (1997). Home is where you make it: Repatriation and diaspora culture among Iranians in Sweden. *Journal of Refugee Studies*, 10(2), 115–133.
- Gray, B. (2004). Women and the Irish diaspora. Routledge.
- Guglielmi, S., Mitu, K., & Seager, J. (2021). "I just keep quiet": Addressing the challenges of married Rohingya girls and creating opportunities for change. *The European Journal of Development Research*, 35(5), 1232–1251.
- Hamada, I. (2017). Transnational domestic masculinity: Japanese men's home cooking in Australia. In M. Szabo & S. Koch (Eds.), *Food, masculinities and home: Interdisciplinary perspectives* (pp. 75–91). Bloomsbury.
- Hanmer, J., & Itzin, C. (Eds.). (2000). Home truths about domestic violence: Feminist influences on policy and practice: A reader. Psychology Press.
- Hochschild, A. R. (2000). Global care chains and emotional surplus value. In W. Hutton & A. Giddens (Eds.), On the edge: Living with global capitalism (pp. 130–146). Jonathan Cape.
- Hondagneu-Sotelo, P. (1994). *Gendered transitions: Mexican experiences of immigration*. University of California.
- hooks, b. (1990). Homeplace: A site of resistance. In *Yearning; Race, gender and cultural politics* (pp. 41–49). Turnaround.
- Joseph, G. (1981). The incompatible menage à trois: Marxism, feminism and racism. In L. Sargent (Ed.), *Women and revolution: A discussion of the unhappy marriage of Marxism and feminism:* A debate on class and patriarchy. South End Press.
- Kaplan, C. (1987). Deterritorializations: The rewriting of home and exile in western feminist discourse. *Cultural Critique*, 6, 187–198.
- Kofman, E., & Raghuram, P. (2015). Gendered migration and global social reproduction. Palgrave.
- Kuhn, A., & Wolpe, A. M. (1982). Feminism and materialism: Women and modes of production. Routledge and Paul.
- Longhurst, R. (2012). Feminist perspectives on home. In S. Smith (Ed.), *International encyclopaedia of housing and home* (pp. 158–162). Elsevier.
- Longhurst, R., Johnston, L., & Ho, E. (2009). A visceral approach: Cooking 'at home' with migrant women in Hamilton, New Zealand. *Transactions of the Institute of British Geographers*, 34(3), 333–345.

- Lourenço, I., & Cachado, R. (2012). Hindu transnational families: Transformation and continuity in diaspora families. *Journal of Comparative Family Studies*, 43(1), 53–70.
- Macey, M. (1999). Religion, male violence, and the control of women: Pakistani Muslim men in Bradford, UK. *Gender and Development*, 7(1), 48–55.
- Madianou, M. (2012). Migration and the accentuated ambivalence of motherhood: The role of ICTs in Filipino transnational families. *Global Networks*, 12(3), 277–295.
- Madianou, M., & Miller, D. (2011). Mobile phone parenting: Reconfiguring relationships between Filipina migrant mothers and their left-behind children. New Media & Society, 13(3), 457–470.
- Mahler, S., & Pessar, P. (2006). Gender matters: Ethnographers bring gender from the periphery toward the core of migration studies. *International Migration Review*, 40(1), 27–63.
- Massey, D. (2013 [1994]). Space, place and gender. Polity Press.
- Mazumdar, S., & Mazumdar, S. (2001). Rethinking public and private space: Religion and women in Muslim society. *Journal of Architectural and Planning Research*, 18(4), 302–324.
- McClintock, A. (1995). Imperial leather: Race, gender and sexuality in the colonial context. Routledge.
- McDowell, L. (1999). Gender, identity and place: Understanding feminist geographies. University of Minnesota Press.
- Meah, A. (2014a). Reconceptualizing power and gendered subjectivities in domestic cooking spaces. Progress in Human Geography, 38(5), 671–690.
- Meah, A. (2014b). Reconceptualising 'masculinity' through men's contributions to domestic foodwork. In A. Gorman-Murray & P. Hopkins (Eds.), *Masculinities and place* (pp. 191–208). Routledge.
- Mim, N. J. (2020). Religion at the margins: Resistance to secular humanitarianism at the Rohingya refugee camps in Bangladesh. *Religions*, *11*(8), 423–440.
- Naples, N. A. (2003). Feminism and method. Routledge.
- Ní Laoire, C. (2008). 'Settling back'? A biographical and life-course perspective on Ireland's recent return migration. *Irish Geography*, *41*(2), 195–210.
- Nititham, D. S. (2016). Making home in diasporic communities: Transnational belonging amongst Filipina migrants. Taylor & Francis.
- Oishi, N. (2005). Women in motion. Stanford University Press.
- Palenga-Möllenbeck, E. (2016). Unequal fatherhoods: Citizenship, gender and masculinities in outsourced 'male' domestic work. In B. Gullikstade (Ed.), *Paid domestic labour in a changing Europe: Questions of gender equality and gendered citizenship* (pp. 217–243). Palgrave Macmillan.
- Parreñas, R. (2008). The force of domesticity. New York University Press.
- Peinhopf, A. (2014). Ethnic minority women in Georgia Facing a double burden? European Centre for Minority Issues Working Paper. No.74. https://core.ac.uk/download/ pdf/71743364.pdf
- Percot, M. (2012). Transnational masculinity: Indian nurses' husbands in Ireland. *E-migrinter*, 8, 74–86.
- Pratt, G. (2012). Families apart: Migrant mothers and the conflicts of labor and love. University of Minnesota Press.
- Povrzanović Frykman, M., & Humbracht, M. (2013). Making palpable connections: Objects in migrants' transnational lives. *Ethnologia Scandinavica*, (43), 47–67.
- Ramirez, M., Skrbiš, Z., & Emmison, M. (2007). Transnational family reunions as lived experience: Narrating a Salvadoran autoethnography. *Identities*, 14(4), 411–431.
- Salih, R. (2003). Gender in transnationalism: Home, longing and belonging among Moroccan migrant women. Routledge.
- Samuri, M. A. A., & Hopkins, P. (2023). Navigating 'Purdah' culture in urban space: The restricted lives of young married Rohingya refugees in Malaysia. In *Refugee youth* (pp. 81–99). Bristol University Press.
- Sandu, A. (2013). Transnational homemaking practices: Identity, belonging and informal learning. Journal of Contemporary European Studies, 21(4), 496–512.
- Shahidian, H. (1999). Gender and sexuality among immigrant Iranians in Canada. Sexualities, 2(2), 189–222. https://doi.org/10.1177/136346079900200203

- Szabo, M. K. (2014). "I'm a real catch": The blurring of alternative and hegemonic masculinities in men's talk about home cooking. *Women's Studies International Forum, 44*, 228–235.
- Thongkrajai, C. (2022). Femininity in transition: Sex, gender, and sexuality experiences of Thai transgender migrants in Europe. *Advances in Southeast Asian Studies*, 15(2), 177–194.
- Ukwatta, S. (2010). Sri Lankan female domestic workers overseas: Mothering their children from a distance. *Journal of Population Research*, 27, 107–131.
- Väänänen, A., Kevin, M. V., Ala-Mursula, L., Pentti, J., Kivimäki, M., & Vahtera, J. (2005). The double burden of and negative spillover between paid and domestic work: Associations with health among men and women. *Women & Health*, 40(3), 1–18.
- Vacchelli, E., & Peyrefitte, M. (2018). From a/topia to topia: Towards a gendered right to the city for migrant volunteers in London. *Cities*, 76, 12–17.
- Vogel, L. (1983). Marxism and the oppression of women: Toward a unitary theory. Pluto Press.
- Wallace, I. (2012). From the geopolitical to the everyday: 'Home' for Muslim women in London and Bristol. Doctoral dissertation, Queen Mary University of London.
- Walsh, K. (2011). Migrant masculinities and domestic space: British home-making practices in Dubai. *Transactions IBG*, 36(4), 516–529.
- Walter, B. (2002). Outsiders inside: Whiteness, place and Irish women. Routledge.
- Werbner, P. (1999). Political motherhood and the feminisation of citizenship: Women's activisms and the transformation of the public sphere. In N. Yuval-Davis & P. Werbner (Eds.), *Women, citizenship and difference*. Zed Books.
- Wilkins, A. (2017). Gender, migration and intimate geopolitics: Shifting senses of home among women on the Myanmar-Thailand border. *Gender, Place & Culture*, 24(11), 1549–1568.
- Wojnicka, K., & Nowicka, M. (2022). Understanding migrant masculinities through a spatially intersectional lens. *Men and Masculinities*, 25(2), 232–251.
- Wong, M. (2014). Navigating return: The gendered geographies of skilled return migration to Ghana. *Global Networks*, 14(4), 438–457.
- Ye, J. (2014). Migrant masculinities: Bangladeshi men in Singapore's labour force. *Gender, Place & Culture, 21*(8), 1012–1028.
- Yeoh, B. S., & Khoo, L. M. (1998). Home, work and community: Skilled international migration and expatriate women in Singapore. *International Migration*, 36(2), 159–186.
- Yeoh, B. S. A., & Ramdas, K. (2014). Gender, migration, mobility and transnationalism. *Gender*, *Place and Culture*, 21(10), 1197–1213.
- Yeoh, B. S. A., & Willis, K. (2005). Singaporeans in China: Transnational women elites and the negotiation of gendered identities. *Geoforum*, 36, 211–222.
- Young, I. M. (2005). House and home: Feminist variations on a theme. In I. M. Young (Ed.), *On female body experience: Throwing like a girl and other essays*. Oxford University Press.

Yuval-Davis, N. (1997). Gender and nation. SAGE.

**Open Access** This chapter is licensed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license and indicate if changes were made.

The images or other third party material in this chapter are included in the chapter's Creative Commons license, unless indicated otherwise in a credit line to the material. If material is not included in the chapter's Creative Commons license and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder.



# Chapter 3 Age and Home in Migration



# 3.1 Introduction

This chapter brings the focus to the importance of age, generation and life-course transitions in understanding migrant home and homing processes. While there is an emerging interest in the ways in which practices and experiences of home in migration are inflected by age, generation and life-course transitions, to date this literature has not been integrated into a focused discussion on the implications of an age and life-course perspective for understandings of migrant home. It could be argued that much of the migration and home literature tends to be based on snapshots in time without considering the complexities of life-course transitions or intergenerational dynamics over time; additionally, it is usually based on research with adults without consideration of children's perspectives. We address these lacunae in this chapter by reviewing existing literature to provide an intersectional perspective on how age, life-course and generation shape experiences of home in migration. We focus in particular on the ways in which regimes of citizenship, migration and residency shape homing possibilities for migrants in different ways at different ages and phases of the life-course.

First, we explain how we conceptualise age and life-course before moving on to discuss the ways in which these are bound up with the doing and imagining of home in migration. The subsequent three sections explore respectively, via reviews of existing empirical studies, home in migration through the lenses of ageing/older age, young adulthood and childhood. In each section, we also pay attention to the ways in which age and life-course stage intersect with other axes of difference, in particular gender, class and migrant status, in shaping structural im/possibilities of home in migration. Through this review, we develop an argument about how migration regimes open up and close off possibilities of making and feeling at home for migrants depending on their age, generational and life-course positions.

# 3.2 Conceptualising Age, Generation and Life-Course

While it has long been recognised that age is a key factor in understanding migration experiences, there has been a shift in recent decades away from viewing migration as an 'age-specific' phenomenon and towards a life-course perspective to understand the relationships between migration and age or ageing (Ní Laoire & Stockdale, 2016). The life-course perspective emphasises the idea of a continuum of phases and transitions that occur over a lifetime. Key to this approach is the recognition that while life-course phases are shaped by socially constructed normative expectations and institutional forces regarding when and how key life events occur, life-courses do not necessarily conform to expectations and are both fluid and geographically and historically contingent (Elder, 1994; Heinz & Kruger, 2001; Konietzka & Kreyenfeld, 2021). A life-course perspective emphasises how people make sense of the socio-structural contexts in which their life-courses unfold and in which age-related life events take place. Therefore, our analysis of home is organised around migration experiences linked to specific age-related life-course phases and transitions that we understand as both socially constructed and socially embedded organizing principles through which people's biographies can be understood.

Taking a life-course perspective on migration is particularly valuable to understanding meanings of home as it recognises the temporal, biographical and relational dimensions to migration (Halfacree & Boyle, 1993; Kilkey & Ryan, 2021). It draws attention to the ways in which migrations and mobilities may be triggered by life-course transitions such as those related to the labour market or family formation, and how migration experiences are shaped by factors specific to different lifecourse stages (Gardner, 2009; Kobayashi & Preston, 2007). A life-course perspective can also enable a longitudinal and intergenerational perspective on migration, which is valuable in understanding how homing in migration contexts is a life-long process that is ongoing and never complete. Indeed, homing can be viewed as a search for temporal or biographical continuity in contexts of discontinuity (Boccagni, 2017). Expected life-course transitions and events can be interrupted by migration and the bordering practices of migration regimes, thus limiting the possibilities of making home and requiring adaptations and negotiations. Furthermore, migration and mobility can be strategies for finding or making home as people respond to changing circumstances associated with transitions through the life-course.

Closely connected to life-course and biographical perspectives on migration are the related concepts of generation and intergenerationality. Family relations play a crucial role in life-course and migration trajectories, and family is also intimately bound up with ideas of home (whether in positive, negative or ambiguous ways). Relations between different generations within the same family form a key part of the landscape of emotions, ties, imaginations and practices through which home-inmigration is constructed; hence a generational, or intergenerational, lens can shed further light on these processes, especially as they relate to age and life-course transitions. We pay particular attention to this aspect in our consideration of how young adults construct home in contexts of migration and im/mobility. The next section focuses on ageing and older age, while the subsequent sections focus first on young adulthood and then childhood. Selecting these particular lifecourse stages is not intended to imply that others, such as middle adulthood stages, are not important or worthy of exploration in terms of their implications for constructions of home in migration. Indeed, there is scope for more migration/home research that pays attention to the role of age and life-course in middle-adulthood stages. However, here we focus on older age and youth as useful lenses through which to tease out the ways in which age and life-course transitions intersect with migration-related social structures and other power relations in framing the structural im/possibilities of home in migration.

# 3.3 Ageing and Home in Migration

The phase, or phases, of life called 'old age', 'older age', 'ageing' or 'later life' relate to a wide range of life-course transitions involving physical ageing as well as changing relationships to work, family and wider society, although it is recognised that 'older age' is a very fluid and broad category that can be defined in many different ways (King et al., 2017). We do not attempt a definition of it here but instead draw on the meanings given to it in studies which explore migration and home. Older migrants have become a focus of attention in policy and literature, in recent decades, in particular as the large cohorts of labour migrants who settled in expanding industrial economies of the West in the latter half of the twentieth century have aged and transitioned from temporary migrants to more long-term and settled residents in their host societies.<sup>1</sup> In addition, as transnational migration flows have become more complex, migrating *in* older age is becoming a significant phenomenon, in the form of, for example, retirement migration, return migration and 'familyjoining' migration (King et al., 2017; Walsh & Näre, 2016). It is recognised that migrants have changing needs and priorities as they age, and there is particular emphasis by policy-makers and researchers on the importance of social policy in supporting older migrants (Wang & Zhan, 2021). However, there is somewhat less emphasis on the emotional and subjective aspects of life, such as feeling at home, for migrants in older age.

Some of the literature on older transnational migrants explores their intentions in relation to staying in their host societies or returning to countries of origin once they reach retirement age (Bolzman et al., 2006), this question itself pointing to normative expectations around the relationship between age/life-course and place (see also Ní Laoire, 2008). This question is fundamentally about the construction of

<sup>&</sup>lt;sup>1</sup>The use of the term 'migrants' to refer to people who may have been settled for a long time in their host society is questionable, but in the absence of an alternative term to refer to people who are living outside their country of origin, we use it in this chapter, with the caveat that we recognise it is not ideal.

home – where and what is home for people who have lived outside their countries of origin for a long time or people who migrate in older age? How do migrants make home in the context of ageing? Walsh and Näre (2016) point out that there has been little research on the concept of home from the perspective of transnational migrants in older age; their edited collection makes a significant contribution to filling this gap, along with a growing number of other studies exploring home and migration in older age, which are discussed in this section.

Popular discourses tend to associate older age with nostalgia (May, 2017) and sedentarism (King et al., 2017), suggesting that home for older migrants must be either 'back home' in the country of origin, or, to be found in an unchanging present location. However, many studies conducted with older transnational migrants challenge these popular presumptions. Instead, the research points to a number of key themes in older migrants' narratives of home, which suggest that as they age, and life circumstances change, their idea of home shifts and changes also. These key themes in narratives of home among older migrants, around which this section is structured, are: temporality and home; deepening home roots in place; transnational home; and finally, ambiguous attachments to homeland.

# 3.3.1 Temporality and Home-Making Among Older Migrants

Temporality and the passage of time are key to understanding processes of homing and home-making for older migrants, both those who have lived a large part of their lives as migrants and those who migrate in older age. The passage of time means that relationships, attachments, circumstances and priorities shift and change with time, but time also allows for accumulation of capital (social, economic) as well as formal rights and entitlements, and the establishment of emotional connections, relationships and families, that may increasingly tie one to a particular place or places, or provide the impetus to move in order to deepen one's place-based ties, that is, that allow one to strengthen a sense of home for older age.

Returning to the imagined 'homeland' or country of origin is one type of strategy undertaken by older migrants to secure home. However, this is often associated with a difficult process of re-making home and studies reveal ambivalent experiences of this process (Walsh, 2018). For some, the sense of loss for what they have left behind in their 'migrant' lives can be more pronounced than anticipated (Walsh, 2016). Ramji (2006) provides a more optimistic perspective. She shows that while Gujarati return-migrant retirees in India (from London) face disappointments and challenges on return, with time, they adjust and develop a sense of home that is located physically in Gujarat but also connected to London. They re-create home by holding on to aspects of their London lives, and continuously re-making their relationship to their 'homeland' (Ramji, 2006). Thus, home-making in older age is revealed as a continuous process of connecting past, present and future.

Deepening a sense of home that is emplaced within the host society is a strategy that is more feasible than return migration for most older migrants. Often, despite the strength of the dream of returning 'home', as time moves on this dream is relinquished due to the realities of weakening ties to past homes, together with the growing realisation that one's social and instrumental attachments are to the current place of residence (for example, Leavey et al., 2004). It seems that locality/place takes on more importance to older migrants' everyday lives, as they accumulate place-based ties over time and transition out of the labour force. As Ryan et al. (2021), point out, this does not mean that a sense of belonging in place is a given; on the contrary, it is constantly worked at through a process they term 'embedding', or developing attachments and belongings over time. Liu et al.'s (2021) study with older Chinese migrants in Australia, for example, explores the role of locality in their home-making processes, highlighting everyday routines in neighbourhoods, as well as the ways in which older migrants make houses and gardens meaningful as repositories of memory and expressions of ethnic identity, but also ways of becoming rooted in the new place. Their research illuminates how these spaces of gardens, homes and local neighbourhoods provide opportunities for older migrants to make home in ways that are firmly rooted in the 'here and now' but also connected temporally to other places and times. We explore this phenomenon of deepening home roots in the current place of residence in the next sub-section.

#### 3.3.2 Deepening Home Roots in Place Among Older Migrants

The 'here and now' of home-making for older migrants is bound up with the nature of their ties and attachments to the current place of residence, which, as migrants age, comes increasingly to mean home to them. This can be related to a number of factors, which we explore here. One of the most important is proximity to family. The importance of living close to grown-up children (and grandchildren) is a recurrent theme discussed by older migrants (those who have children) with regard to their decisions to either move to, or remain in, their host societies (for example, Buffel, 2017; Gardner, 1999; Ryan et al., 2021; Wang & Zhan, 2021). In other words, living close enough to immediate family members so that they are part of their everyday lives is a key element of feeling at home in that place, and, for many older migrants, home is simply where their children and grandchildren are. Co-presence enables older migrants to provide care to their children and grandchildren and to construct a shared sense of home in place, in an ongoing process of meaning-making with their family members (Liu et al., 2021). Home in this sense is about the familial relationships that structure and give meaning to everyday life, and the importance of co-presence to this. Co-presence is, of course, integrally bound up with migration/staying possibilities, as the possibility of living close to one's family is tied up with *freedom* to move or stay or to be joined by family members.

This points to the crucial role of immigration status and social class as intersectional power relations that produce differential access to transnational mobility and family reunification.

A second rationale commonly presented by older migrants for deciding to stay in the host society and make their home there is access to social security, welfare and healthcare services (Buffel, 2017). For example, Hunter's (2016) research highlights the key role played by access to healthcare and welfare in France for retired labour migrants, many of whom use instrumental narratives of home when explaining their decisions not to return permanently to countries of origin. Hunter (2016) distinguishes between these instrumental narratives of home (related to access to services) and the more emotional narratives of home used by participants when referring to family and community ties to countries of origin. However, Wang and Zhan (2021), in their research with older Chinese migrants in the US, bring the instrumental and the emotional together. They propose that instrumental decisions are also about *feeling* at home in the place where one's needs can be met – in other words, pragmatic considerations are important not just for meeting health and welfare needs, but because of the sense of security and belonging that comes from knowing these needs are being and will be met in the future. In this way, the work of both Hunter (2016) and Wang and Zhan (2021) highlights how home for older migrants needs to be understood in relation to the structural possibilities of material realities related to legal status, economic security and accumulation of statutory entitlements that become particularly important in later life. Such entitlements can accumulate over time for ageing migrants who, during the life-course, can become more settled in a host society, thus enabling structural possibilities of feeling secure, safe and cared-for, and hence of feeling at home. This gives added emphasis to the notion of spatial security being at the heart of homing in migration, connecting the emotional to the legal/instrumental aspects of feeling a secure sense of belonging in place.

Local co-ethnic communities are also a key aspect of feeling at home in the place of residence for older migrants (Ballantyne & Burke, 2017; Buffel & Phillipson, 2016; Meijering & Lager, 2014). While studies that access migrant participants through ethnic community organisations are likely to find that these communities are significant to their participants, and it is important not to generalise from these, their insights are nevertheless valuable and suggest that the connections within local ethnic communities that migrants actively foster can be central to the process of making home in the here and now. In fact, local demographic shifts which change the ethnic composition of neighbourhoods, leading to the disintegration of former ethnic communities, can be experienced by long-term migrant residents as a loss, undermining their sense of local belonging and contributing to a sense of dislocation, as found by Ryan et al. (2021) in their research with older migrants in England.

However, the lived experiences of migrant community life can be very different for men and women, pointing to gender as a differentiating factor intersecting with age and migration in shaping the im/possibilities of home for migrants and their family members. Gardner (1999) found that experiences of life in London were very different for the older Bengali men and women who were participants in her research. The female participants, who had arrived later than their male counterparts and who worked mostly in the home, were more affected by a sense of loss of the community-oriented life they had left behind in Bangladesh. Gender therefore was a key factor shaping the structural possibilities of feeling at home in London; for many older Bengali women, their role as 'home-makers' (in the conventional sense) meant that their lives were to some extent marked by isolation and the un-homeliness of the structures of family life and immigrant housing in London (Gardner, 1999). Similarly, Buffel's (2017) research with older Turkish migrants in Brussels points to the role of gender in shaping the possibilities of doing home for older migrants, in a context where men tend to have more access to neighbourhood public spaces than women and, hence, more possibilities for placemaking and developing the local attachments that are important to home-making for older migrants.

### 3.3.3 Transnational Home and Older Migrants

While ties in the place of residence (relating to family, neighbourhood, community and services) are key to feeling at home for older migrants, transnational dynamics and mobilities are also part of ongoing constructions of home for this group. Developing transnational living arrangements can be a strategy for meeting social, emotional and instrumental needs that are important for feeling at home, that change as one ages and that cannot be met in one place. For many people, retirement and related life changes can provide opportunities to live more transnational lives, whether that is in the form of retirement 'lifestyle' migration (often seasonal), seasonal migration to join family members who have migrated, or for existing migrants, more frequent and regular trips to the country of origin, or a more permanent return (Ciobanu & Hunter, 2017; Gustafson, 2008). For example, Tiaynen-Qadir (2016) refers to 'transnational grandmothers' who are key actors in transnational homemaking among families living between Russian Karelia and Finland, through their frequent cross-border movements and practices to sustain their families and for whom 'home' can be understood in transnational and mobile terms.

However, an overview of studies with older migrants who develop transnational mobilities in later life demonstrates the very different circumstances and levels of resources that distinguish different migration types and highlights the role of intersectional power relations and migration regimes in shaping the possibilities of transnational home-making. For example, there are important differences, rooted in migration regimes, between the circumstances of affluent older lifestyle migrants maintaining dual homes and those older migrants for whom transnational living arrangements are a manifestation of a precarity that continues into later life and that limits the possibilities of feeling at home.

For example, middle-class retirees moving from northern to southern Europe can often afford to maintain homes (and hence social and emotional ties) in two countries (Gustafson, 2008; Huber & O'Reilly, 2004). According to Huber and O'Reilly (2004), for British and Swiss retirement migrants in Spain, who often spend part of

the year back in the country of origin, the ability to maintain these social and family connections to their countries of origin is a key factor in feeling at home in Spain. In a different context, Chi-Yan Sun's (2016) research with Taiwanese ageing return migrants shows that the predominantly middle-class backgrounds of the participants had enabled them to retire comfortably to Taiwan while continuing to visit their children in the US regularly, thus enabling a comfortable sense of home through ongoing transnational mobility.

On the other hand, a number of studies seem to suggest that for ageing migrants from working-class or low-income backgrounds, while return migration or seasonal circulation might be desired, it is often not considered to be feasible due to lack of economic resources or reliance on social security entitlements that tie one to one place of residence (Hunter, 2016; Leavey et al., 2004). In these circumstances of spatial immobility, the structural possibilities of home may be more restricted, requiring more work to create a sense of home. Some studies refer to the sense of security that ageing migrants gain from having lived in the same place for a long time (Buffel & Phillipson, 2016; Leavey et al., 2004), key aspects of feeling at home. This may be particularly significant to ageing migrants of limited means, for whom a transnational move would be a risky strategy. In this way, older migrants of limited means may experience immobility and have fewer (structural) possibilities of constructing a continuous sense of home that is connected to a place of origin. Thus, the role played by social class and its intersection with ageing and migration regimes cannot be overlooked.

# 3.3.4 Attachments to 'Homeland' in Older Age

For economic, personal or family reasons, many older transnational migrants decide not to return, or do not even consider returning, to their country of origin. A realisation that return is not feasible is highlighted in some studies as part of the process of ongoing home-making in the place of current residence – returning 'there' is unlikely and as a result there is more emotional investment in making 'here' home (Buffel, 2017). However, this in itself may require a process of adjustment for older migrants, reassessing relationships to place of origin and place of residence and managing the tensions between their attachments to both as they work at homemaking in later life. For example, in Ballantyne and Burke's (2017) research with older Irish migrants in Australia, participants expressed both a sense of instrumental attachment to Australia and symbolic or emotional connection to Ireland, articulated in an expressed but not-realised longing to return 'home' to Ireland.

Gardner (2002) is particularly illuminating on the important symbolic role of narratives of attachment to 'homeland' for older migrants who remain in the host society. She shows how older Bengali migrants in the UK maintain a narrative of longing for the homeland, even though their personal or social connections to the

country of origin may be quite weak, given the length of time since their original migration, and return is no longer a realistic possibility. Gardner (2002) argues further that the shared narrative of longing for the homeland plays a crucial role in identity construction in the present, bound up with being part of a Bengali community in London and as a way of connecting to cultural memory and expressing dissatisfaction with aspects of life in the UK.

Buffel's (2017) research, with Turkish migrants in Brussels, finds that the maintenance of ties both to co-ethnic communities in the host society and to country of origin are very important to older migrants. A strong sense of localised social (and familial) capital in Brussels as well as a persistent connection to Turkey enable the older migrants to feel at home in Brussels while also maintaining a transnational sense of home. The latter is facilitated by the ease of transnational mobility in their case. Paradoxically, it is this ability to travel and keep in touch frequently that enables them to make their lives and homes in Brussels. Therefore, in very different ways and contexts, both Gardner's (2002) and Buffel's (2017) studies point to the importance having a relationship of some kind with the 'homeland' or country of origin as an integral element of making one's home in the host society, a suggestion which also finds resonance in research with transnational retirement migrants (Huber & O'Reilly, 2004).

This point is supported by Liu et al. (2021) in their study with older Chinese migrants in Australia, in which they argue that being able to integrate aspects of their old and new lives is a central aspect of making home in the new context. The enduring connection to 'there' makes it possible to make one's home 'here'. However, while this bifocal dynamic is a way of negotiating tensions between here and there and finding a sense of home, it can also be a constant reminder of unfulfilled longing to return for some (Ballantyne & Burke, 2017). As the bifocal dynamic is not equally available to all, older migrants of more limited means, who are by necessity tied to a less mobile lifestyle, have fewer (structural) possibilities of constructing a continuous sense of home that is tied to a place of origin. Pérez Murcia (2023) poignantly demonstrates how a sense of home is profoundly disrupted for people ageing in transnational families when geographical distance and migration circumstances make it impossible to ensure co-presence for end-of-life and death-related family obligations and expectations to be fulfilled.

Maintaining transnational ties in older age is also a gendered, as well as classed, phenomenon. Fesenmayr's (2016) research with older Kenyan female migrants in London points to normative gendered expectations regarding their responsibilities towards family and kin in the place of origin – meaning that return is deferred or avoided. This means that life is constantly on hold as they do not view London as home, though they are also reluctant to return to where 'home' is (Fesenmayr, 2016). Thus, intergenerational and gendered tensions in transnational families regarding financial and caring responsibilities shape structural possibilities of home for older migrants.

# 3.3.5 Summary: Structural and Intersectional Im/Possibilities of Home for Older Migrants

This review of literature points to the importance of connections and capital, relating to family, locality, community, the state and the transnational dimension, in enabling a sense of spatial security for older migrants in their host societies. Having a sense of material security while being close to, and having everyday connections with, family and others within the local neighbourhood, including co-ethnic communities, and while not losing important transnational connections, together play a key part in feeling at home in the place of residence. For some, this sense of spatial security is further bolstered by the ability to maintain more than one home and to move frequently between them. Regardless of the circumstances, as Walsh and Näre (2016) argue, home is very much about social relationships, whether these are localised or transnational, and for older migrants, the nature of these relationships and connections may shift as they age. Contrary to popular discourses that associate older age with stasis, the literature on older migrants demonstrates the ongoing and dynamic nature of homing in later life. Hence home-making, especially for older migrants, is an ongoing process of rethinking and negotiating identities and relationships.

Feeling at home is realised through a sense of belonging linked to personal histories in place/s as well as the legal and socio-economic status that permits the imagining of future lives in that place. Most of the empirical studies discussed above have been conducted with older migrants who have some kind of citizenship or legal status in their place of residence. This is likely to be a function of the passage of time, as migrants accumulate rights and entitlements over the life-course, though may also reflect a gap in the literature in relation to meanings of home for older migrants with precarious legal status. The security afforded by legal status in older age is clearly an important aspect of being able to create a sense of home in migration. For example, although not focused on the notion of home, research by Benson (2020) reveals the implications of legal insecurity for older migrants, using the case of the post-Brexit situation of British migrants in France, pointing out that some have the resources with which to navigate the uncertainties while others do not and are left in precarious situations – indicating that there is potential for further research which explores how older migrants in situations of legal/rights insecurity construct home.

Finally, the possibilities of feeling at home in migration in older age are integrally bound up with socio-spatial structures relating to age, gender, social class, race/ethnicity and residence status, as these variably open up or close down possibilities in places of residence for feeling safe and secure, for co-presence with families and social networks and for maintaining valuable transnational and transcultural ties. Changing circumstances in older age can limit the possibilities of feeling at home as not all older migrants have the same access to capital and resources with which to manage emerging tensions surrounding 'home'. Therefore, home as a sense of spatial belonging and security is not equally available to all older migrants; instead, it can be elusive or problematic, and should be understood within the context of intersectional power relations and migration regimes.

# 3.4 Home, Migration and Young Adulthood

Powerful assumptions about what home is, and where it is located, are closely bound up with questions of 'the right time' in one's life to be mobile or not. While western normative ideas tend to seek to fix the very young and very old in place, young adults are expected to be on the move and free of place-based ties. Of course, the social meanings of 'childhood', 'young adulthood' and 'older age' vary significantly from context to context, and transitions associated with these life stages are geographically, socially and historically contingent, diverse, and much more complex in reality than the norms and expectations that surround them (for example Harris et al., 2020). The rest of this chapter draws on existing literature to challenge normative ideas about home and migration in both childhood and young adulthood, by exploring how home is understood, constructed and experienced by young people living in contexts of transnational migration, focusing firstly on young adulthood before moving on in the next section to childhood.

### 3.4.1 Youth, Mobility and Home

As Thomson and Taylor (2005) articulate, young people navigate questions about home and leaving/staying while they are also navigating the journey towards the kinds of adults they want to be and becoming gendered and classed selves – processes of identity formation that are often bound up with questions of place attachment, belonging and im/mobility. Young adulthood is also a time of societal expectation, as societies impose their ideals of future full adulthood on young people, who are expected to orient towards becoming their future selves, in a sense putting the 'here and now' in a liminal state and putting certain human needs on hold.

In western contexts, it can be argued that 'youth' has become normatively constructed as a life-stage associated with mobility (King, 2018). The future achievement of personal and career success has become constructed, through neoliberal discourse, in terms that are associated with the idealisation of a kind of hypermobility, linked to normative constructions of 'youth' or young adulthood (Yoon, 2014). This neoliberal construct has been critiqued in studies which have explored how young people actually navigate and experience the competing tensions and pressures in their lives as they contemplate or undertake migration. In particular, these critiques highlight the need to foreground the important role of social relations, emotions, place belonging and place embeddedness in young people's lives (Allen & Hollingworth, 2013; Cheung Judge et al., 2020; Ní Laoire, 2020). In other words, young adults are not 'unmoored' beings who can bounce from place to place accumulating experiences, but social beings who have ties and relationships to people and places, in the form of resources which they carry with them and with which they carve out a sense of home as they move (Botterill, 2014; Harris et al., 2020). We discuss this theme further here, engaging with studies that explore how young adults create a sense of home both in and against mobility. We explore the diverse meanings of home and migration in young adulthood, recognising it as a period when questions of home, migration and im/mobility take on particular significance, focusing first on the concept of 'leaving home' and secondly on 'home in mobility'.

# 3.4.2 Young Adulthood and 'Leaving Home'

Decisions about migration/staying are often triggered by aspects of transitions to adulthood, such as decision-making in relation to transitions to the labour market, higher education or family formation, or by wider questions of identity and belonging. As young people navigate these significant life transitions (for example, as they transition out of the school system) they consider where home is, what home means, and whether or not to migrate from 'home'. Here we discuss this literature, focusing on the pre-migration phase as one that brings questions of migration and home into sharp focus in a particular way.

Literature on migration/staving decisions often focuses on young people growing up in economically disadvantaged, geographically peripheral or rural areas in the Global North, where there may be a 'mobility imperative' (Corbett, 2007) or economic/social pressures to migrate (Bjarnason & Thorlindsson, 2006). Another area of literature focuses on a broader mobility imperative associated with discourses of cosmopolitanism and neoliberal competition in which 'migration experience' is idealised as a qualification for entry into successful professional adulthood for all young people (Holdsworth, 2017; Ungruhe, 2010). Both of these sets of pressures bring the question of 'home' into sharp relief for young people as they consider whether to 'leave home' or where their homes in the future might be located. Such dilemmas are often conceptualised in terms of a tension between social/spatial mobility aspirations on the one hand and attachments to place/home, on the other (Cairns, 2013; Thomson & Taylor, 2005). For example, Cairns' (2013) research demonstrates the tensions in young people's narratives of the future between imaginaries of modest homes 'in place' and imagined futures that take them elsewhere. Studies such as those by Stockdale and Haartsen (2018) and Forsberg (2017) highlight the agency involved in electing to stay, to make and sustain homes in contexts where leaving is associated with success. It can be said then that these types of tensions frame the contexts in which young people develop narratives of home and belonging and within which they make decisions about, or enact, trajectories of mobility and immobility.

What emerges most potently from the body of research with young adults facing 'leave or stay' choices are the emotional struggles inherent in this dilemma, in which young adults find themselves at the coalface of contradictions embedded in their life-worlds, contradictions reflected in different visions of what and where 'home' is now, was in the past, or will be in the future. At the heart of these dilemmas are questions about identity and social relations; home is not just about where one belongs, but 'who I am', a negotiation that is influenced by prevalent local and global discourses of youth, gender, social class, sexuality and other social and cultural identities, and is navigated in the context of social relations such gender and social class in which young people are involved.

Allen and Hollingworth (2013, p. 513) draw on the concept of place-based and classed habitus, referring to the structural conditions that shape the possibilities of aspiration and desire, to demonstrate how 'localised sets of material, social and imagined relations are central in producing young people's sense of place in the world and their possibilities of mobility'. Their concept of the 'sticky effect' of social class illustrates how young people's dispositions and desires relating to home and mobility cannot be easily disentangled from the (im)possibilities thrown up by their localised social class positions. Indeed, research suggests that there is a strong association between staying and working-class or low-income status (Corbett, 2007), and some studies point to the role of factors such as the safety of home or family obligations in this. In other words, there is a sense in which electing migration/mobility is potentially too risky and costly for those with limited mobility capital, and therefore they hold onto, and invest in, the sense of home that already exists, or that is more secure, by not migrating (Ní Laoire, 2020). Migration/leaving can produce or increase precarity; this risk is not universal but is highly classed and is intimately connected to meanings of home. In addition, contrary to popular constructs of young people as independent of family ties, the sense of home that is 'at risk' is usually relational – it is formed through ties of care, love and obligation that young people have to family members, ties that are place-based, gendered, and cannot easily be stretched across space in contexts of limited resources.

Young people growing up in *migrant* families, or with familial migration histories, occupy a very particular position in this dynamic of im/mobility and homemaking. For these young 1.5, second or even third generation adults, whose parents or grandparents were migrants, the experience can be one of growing up in one place ('home') and simultaneously having a sense that the familial home is elsewhere. In other words, intergenerational and transnational relations form a key part of the landscape of homing and home-making. For example, Nikielska-Sekula's (2021) research with Norwegian Turkish second-generation youth in Norway finds that in different ways, they identify with 'home' in both Norway and Turkey; she develops the idea of *transdimensional home* to capture how different dimensions of home are more or less intense in the different places deemed to be 'home'. She argues that this transdimensional and transnational way of constructing home is not about connecting past and present lives (as it is for first-generation migrants) but about different dimensions or aspects of what makes home being experienced in relation to here and there simultaneously (Nikielska-Sekula, 2021).

For some young descendants of immigrants, who are navigating the emotional and complex landscape of home between the place in which they grew up and their parental or ancestral 'homeland', their life journeys may even involve a 'return' migration to the parental/ancestral 'homeland' in an effort to make their home there. A rich body of research has explored this phenomenon, labelled second-generation return migration or ancestral return migration, or even 'homecomings' (King & Christou, 2011; Tsuda, 2009). What is striking about this phenomenon is its foregrounding of the desire for home, a home that might be physically distant from where one grew up but is emotionally, symbolically and perhaps socially and culturally felt to be very close. 'Returning' to an ancestral homeland can be viewed as an expression of enduring connections across generations within families. In fact, Ní Laoire (2023) argues that we need to pay more attention to intergenerational im/ mobility legacies in transnational families. Meanings of home are made, passed on and transformed between different generations in transnational families; young adults in such families form their own attachments and home journeys but do so against the backdrop of family migration histories, transnational family entanglements and the lived experiences of growing up in im/migrant households.

In other words, social class, geography, migration regime, generation and gender all play a role in shaping the meanings and structural possibilities of home, vis-à-vis im/mobility, for young adults contemplating migration or staying 'at home' or constructing home in contexts of family migration histories. And importantly, by focussing on the meanings of home for young adults, our attention is drawn to the material, social and emotional dimensions of their lives, challenging prevailing discourses that construct young people as detached, free-floating, flexible workers and consumers. Young people are of course also agentic in mobilising the resources available to them, as classed and gendered beings, in working towards shaping and finding their sense of home in the world. This can be by staying 'at home' geographically while being mobile in different ways – for example, commuting, travelling, forging new global and local connections or identifying with cosmopolitan identities (Thomson & Taylor, 2005; Yeh, 2014). For many more young adults, a sense of home is sought through migration (either by leaving the childhood home, seeking another home, returning home, remembering home or by home-making on the move) which we explore in the next section.

# 3.4.3 Young Migrants and Home in Mobility

Given the globally interconnected nature of the world in which transnational youth migration occurs, conventional notions of 'leaving home' are no longer adequate to capture how home is understood and constructed by young migrants on the move. Cheung Judge et al. (2020) draw our attention to the under-recognised role of emotion in young people's mobilities and the complex entanglements of emotion, relating to both past and present, that characterise the lives of young transnational migrants. The concept of home can help to illuminate young migrants' emotional

worlds, drawing attention to the less visible relational and affective dimensions of youth migration. While youth mobilities often disrupt and complicate normative assumptions that associate home with settlement and immobility, a significant body of research points to the importance of different, sometimes unexpected, ways of feeling and doing home among young migrants on the move, challenging conventional understandings of home.

For example, youth migration can disrupt expected life-course transitions relating to home-making. According to Harris et al. (2020), mobility complicates notions of the right time and right place to settle, form a family and make a home. Contemporary transnational migration is characterised not so much by linear trajectories of youthful mobility followed by settlement, but by fragmented and diverse transitions tied up with more complex migration journeys (King, 2018). This can often mean that gendered heteronormative expectations are not easily fulfilled as young migrants face challenges in establishing and maintaining social and intimate relationship across borders and long distances and in contexts of transience and uncertainty. In this context, Harris et al. (2020) argue that young transnational migrants' engagements with places may not be about long-term integration in specific places, but about wider networks of relationships that are developed and maintained across and through places. This idea coheres with conceptualisations of home as a transnational or mobile construct, suggesting that for many young migrants, home may be 'made' in mobility, as they move, often alongside peers, maintaining meaningful ties to, and through, different places, often virtually, and constructing a feeling of home that is not about place embeddedness in the sense of deeply rooted local lives, but about dynamic transnational and local relationships.

Ahmed (1999) draws attention to the complexities of the migration-home nexus, highlighting the ways in which migration can be a movement towards home, or a way of being at home that is not fixed or rooted, in contrast to normative notions of migration as leaving home. In this sense, while home as a concept denotes feelings of belonging, security and familiarity, these qualities are not necessarily found through fixed rootedness in place. For example, Parutis (2011) conducted research with Polish and Lithuanian migrants aged 18-40 in Britain, and argues that, unlike families with children, these young migrants do not look for stability or physical comfort in the present and do not attach great value to the material culture of home. They value the freedom of their mobile lives. However, even though they avoid settlement in one particular place, according to Parutis (2011), they still find ways of making unhomely temporary accommodation more homely, such as through developing friendships with flatmates. This points to the importance of interpersonal relationships as well as a sense of freedom in cultivating a feeling of home. 'Freedom' in this sense refers to the ability to change life decisions as circumstances change (Parutis, 2011) and indicates the value young migrants place on agency and autonomy. Similarly, Damery (2021), in her research with young adults from migrant families in Brussels, points to the young people's agency in homemaking: importantly, she argues that the young people were not always willing to accept the choices that were being made for them by others that in effect would have limited their possibilities of feeling at home. The young people created spaces of belonging and family-like relationships for themselves despite the lack of stability of their migrant and family statuses (Damery, 2021). Home for young migrants like those in Parutis' (2011) and Damery's (2021) research then can be understood in terms of a meaningful space in which they can develop interpersonal relationships and make life decisions on their own terms.

However, forming intimate relationships and making home in transnational mobility is not easy and requires resources and mobility capital that are not equally available to all young migrants. The task of home-making can be challenging, elusive and uncertain. Fathi's (2022) research exploring constructions of home among young migrant men (aged 18–35) in Ireland finds that, for many, home is something that is 'on hold' due to the temporary and transitory nature of their migrant status and socio-economic positions in the host society. Particularly in the case of young men who have refugee status, their life-course transitions have been interrupted and disrupted by an asylum process which puts their lives on hold for a long time. This means that 'home' is something that does not exist in a meaningful way in the present but instead is an ideal to strive for in the future. Similarly, Sirriyeh's (2016, p. 13) research with young refugee women in Britain (aged 16-25) explores the intersections of life-course transitions (related to young adulthood) and forced migration experiences, highlighting how life-course time becomes 'stretched, contracted and otherwise constructed in the migration process'. Her research shows how the young women work to create homely spaces even in contexts of disruption and instability but also draws attention to the costs of this disruption.

These costs can be understood through the lens of structural im/possibilities of home, intersectionality and disrupted life-course transitions. Being a young adult while going through the disruption and uncertainty of an asylum process can mean that expected life-course events such as finding a life partner, settling down and forming a family are postponed, while simultaneously ties to families of origin may be severed. Moreover, these family and life-course disruptions can continue after achieving refugee/settled status due to the legacy of the asylum process in terms of its disembedding and isolating effects (Howlett-Southgate, 2021). Gendered heteronormative expectations are not fulfilled and home is often envisaged as something in the future, that is, it represents a hope and an ideal rather than something that exists in the here and now. Kim and Smets' (2020) research with young-adult Syrian refugees in a housing project in the Netherlands points to the importance to them of the private domestic space for creating a sense of home that is origin-oriented, by enabling them to recreate familiar surroundings and practices. However, although this sense of home is transnationally connected to other Syrians, it does not seem to extend to wider Dutch society (Kim & Smets, 2020). Similarly, in our research in Ireland, young migrant men experienced a paucity of opportunities for meaningful connections to others locally and for spaces of care that could provide them with the spatial security that could make a sense of home in the city and wider society possible (Fathi & Ní Laoire, 2023). These impossibilities of home must be understood in the context of migration histories that have temporally disrupted lives and intimate relationships during crucial youthful life-course transitions as well as the racialised exclusion and segregation of refugees in European cities. In this way,

states seek to control migration through strategies of temporality, disrupting young migrants' expected life-course transitions, denying them spatial security in the present and delaying future home. Thus, the structural im/possibilities of home for young migrants are closely linked to intersections of migration regime, age and race.

The ways in which migration regimes intersect with structures of class, gender, race and age to circumscribe the possibilities of home differently for different groups of young migrants are complex. However, some studies provide further glimpses of how these complexities manifest. For example, in their research on international students' constructions of home, Wu and Wilkes (2017) find that for some students, the host society cannot feel like home because of their experiences of racism and cultural dislocation there, while for others, home is felt to be wherever in the world they want it to be, an outlook that indicates a degree of privilege and freedom not available to all young migrants. The significance of meaningful local and city-scale feelings of attachment among young-adult migrants, particularly those in privileged positions, contrasts sharply with the young refugees in Kim and Smets' (2020) research above. For example, Prazeres' (2018) research with Canadian international students in Global South cities reveals how they form a sense of home in their host societies through their attachments to local neighbourhoods and the city itself. Prazeres (2018) clearly positions this urban/local sense of home as a function of these young people's privilege as white western international students in being able to comfortably feel at home in a new place.

### 3.4.4 Summary

Bringing the focus onto young-adult migrants in the discussion of migration and home is valuable as it draws our attention to the multiple meanings of home and its fluid and shifting nature. It also reveals how particular ideas about home can become normative, in ways that are bound up with expectations about young adulthood, mobility and migration. These norms and expectations form an integral element of the structural im/possibilities of home in migration. Migration regimes seek to facilitate the hypermobility of young people, as part of neoliberal regimes, and supported by assumptions about young adulthood being a life-stage when home is not a priority. Thus, the temporalities of migration regimes are revealed, firstly in the ways in which they are imbricated with biographical time and secondly, how many young migrants become caught in protracted waiting in migration, that is, immobility in migration, thus delaying their possibilities of home-making.

The literature reveals how certain aspects of what makes home become particularly significant in the context of youth migration, such as, the importance of autonomy over one's life choices and the maintenance of interpersonal relationships across long distances while on the move. Research with young migrants themselves shows how they challenge some of the normative assumptions about youth, mobility and home/homing. Young migrants in very different contexts, through their practices, challenge expectations of what home should be – whether that is by leaving childhood homes, or choosing to stay despite pressures to leave, or choosing between family and asylum status (Damery, 2021), making homes in situations of uncertainty, maintaining familial relationships virtually, or in the emphasis they place on emotional over material qualities of home.

#### 3.5 Childhood, Migration and Home

Holloway and Valentine (2000), over two decades ago, pointed out that normative Western constructions of childhood have contributed to powerful assumptions that the best place for children is 'at home' (understood as a domestic familial space) and that the existence of mobile, nomadic or homeless children has often resulted in moral panics. Inherent in this construct is the notion that the ideal childhood is one that is marked by immobility and rootedness in place through a singular domestic home. However, this notion has been challenged by research that has emerged over the past two decades to bring child-centred and social constructivist perspectives to bear, drawing also on the mobilities and transnational turns in social science, to understandings of childhood, home, mobilities and migration (Dobson, 2009; Knörr, 2005; Moskal, 2023; Ní Laoire et al., 2010; Tyrrell & Kallis, 2017; White et al., 2011). Some of this literature demonstrates the many ways in which 'normal' childhoods are marked by mobility and migrancy and reveals how children construct complex and fluid notions of home in these contexts. In doing so, this body of literature highlights the importance of taking a child-centred perspective in any study of home and migration. Children are social beings who are active agents in shaping their own social worlds and constructing their own sense of home, in ways that are often different to those of adults. However, despite this, most literature on home and migration is based on research conducted with adults (Moskal, 2015; Damery, 2021). Because of the nature of childhood itself, children's worlds are not easily entered or grasped by adult researchers or stakeholders. Therefore, to avoid an adult-centric view of home and migration, we need to look to studies that have attempted to engage with children on their own terms if we are to understand what home means to children who have migrated or whose worlds are marked by migrancy and transnational mobility.

# 3.5.1 Children and Home-Making in Migration: Agency and the Everyday

Over the pasts two decades, many researchers have engaged directly with children who have experienced migration or mobility, to explore their constructions of home and belonging (for example, Ní Laoire et al., 2011; Tyrrell et al., 2019). A key insight of this body of research is, first, to highlight the creativity and agency of

migrant children's complex constructions of home and belonging. Thus, the active nature of home-in-migration as an ongoing process for children is emphasised – children do not simply inherit or absorb a given sense of home but they actively construct it, and they do so in contexts of different structural im/possibilities of which they are more aware than they are given credit for. Second, many authors also agree that everyday social and familial relations and materiality are central to how home is constructed and experienced by children in contexts of migration/mobilities. As argued by Ní Laoire et al. (2010, p. 159), "'home' emerges as a concrete site of social relations and practices, involving familial and other social relations, daily practices and materiality, often conducted and experienced across territorial boundaries". We explore both of these insights further by looking at three importance themes in relation to how children construct home in the context of different migration-related structural im/possibilities: family, materiality and the everyday/ small-scale.

First, across numerous studies, and perhaps unsurprisingly, family is emphasised as a particularly important site of belonging at the very heart of what home means to children (for example, McGovern & Devine, 2016). For children who migrate with family members, family may replace 'home' as the main pillar of identity after migration. Christopoulou and de Leeuw (2005) point to the role of family as the main mediator between past and present, between old and new worlds, as migrant children negotiate shifting meanings of home. This highlights the particular difficulties experienced by migrant children living in contexts of enforced family separations. However, children can and do develop strategies to cope with such ruptures. For example, Zhang's (2015) research with children in China who are separated by internal migration from family members points to the children's resourcefulness in constructing a sense of home on the move, for example, through their work to maintain family ties despite physical separations or to create new family-like connections. According to Zhang (2015), the emotional importance of a sense of home to these children cannot be overstated; even though the household structure may be broken, the children maintain a feeling of home through new and different spaces and practices of familial belonging.

Second, the importance of material objects (see also Chap. 4) takes on a particular significance to constructions of home among children who have experienced sharp ruptures in their lives due to migration. The importance of keeping and displaying family photographs as a way of maintaining happy memories of past lives and of distant family members is highlighted in some research with migrant children (Christopoulou & de Leeuw, 2005). Often there is an understandable reluctance to talk about the past among refugee families (Chase, 2010). Christopoulou and de Leeuw's (2005) research shows that photos are a way for children of keeping it alive (even while being silent); they write about what they term an 'economy of memories' (p. 120) where only those memories of past lives that are useful in the present are remembered. In other words, contrary to popular belief, children are often very aware of the structural im/possibilities of home as shaped by the migration realities of their lives, and they adjust their constructions of home accordingly. Hatfield's (2010) research with children in elite migrant families reveals that they do not invest a lot of emotion in physical places but in more portable objects and practices that make home on the move. In these ways, the children adjust their understanding of home in response to the knowledge that their place of residence is temporary and that future places of residence may also be temporary, and in doing so, demonstrate their capacity to grasp the material (structural) possibilities and impossibilities of their lives as children in migrant families.

Finally, Hatfield (2010) makes a particular case for recognising the importance of the everyday and small-scale for understanding what home means in the social worlds of children who move. She argues that the small-scale, mobile and transient nature of home-making among children means that adult-centred ways of understanding home-in-migration cannot suffice for theorising about children's worlds. She pays particular attention to the ways in which the migrant children use material objects and domestic practices to make and transport home (Hatfield, 2010). She finds that children are active participants in home-making and that they use material objects (such as soft toys) and domestic practices (such as eating together as a family) that are easily transportable from one home to another as they migrate as a family. In this way, these small-scale material objects and practices enable them to maintain an element of continuity between homes while also adapting to the changes brought about by migrating (Hatfield, 2010). Focusing on the materialities and everyday social practices through which home comes into being for children draws our attention to the immediate surroundings, the daily routines and the social connections that make home/s even as they are stretched across space (Mand, 2010). In other words, the every-day of migrant children's social worlds in their contextspecificity, their child-ness and their materiality, even though such worlds are often transient, provide the material with which they construct home, and they do so on the basis of their competence as social agents who have an understanding of the structural im/possibilities of home for them as children in mobile transnational families.

# 3.5.2 Multi-scalarity and the Politics of Home in Migrant Childhoods

Therefore, even though children's constructions of home are often grounded in the small-scale and the everyday, this does not mean that children's home-making processes are insignificant, or local-only, or limited to the micro-scale; on the contrary, they reflect at a micro scale the macro-scale processes that are shaping migrant childhoods and being shaped by them. Indeed, homing in migrant childhoods is recognised as a multi-scalar and complex process, in which children's constructions of home and belonging involve complex navigations at the domestic, local, national and transnational scales (Den Besten, 2010; Mand, 2010).

The small-scale and everyday relations and practices that make home for children are intimately interconnected with wider local, national and transnational frames of reference, and this is particularly the case for children who grow up in contexts of transnational migrations and families. This understanding also challenges the hegemonic Western tendency to tie notions of childhood home to the domestic sphere only and highlights the different sites and scales of belonging that are meaningful to children. For example, in different contexts, both Ní Laoire et al. (2011) and Vathi and King (2021) explore how children in return-migrant families navigate and disrupt hegemonic *national* constructs of return migration as 'homecoming', which are contradicted by their everyday experiences of more ambivalent belongings at *local* level in the presumed 'homeland'. Other studies illustrate how migrant children actively draw on frames of belonging at different scales. For example, White (in Ní Laoire et al., 2011) illustrates how African/Irish children blend domestic, local and transnational spaces in the making of a home-space that connects them into transnational family networks. Mand (2010) similarly sheds light on the complexities and multiscalarity of children's constructions of home in a transnational context. Her research with children in transnational Bengali families between London and Bangladesh illustrates the translocal nature of home for the children, as they locate their homes as being in both London and in their family's village in Sylhet, and as such, linked to familial ties and relationships that are lived out through concrete social practices in both places. Their constructions of home reflect their own age-specific and generation-specific attachments to London as home, though these may be different from those of their parents, while simultaneously maintaining familial connections to a home elsewhere (Mand, 2010). In other words, they form complex constructions of home that reflect the realities of their translocal lives and that do not conform to normative (or parental) expectations of a singular homeplace or homeland.

In this way, children-centred research draws attention to children's small-scale negotiations of, and resistances to, normative, adultist or hegemonic constructs of home and emphasises how they respond to the ways in which migration regimes seek to limit their possibilities of home. Even in apparently unhomely and precarious living conditions, children are active players in doing and making home. Beazley's (2000) research with street children in Indonesia shows how these children challenge national/state ideals of home and family by leaving and often rejecting their family homes, which are often experienced by them as oppressive. Instead they construct multiple homes on the move, including spaces attached to streetchild subcultures where they have strong and meaningful connections and investments. In this way, they make home in very difficult and dangerous circumstances, and according to Beazley (2000), in doing so they subvert dominant state and familial ideological constructions of the ideal childhood home. Research such as this highlights children's agency and creativity in shaping home and at the same time resisting narratives of what a home should be, in particular resisting narratives that marginalise them. McDonnell (2021) makes a further valuable contribution here. Based on her research with migrant children in Ireland, she argues that while their home-making narratives and practices may appear on the surface to be small-scale and unimportant, they can also be a site of everyday politics and negotiation of possibilities of belonging. Through a case-study of one child living in the Direct Provision system in Ireland, that is, in an apparently and intentionally unhomely institution (see also Howlett-Southgate, 2021), she shows how he makes home through forging connections and attachments that are meaningful to him, in this way, challenging practices of spatial exclusion that seek to limit his possibilities of making a home (McDonnell, 2021).

#### 3.5.3 Summary

Emerging literature on migrant childhoods presents an important challenge to scholarship on migration and home. That challenge is: how can the existing scholarship move beyond adult-centrism and take children's migration worlds seriously? We argue that this is both essential and possible. Engaging with children's social and cultural worlds sheds a different light on how home is constructed, experienced and done in migration. It draws attention to the everyday and small-scale nature of how home gets done, in its materiality, its apparent banality and its interpersonal nature. But this does not mean that home from this perspective is only banal or small-scale – it is through these practices and connections that the wider structural im/possibilities of home in migration are lived and experienced by migrants, children and families. Migrant children actively challenge exclusionary migration regimes and adult expectations by forging and sustaining the ties and connections that mean home to them. Therefore, they are key players in the doing of home in migration and the specificities of their homing and home-making experiences need to be acknowledged.

# 3.6 Conclusions

Drawing on the existing literature from across the different life-courses stages discussed in this chapter (older age, young adulthood and childhood), a number of key points can be made about the relationships between age, home and migration. The first is that it is clear that both the emotional/subjective and the material/practical are important aspects of home for all migrants – however, each one takes on different significance, and in different ways, at different stages of the life-course and migrant journey. Second, feeling at home in migration is a constant process. It is through a life-course perspective that this processual and life-long aspect of home in migration becomes clear - it is never finished but is constantly worked at through childhood and into older age, usually involving a negotiation of tensions between the emotional, the social and the material/practical aspects. Third, the possibilities of feeling at home in migration are integrally bound up with age and life-course as socio-spatial structures. In addition, the ways in which these intersect with gender, social class, race/ethnicity and residence status means that the possibilities of home in migration are contingent on migrants' complex social positioning, of which age and life course are crucial components that should not be overlooked.

These im/possibilities of home are also bound up with migration regimes; migration regimes treat migrants differently depending on age, and in turn are experienced differently at different life-stages. These dynamics open up and close down possibilities of feeling and doing home, whether through, for example, the im/possibilities of maintaining multiple transnational homes in older age, or of building a future home during young adulthood, or for children of re-constructing a sense of home in the face of disruptions. Therefore, home as a sense of spatial belonging and security is not equally available to all migrants who may share a social positioning in terms of age; instead, it can be elusive or problematic for many, and should be understood within the context of age-related intersectional power relations and migration regimes.

The overview of literature suggests a few areas for further research. While much migration research tends to focus on adult migrants in middling age groups and lifecourse stages, this research does not always consider the role of age/life-course in their experiences of home. Therefore there is space for research that explores how norms and expectations of middle-adulthood life-stages are tied up with the structural im/possibilities of home in migration. Secondly, more research is needed on home in migration that incorporates children's perspectives, but without siloing them. In other words, we have much to learn from children's perspectives that span long time-periods within families or migrant communities or networks could add considerably to existing understandings of how home is constructed and reconstructed in migration over time and across generations.

# References

- Ahmed, S. (1999). Home and away: Narratives of migration and estrangement. *International Journal of Cultural Studies*, 2(3), 329–347.
- Allen, K., & Hollingworth, S. (2013). 'Sticky subjects' or 'cosmopolitan creatives'? Social class, place and urban young people's aspirations for work in the knowledge economy. Urban Studies, 50(3), 499–517.
- Ballantyne, G., & Burke, L. (2017). "People live in their heads a lot": Polymedia, life course, and meanings of home among Melbourne's older Irish community. *Transnational Social Review*, 7(1), 10–24.
- Beazley, H. (2000). Home sweet home? Street children's site of belonging. In S. Holloway & G. Valentine (Eds.), *Children's geographies: Living, playing, learning* (pp. 194–212). Routledge.
- Benson, M. (2020). Brexit and the classed politics of bordering: The British in France and European belongings. Sociology, 54(3), 501–517.
- Bjarnason, T., & Thorlindsson, T. (2006). Should I stay or should I go? Migration expectations among youth in Icelandic fishing and farming communities. *Journal of Rural Studies*, 22, 290–300.
- Boccagni, P. (2017). Migration and the search for home: Mapping domestic space in migrants' everyday lives. Palgrave Macmillan.
- Bolzman, C., Fibbi, R., & Vial, M. (2006). What to do after retirement? Elderly migrants and the question of return. *Journal of Ethnic and Migration Studies*, *32*(8), 1359–1375.

- Botterill, K. (2014). Family and mobility in second modernity: Polish migrant narratives of individualization and family life. *Sociology*, 48(2), 233–250.
- Buffel, T. (2017). Ageing migrants and the creation of home: Mobility and the maintenance of transnational ties. *Population, Space and Place, 23*(5), e1994.
- Buffel, T., & Phillipson, C. (2016). Constructions of 'home' among first-generation migrants living in Belgium and England. In *Transnational migration and home in older age* (pp. 63–74). Routledge.
- Cairns, K. (2013). The subject of neoliberal affects: Rural youth envision their futures. *The Canadian Geographer/Le Geographe Canadien*, 57(3), 337–344.
- Chase, C. (2010). Agency and silence: Young people seeking asylum alone in the UK. *The British Journal of Social Work*, 40(7), 2050–2068.
- Cheung Judge, R., Blazek, M., & Esson, J. (2020). Transnational youth mobilities: Emotions, inequities, and temporalities. *Population, Space and Place*, 26(6), e2307.
- Chi-Yan Sun, K. (2016). Changing notions of home across the life-cycle: How ageing Taiwanese return migrants rethink their relationship with the homeland. In *Transnational migration and home in older age* (pp. 126–138). Routledge.
- Christopoulou, N., & de Leeuw, S. (2005) Children making media: Constructions of home and belonging. In J. Knorr (Ed.), *Childhood and migration: From experience to agency* (pp. 113–135). Transcript Verlag.
- Ciobanu, R. O., & Hunter, A. (2017). Older migrants and (im)mobilities of ageing: An introduction. *Population, Space and Place, 23*, e2075.
- Corbett, M. (2007). *Learning to leave: The irony of schooling in a coastal community*. Fernwood Publishing.
- Damery, S. (2021). Home without family, family without home: Young migrants' experiences of home and relationships in the city of Brussels. In M. Montero-Sieburth et al. (Eds.), *Family* practices in migration (pp. 103–120). Routledge.
- Den Besten, O. (2010). Local belonging and 'geographies of emotions': Immigrant children's experience of their neighbourhoods in Paris and Berlin. *Childhood*, 17(2), 181–195.
- Dobson, M. E. (2009). Unpacking children in migration research. *Children's Geographies*, 7(3), 355–360.
- Elder, G. H. (1994). Time, human agency, and social change: Perspectives on the life course. Social Psychology Quarterly, 57(1), 4–15.
- Fathi, M. (2022). 'My life is on hold': Examining home, belonging and temporality among migrant men in Ireland. *Gender, Place & Culture, 29*(8), 1097–1120.
- Fathi, M., & Ní Laoire, C. (2023). Urban home: Young male migrants constructing home in the city. *Journal of Ethnic and Migration Studies*, 49(3), 813–831.
- Fesenmayr, L. (2016). Deferring the inevitable return 'home': Contingency and temporality in the transnational home-making practices of older Kenyan women migrants in London. In K. Walsh & L. Näre (Eds.), *Transnational migration and home in older age* (pp. 115–125). Routledge.
- Forsberg, S. (2017). 'The right to immobility' and the uneven distribution of spatial capital: Negotiating youth transitions in northern Sweden. *Social & Cultural Geography*, 20(3), 323–343.
- Gardner, K. (1999). Narrating location: Space, age and gender among Bengali elders in East London. *Oral History*, 27(1), 65–74.
- Gardner, K. (2002). Age, narrative and migration: Life histories and the life course amongst Bengali elders in London. Berg.
- Gardner, K. (2009). Lives in motion: The life-course, movement and migration in Bangladesh. *Journal of South Asian Development*, 4(2), 229–251.
- Gustafson, P. (2008). Transnationalism in retirement migration: The case of North European retires in Spain. *Ethnic and Racial Studies*, *31*(3), 451–475.
- Halfacree, K. H., & Boyle, P. J. (1993). The challenge facing migration research: The case for a biographical approach. *Progress in Human Geography*, *17*(3), 333–348.

- Harris, A., Baldassar, L., & Robertson, S. (2020). Settling down in time and place? Changing intimacies in mobile young people's migration and life courses. *Population, Space and Place*, 26(8), e2357.
- Hatfield, M. E. (2010). Children moving 'home'? Everyday experiences of return migration in highly skilled households. *Childhood*, 17(2), 243–257.
- Heinz, W. R., & Krüger, H. (2001). Life course: Innovations and challenges for social research. *Current Sociology*, 49(2), 29–45.
- Holdsworth, C. (2017). The cult of experience: Standing out from the crowd in an era of austerity. *Area*, 49(3), 296–302.
- Holloway, S., & Valentine, G. (2000). Children's geographies: Playing, living, learning. Routledge.
- Howlett-Southgate, R. (2021). *Life after Asylum: Investigating the precarious geographies of home with refugees leaving Direct Provision in Cork, Ireland.* Unpublished PhD thesis, University College Cork.
- Huber, A., & O'Reilly, K. (2004). The construction of Heimat under conditions of individualised modernity: Swiss and British elderly migrants in Spain. Ageing & Society, 24(3), 327–351.
- Hunter, A. (2016). Emotional or instrumental? Narratives of home among North and West African seniors in France. In *Transnational migration and home in older age* (pp. 75–86). Routledge.
- Kilkey, M., & Ryan, L. (2021). Unsettling events: Understanding migrants' responses to geopolitical transformative episodes through a life-course lens. *International Migration Review*, 55(1), 227–253.
- Kim, K., & Smets, P. (2020). Home experiences and homemaking practices of single Syrian refugees in an innovative housing project in Amsterdam. *Current Sociology*, 68(5), 607–627.
- King, R. (2018). Theorising new European youth mobilities. *Population, Space and Place,* 24, e2117.
- King, R., & Christou, A. (2011). Of counter-diaspora and reverse transnationalism: Return mobilities to and from the ancestral homeland. *Mobilities*, 6(4), 451–466.
- King, R., Lulle, A., Sampaio, D., & Vullnetari, J. (2017). Unpacking the ageing–migration nexus and challenging the vulnerability trope. *Journal of Ethnic and Migration Studies*, 43(2), 182–198.
- Knörr, J. (Ed.). (2005). Childhood and migration: From experience to agency. transcript Verlag.
- Kobayashi, A., & Preston, V. (2007). Transnationalism through the life course: Hong Kong immigrants in Canada. Asia Pacific Viewpoint, 48(2), 151–167.
- Konietzka, D., & Kreyenfeld, M. (2021). Life course sociology: Key concepts and applications in family sociology. In *Research handbook on the sociology of the family* (pp. 73–87). Edward Elgar Publishing.
- Kreyenfeld, D., & Schneider, N. (Eds.). (2021). *Research handbook on the sociology of the family*. Edward Elgar Publishing.
- Leavey, G., Sembhi, S., & Livingston, G. (2004). Older Irish migrants living in London: Identity, loss and return. *Journal of Ethnic and Migration Studies*, 30(4), 763–779.
- Leira, A., & Saraceno, C. (Eds.). (2008). *Childhood: Changing contexts* (Comparative social research) (Vol. 25). Emerald Group Publishing Limited.
- Liu, S., Liu, Y., & Wang, S. (2021). Constructing a sense of home: Older Chinese migrants ageing in a foreign land. *Journal of Sociology*, 57(3), 485–500.
- Mand, K. (2010). 'I've got two houses. One in Bangladesh and one in London... Everybody has': Home, locality and belonging (s). *Childhood*, 17(2), 273–287.
- May, V. (2017). Belonging from afar: Nostalgia, time and memory. *The Sociological Review*, 65(2), 401–415.
- McDonnell, S. (2021). Narrating homes in process: Everyday politics of migrant childhoods. *Childhood*, 28(1), 118–136.
- McGovern, F., & Devine, D. (2016). The care worlds of migrant children Exploring intergenerational dynamics of love, care and solidarity across home and school. *Childhood*, 23(1), 37–52.
- Meijering, L., & Lager, D. (2014). Home-making of older Antillean migrants in the Netherlands. Ageing & Society, 34(5), 859–875.

- Montero-Sieburth, M., Mas Giralt, R., Garcia-Arjona, N., & Eguren, J. (Eds.). (2021). Family practices in migration. Routledge.
- Moskal, M. (2015). 'When I think home I think family here and there': Translocal and social ideas of home in narratives of migrant children and young people. *Geoforum*, 58, 143–152.
- Moskal, M. (2023). Migration and home in research with children and young people: Story, participation, agency. In P. Boccagni (Ed.), *Handbook on home and migration* (pp. 411–425). Edward Elgar Publishing.
- Ní Laoire, C. (2008). 'Settling back'? A biographical and life-course perspective on Ireland's recent return migration. *Irish Geography*, *41*(2), 195–210.
- Ní Laoire, C. (2020). Transnational mobility desires and discourses: Young people from returnmigrant families negotiate intergenerationality, mobility capital, and place embeddedness. *Population, Space and Place, 26*(6), e2310.
- Ní Laoire, C. (2023). Young people, intergenerationality and the familial reproduction of transnational migrations and immobilities. In J. Waters & B. Yeoh (Eds.), *Handbook on migration and the family* (pp. 118–134). Edward Elgar Publishing.
- Ní Laoire, C., & Stockdale, A. (2016). Migration and the life course in rural settings. In M. Shucksmith & D. Brown (Eds.), *Routledge international handbook of rural studies* (pp. 36–49). Routledge.
- Ní Laoire, C., Carpena-Méndez, F., Tyrrell, N., & White, A. (2010). Introduction: Childhood and migration – Mobilities, homes and belongings. *Childhood*, 17(2), 155–162.
- Ní Laoire, C., Carpena-Méndez, F., Tyrrell, N., & White, A. (2011). Childhood and migration in Europe: Portraits of mobility, identity and belonging in contemporary Ireland. Routledge.
- Nikielska-Sekula, K. (2021). Towards a transdimensional home: Home-making in the narrations of Norwegians of Turkish descent. *Journal of Housing and the Built Environment*, 36(4), 1483–1499.
- Parutis, V. (2011). "Home" for now or "home" for good? Home Cultures, 8(3), 265-296.
- Pérez-Murcia, L. E. (2023). 'The sweet memories of home have gone': Displaced people searching for home in a liminal space. *Journal of Ethnic and Migration Studies*, 45(9), 1515–1531.
- Prazeres, L. (2018). Unpacking distinction within mobility: Social prestige and international students. *Population, Space and Place*, 25(5), e2190.
- Ramji, H. (2006). British Indians 'returning home': An exploration of transnational belongings. Sociology, 40(4), 645–662.
- Ryan, L., Kilkey, M., Lőrinc, M., & Tawodzera, O. (2021). Analysing migrants' ageing in place as embodied practices of embedding through time: 'Kilburn is not Kilburn any more'. *Population, Space and Place,* 27(3), e2420.
- Shucksmith, M., & Brown, D. (Eds.). (2016). Routledge international handbook of rural studies. Routledge.
- Sirriyeh, A. (2016). Inhabiting borders, routes home: Youth, gender, asylum. Routledge.
- Stockdale, A., & Haartsen, T. (2018). Editorial introduction: Putting rural stayers in the spotlight. *Population, Space and Place, 24*(4), e2124.
- Thomson, R., & Taylor, R. (2005). Between cosmopolitanism and the locals: Mobility as a resource in the transition to adulthood. *Young*, *13*(4), 327–342.
- Tiaynen-Qadir, T. (2016). Transnational grandmothers making their multi-sited homes between Finland and Russia. In *Transnational migration and home in older age* (pp. 25–37). Routledge.
- Tsuda, T. (Ed.). (2009). *Diasporic homecomings: Ethnic return migration in comparative perspective*. Stanford University Press.
- Tyrrell, N., & Kallis, G. (2017). Children in transnational family migration. In C. Ní Laoire, A. White, & T. Skelton (Eds.), *Movement, mobilities, and journeys*. Springer.
- Tyrrell, N., Sime, D., Kelly, C., & McMellon, C. (2019). Belonging in Brexit Britain: Central and Eastern European 1.5 generation young people's experiences. *Population, Space and Place*, 25(1), e2205.
- Ungruhe, C. (2010). Symbols of success: Youth, peer pressure and the role of adulthood among juvenile male return migrants in Ghana. *Childhood*, *17*(2), 259–271. https://doi.org/10.1177/0907568210365753

- Vathi, Z., & King, R. (2021). Memory, place and agency: Transnational mirroring of otherness among young Albanian 'returnees'. *Children's Geographies*, 19(2), 197–209.
- Veale, A., & Doná, G. (Eds.). (2014). Child and youth migration: Mobility-in-migration in an era of globalization. Palgrave Macmillan.
- Walsh, K. (2016). Expatriate belongings: Traces of lives 'abroad' in the home-making of English returnees in later life. In *Transnational migration and home in older age* (pp. 139–152). Routledge.
- Walsh, K. (2018). Materialities and imaginaries of home: Geographies of British returnees in later life. Area, 50(4), 476–482.
- Walsh, K., & Näre, L. (Eds.). (2016). Transnational migration and home in older age. Routledge.
- Wang, Q., & Zhan, H. J. (2021). The making of a home in a foreign land: Understanding the process of home-making among immigrant Chinese elders in the US. *Journal of Ethnic and Migration Studies*, 47(7), 1539–1557.
- Waters, J., & Yeoh, B. (Eds.). (2023). *Handbook on migration and the family*. Edward Elgar Publishing.
- White, A., Ní Laoire, C., Tyrrell, N., & Carpena-Méndez, F. (2011). Children's roles in transnational migration. *Journal of Ethnic and Migration Studies*, 37(8), 1159–1170.
- Wu, C., & Wilkes, R. (2017). International students' post-graduation migration plans and the search for home. *Geoforum*, 80, 123–132.
- Yeh, D. (2014). New youth mobilities: Transnational migration, racialization and global popular culture. In A. Veale & G. Doná (Eds.), *Child and youth migration: Mobility-in-migration in an era of globalization* (pp. 91–115). Palgrave Macmillan.
- Yoon, K. (2014). Transnational youth mobility in the neoliberal economy of experience. *Journal of Youth Studies*, 17(8), 1014–1028. https://doi.org/10.1080/13676261.2013.878791
- Zhang, N. (2015). Home divided, home reconstructed: Children in rural-urban migration in contemporary China. *Children's Geographies*, 13(4), 381–397.

**Open Access** This chapter is licensed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license and indicate if changes were made.

The images or other third party material in this chapter are included in the chapter's Creative Commons license, unless indicated otherwise in a credit line to the material. If material is not included in the chapter's Creative Commons license and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder.



# **Chapter 4 Materialities of Home in Migration**



# 4.1 Introduction

Contemporary debates on home in migration have taken a new exciting path towards understanding the material aspects of home, although the intersection of migration studies and material culture studies requires further research. Basu and Coleman (2008) argue that objects form an important part of migratory journeys. Here we focus on understanding the materiality of migrant homes and the relationship between objects and migration as a personal as well as a social process. Understanding the importance of objects, particularly for migrants who have limited means to physically move objects with them across transnational borders (such as refugees and undocumented migrants), is important and there is much room for exploration here (Neumark, 2013; Giorgi & Fasulo, 2013).

It is important to stress that a home is not just about its material and structural boundaries (as this book overall attempts to demonstrate), but, acknowledging the deeper meanings in relation to home, we focus here on the physical facets of home that present themselves as vital in relation to its social and symbolic components. By doing this, the intention is not to isolate the space of home in migration from its wider social order or structures of im/possibilities; instead, the chapter considers how material aspects of home have meaning in relation to memory and nostalgia, everyday life and identity categories of gender, age and race, among others and in relation to social and emotional structures of migrants' lives. Materialisation here means the process of translating meaning into objects and structures and, vice versa, using objects by attributing to them significant underlying meanings beyond their immediate physical and instrumental use. This chapter presents a discussion of the existing scholarship on materiality of migrant home, first highlighting the importance of objects in migration narratives before moving on to present a categorisation of objects in migrant home-making around which the rest of the chapter is structured.

## 4.2 Importance of Objects in Migrants' Lives

Woodward (2001, p. 132) argues that 'the aspiration people have for their home, and their ideal ways of presenting and talking about their home and the objects inside it, are just as important as how they might actually live in their home'. The reason this relationship (between the notion of home and objects) is important is because the structure of a dwelling or other physical space where practices of home-making take place cannot be conceived of without the objects that occupy and constitute the space and, if we are interested in the spatial meanings of a home, then objects play a vital role in this interplay. Miller (1998) argues that our social worlds are constituted through materiality. As we live with an array of objects, and at some stages, we decide what to do with things that surround us, we are making decisions to re/construct our homes that impact our sense of who we are. Miller (1998) argues that practices of decoration, using ornaments or specific items to transform an inhospitable dwelling into a meaningful place, can have deep impacts on our core beings and sense of belonging that are central to identities and the sense of who we are.

We understand materiality to refer to a variety of tools, aesthetic objects, ornaments and everyday goods such as food that can sustain routine life, give meaning and provide comfort (both mental and physical) to a person. This chapter offers a discussion about the materiality of migrants' lives and its significance in understanding home in the contexts of mobility and resettlements.

Pérez Murcia and Boccagni (2022, p. 590) recently endeavoured to present four functions that objects have in a migrants' home: '1) embodying collective backgrounds and identities, 2) affording migrants to feel at home, 3) encapsulating their biographical memories and ties, 4) eliciting connections with settings and events that meant 'home' over their life course'. This categorisation, based on their field-work with Ecuadorian migrants in Italy, although useful in understanding the different functions that objects can have, particularly when migrants settle in host societies, remains somewhat limited in scope. There are many similarities across the different categories. For example, categories one and four are very similar and it is not clear how collective backgrounds and identities can be separated from events and settings that have been shared with others.

It is clear from the literature on materiality in home that the use of objects is linked to three major and overarching factors. Firstly, the majority of the literature in this area links past lives including biographies to larger cultural and social contexts of past homes and argues in depth that these relationships are encapsulated in objects. The second aspect is that objects that are taken with migrants have an everyday use aspect. They are functional and instrumental in the everyday lives and home-making practices of migrants. In this sense the significance of objects in the migrant home lies in their practicality and their role in facilitating everyday life as a migrant. Finally objects that are related to individualised identities such as class and gender are used as vehicles of identification and recognition.

We all like to define spatial territories around us to designate what belongs to us and not to others, demarcating the boundaries of home and outside. An important aspect of what marks these spaces as one's own are the physical markers through activities as well as use and repurposing of tools and objects that allow us to 'use' but also to 'control' our environments. This territorial interpretation of home (Després, 1991) is directly linked to the sense of satisfaction from owning a space. Additionally, the desire to modify a place that seems 'out of character' or empty, by giving meanings to specific aesthetic properties around the home is a universal experience. This is what Becker (1977) calls 'personalization'. Personalisation as a practice of ownership (Wells, 2000) and attachment is a gendered act, particularly in turning non-home environments into some form of home (see Dinç, 2009). This phenomenon is discussed later in the chapter in relation to refugee camps and the empty and soul-less spaces of shelters and other migrant accommodations. Here we focus on another type of object personalisation, that is, the role of materiality in mobile life.

Tolia-Kelly (2004) in her research with South Asian women shows that material culture is important in shaping diasporic homes by evoking pre-migration memories and the status associated with lives before displacement. By talking about photos, and meaningful objects that moved with Indian women migrants in the UK, she shows how identities of self and other were constructed through materiality and 'lifestyle' (Tolia-Kelly, 2004). Here materiality in mobile lives refers to how different forms of objects are used in the structures of homes as part of everyday life. In what follows, we present a framework for understanding the role of materiality in migrants' home lives.

Whilst much of the current literature discusses the importance of objects of the past (Christou & Janta, 2019; Edwards, 2010; Fathi, 2020; Golovina, 2018), there is still much to be learned about the material traces migrants leave behind and what they take and carry with them along their journeys to be used in their immediate and future lives. Objects can serve an instrumental purpose as well as having deeper meanings and can convey messages about the collectivities and individual lives of which they have been part. The following categorisation by no means aims to place each study into one of the three overarching categories we identify above but instead to elucidate the distinct significance of each one. As such the rest of this chapter is divided into three sections:

- Objects and memory (linking past to present)
- Objects and instrumentality (practicality, ordinariness, everyday use, mostly linked to present lives)
- Objects of identity: classed, gendered, national identities (lifestyles, consumption patterns and visions about future life)

### 4.3 Objects, Memories and Nostalgia

The first category refers to objects that capture symbolic historical meanings. Several scholars show that artefacts can evoke strong emotions (Navaro-Yashin, 2009; Svašek, 2008). Objects of memory refer here to those items that are not used

on an everyday basis, but on reflection, they mean more than a mere tool or an ornament and there is a particular biographical history attached to that object. They evoke a variety of feelings, that are remembered, cherished and reflected upon with the work of memory. As such these objects are always imbued with emotional attachments that are associated with a particular event, ceremony, relationships or person(s) from the past. These objects could evoke individual or collective memories fuelled by reflective nostalgia, glorifying the past homes. Emotions generate knowledge about oneself and one's surroundings (society) (Svašek, 2012), which can establish, sustain or challenge power relations, influencing one's subjectivity. As such objects of memory, due to their emotional load, are very important in the process of home-making after displacement.

Encapsulating memories of past lives, and in particular of relationships with 'dear ones', we try to keep creating links to what makes us feel wanted, belonged, warm and emplaced, such as to objects that remind us of these connotations. This is particularly true for forcedly displaced people for whom longing and belonging become an everyday emotional experience. These warm feelings can be individual or collective. Turan (2010) focuses on the significance of objects for the Palestinian diaspora from a psychological and material perspective. She discusses how objects help to form collective identity and memory. Turan (2010) calls these objects (that bring a sense of collectivity) 'objects of legacy' and argues that they provide a form of 'transition' between homeland and the lives of migrants of first generation, but turn into 'objects of legacy' for the following generations as the direct link between the migrant and homeland does not exist in younger generations, but objects remain an important part of this legacy (Turan, 2004).

So, some objects provide a 'sheltering and nurturing' environment in diaspora groups (Turan, 2010, p. 53), but we need to remember that not all memories of home are as positive and nurturing due to past events that can bring a sense of loss, detachment and sorrow. However, what people tend to 'keep' as a method of survival are those objects that bestow a happy feeling, a positive attribution and those that create a strong sense of attachment to important people such as the mother, what Winnicott (2001) called transitional objects: transitioning from one mode of attachment to another, replacing their primary care giver with an object of comfort. Here, objects of memory facilitate the environment by helping the transition from one home to another. As such this first category is related to those material tools that elicit and create emotional connections between us and places, settings, as well as collectivities and significant others that give a feeling of home to one's life. It is noteworthy that the literature, in discussing materialities of migration, tends to focus on objects of memory that link migrants' lives strongly to their past homes, homeland and transit spaces. One of the areas that needs fresh investigation is how objects of the present shine a light on the current lives of migrants.

Objects and material structures act as bridges between the home now and the home in the past. The bridge is the work of memory. In relation to how cities retain elements of past systems such as colonisation, Ley (2004, p. 155) argues that memories of home are recharged through preservation of such material fields but at the same time they transfer home into 'unfamiliar and distant territory'. We also argue

that memory objects have the capacity to open space for people to adapt to their surroundings; in other words, their main application is a facilitation of the present home. This happens through shared memories that lead to collective narratives and discussions around how material structures and objects are used and understood. These similarities in the practice and acts of remembering are what brings people of the present time closer to each other as well as to their shared past. So, objects of memory act horizontally in time and rhizomatically in their social networks. Equally, shared memories make stronger and more frequent transnational connections to others easier within the contemporary digital world. Pérez Murcia and Boccagni (2022) argue that objects can create emotional connections to past homes and contribute to 'home reproductions' or make a positive connection to their current life environments as a form of 'home making' (p. 589). It can also be added that it is through this bridging to the past, and past homes, that one can make a home in the present time (Povrzanović Frykman, 2016; Povrzanović Frykman & Humbracht, 2013).

The place that is called home includes objects and things that facilitate the process of living, increase the sense of comfort, and finally facilitate emotional attachments to places. The 'feeling at home', which is usually associated with safety, security and control (Ahmed, 1999; Boccagni, 2017) is often conveyed through how we create these attachments to spaces, people and to times. Much of the literature on material cultures and migration focuses on objects' roles in recreations of homecountry homes: either creating a material house in the home country (country of origin) or using objects from there in the migrant home in the host society. For example, in an ethnography by Boccagni (2013) and Boccagni and Pérez Murcia (2021) about Ecuadorian migrants living in the north of Italy, it is evident that migrants engage actively in a process of sending remittances to their home country of Ecuador. They do this in order to create a home over 'there' by building 'better' houses that would differentiate them from other Ecuadorians who stayed put. The study also shows that they persist in constructing their modest homes in Italy in order to resemble 'home in Ecuador' (Boccagni, 2013) as this resemblance acts as a form of memory card to connect them across the two worlds. Boccagni and Pérez Murcia (2021) argue that 'remittance houses' (Lopez, 2010) (houses that are rebuilt materially using remittances), negotiate class, status and belonging and act as 'agents' of social change in their own right.

The narratives that give life to these connections are of course important as they shape the stories that provide 'cultural continuity' between the host country and the home and over a long period of time for displaced people (Turan, 2010, p. 45). As Woodward (2001, p. 132) argues, narratives about objects are 'carefully constructed texts' that are 'embedded within a wider discourse about taste, the home and the family, which serve to make accountable everyday aspect of people's reality'. Thus, the meanings and intentions that are attached to material components of making a home are fundamental to understanding home in migration. In the above example, sending remittances is done in order to create an image of a life that distinguishes the migrant character from those who did not migrate. Here the object of *everyday use* (the house) becomes an object of *identity* (class and consumption). It must be

noted that these categorisations are never clearly divided and are co-constitutive of each other, but their individual meanings are uniquely distinct.

The relationship between objects and migration experiences is nuanced and complex. Fox (2016) notes that there is a common assumption of a difference in the meaning of objects for those who live sedentary and nomadic lives. The former tends to accumulate objects throughout their lives (even those that are not instrumental) whilst the latter group tend not to have possessions (carrying only objects necessary for survival). He argues that this is a misunderstanding of how we live with materiality whilst on the move. As we study materialities within migration, we need to move past objects as 'necessary' for survival and look more carefully at how survival is intersected with other necessities of our human condition such as 'biographical continuity' (Alonso Rev, 2016). This line of thinking has been taken up in studying photographs in mobile lives and how they act as 'multi-sensory objects' (Edwards, 2010) in creating a sense of stability in migrants' lives as well as using them as a method of narrating migration stories (Edwards, 2012; Kuhn, 2007; Pink, 2001; Rose, 2003; Yefimova et al., 2015 to name a few). In studying photographs that were sent between Ukrainian migrant mothers in Italy and their transnational families, Fedyuk (2012) shows that photographs can illuminate the emotional labour within transnational families mostly done by mothers living far away from their families. Tolia-Kelly (2004) argues that photographs symbolise relationships with colonial landscapes in South Asia and East Africa and connect South Asian migrant women in Britain to their past homes, nature and family life, connecting premigration life to post migration life. Smith and Pitt (2009) and Chuck et al. (2005) argue that there is a strong link between mental/physical wellbeing and personalisation of living- and work-spaces and photographs are one of the main tools to transform these spaces. Closely related, White and Beaudry (2009) and Rechavi (2009) argue that objects such as photographs can be used as a marker of identity and reflection on past decision making.

As we see, objects represent a much wider world in relation to what migrants lose by leaving homes behind and gain through setting up new homes. Objects' roles in this way need to be regarded in relation to identity expressions and cultural survival (Trabert, 2020). Ballinger (2003, p. 220) in her research with Italians living in post-Yugoslavia argues that in her participants' opinion, they had a sense of 'interior displacement, an exile of the heart and mind, if not the body'. These Italian migrants took with them a variety of objects and images, such as fragments of a home's foundations or vials of seawater to stay connected with those physical structures that they were forced to leave.

Most of the literature discussed here argues that migrants use material possessions, objects, food, within the structures of their dwellings as these materialities can create links between personal biographies of migrants and those of their significant others set within larger social, historical and cultural contexts within which they make new homes. Material homes as such offer novel spaces that create a link between biography and macro structure (Miller, 1987). For example, Tolia-Kelly (2004) shows how nation is linked to objects as mundane as photographs. Objects from homeland gain significance within a new context that accentuates their role in home-making and conveys a sense of belonging and attachment to the past life and homes. Whilst 'bridging' these two life worlds (homeland and host) is facilitated through objects (Povrzanović Frykman & Humbracht, 2013), by reflecting on how objects can evoke memories, we can understand the larger contexts around materialities. Svašek (2012) argues that materialities from homeland can trigger a sense of belongingness. Objects help migrants to preserve and negotiate bonds in their native country and the country in which they now reside. The socio-economic status of migrants also influences such negotiation and preservation. The reason for this is that migratory circumstances can vary greatly; political and economic hardships often prevent poorer migrants or refugees from carrying excessive possessions and lack of time impacts their engagement with objects and images, while affluent migrants are less likely to be subjected to these situations, often allowing continuous movement between homeland and new residential location. Thus, materiality in migration is closely bound up with migration regimes and the structural im/possibilities of home.

#### 4.4 Instrumentality and Everyday Life

Migrants, on one hand, have complex biographies due to the transnational movements and different spaces they call home. Moving from one context to another means going through difficult pathways and transiting through various countries, making the process of taking and maintaining physical objects between homes more difficult than when one is sedentary. So, a lot of investment and energy is demanded on the part of migrants to physically move home and the objects within it, so that decision-making about objects is a more rigorous process than in a sedentary house. On the other hand, migrants' lives are no different to non-migrants' in terms of the importance of memory objects, the necessity of having objects to facilitate everyday life, to connect one to the surrounding and to express ethnic, classed and gendered identities. In order to feel at home, one needs to be living the everyday, which relates to another categorisation of material life, that is, the instrumentality of objects in migrants' lives.

One line of inquiry in relation to materiality of home and instrumentality of objects in everyday life explores the role of food in migration. Povrzanović Frykman (2018) draws attention to food-related objects as everyday but important tools that can facilitate continuity in migrants' lives. She refers to the significance of a coffee pot from Italy for an Italian migrant in Sweden and argues that objects of this sort help create 'continuity in transnational lives' (p. 41) whilst at the same time, the use of some objects from home is more conditioned towards a routine practice and becomes part of the everyday life of a migrant. She argues that if we focus on the individuals and journeys, we enable understanding of how belonging can turn into 'ways of being'. It means that although we tend to attach meanings to objects that are symbolic in relation to the wider socio-cultural contexts within which they are produced and consumed, their use and role in facilitating our everyday life becomes part of our simple being.

Food production and consumption in migration and displacement may be seen as a form of expression of gender, ethnic, cultural or national identity but studies that have focused on this topic show that the importance of food lies also in everyday and quotidian practices (Christou & Janta, 2019; Gregory, 1999) as a 'normal' and routine practice rather than an extraordinary ethnic experience (Kershen, 2017). At the same time, we need to remember that food as an essential object and cooking as an everyday practice, can have deeper meanings (as we mention, these categories do overlap) (see Law, 2001). But we argue that in large part food production/consumption does remain a simple act of everyday home-making when it is done as a form of sustenance other than anything else and this dual representation of food remains quite strongly in some studies. For example Mata-Codesal and Abranches (2019) in studying food production, consumption and transportation among migrants argues that these practices evoke sensorial re-creations of home but on an everyday platform. It means that although items transported across borders have specific meanings, they are consumed with the purposes of a quotidian aim. Brown and Paszkiewicz (2017, p. 62) focusing on British Poles who bring food from their homeland to Britain describe through the words of their participants that this act of transportation is like 'bringing back a little bit of paradise'. Sending and receiving food parcels containing industrially produced material and home-made edibles is a typical migrant homemaking process (Povrzanović Frykman (2019) and way of making intimate transnational connections (Mata-Codesal & Abranches, 2018). These narratives about the particularity of food as a piece of the past or left-behind homes as well as an essential everyday item for life, can place food in the category of everyday objects.

Rabikowska (2010) for example shows that food contains emotional and mnemonic significance that can facilitate that sense of belonging whilst helping with the routinised life of a migrant. Walsh (2006) on a similar terrain focuses on everyday and mundane objects such as the role of an 'orange plastic bowl' in the life of Susan, a British 'expatriate' living in Dubai. Susan argues:

I bought [sic] a plastic container with me, the awful orange one at the back that I had when I first got married. I very rarely use it. I should chuck it away and get a new one I suppose, but it was a set. (Walsh, 2006, p. 134)

Later Walsh (2006) argues that the bowl contributes to Susan's sense of identity and how she sees herself as a 'kitchen person' and provider of food for the family. In a more recent study and through exploring different home activities with a group of Italian families, Arcidiacono and Pontecorvo (2019) argue that a family's identity can be studied through their everyday interactions with each other and with the materiality of their home. The experiences of those living together in a space are entangled with connections they make to objects within those spaces. In migrants' lives, similarly, Povrzanović Frykman (2016) argues that it is not true that migrants keep using only objects that remind them of their homeland. When migrants use objects other than those brought from their home country, they can continue their lived experiences across time and space (transnational social fields) by making new objects part of their own familiar array of objects. The continuation of those practices helps migrants to exercise their very being regardless of geographical location, which ultimately enhances their sense of belonging. The sense of belonging must be understood, not in the context of cultural representation (like feeling that you belong as an ethnic minority), but as more of a belonging to the routinised life surrounding one or what Christou and Janta (2019, p. 657) refer to as 'material consciousness' or 'embodied, emotional, performative and narrated accounts central to the notion of practice in everyday migrant lives'. However, because the things we do on an everyday basis unconsciously enable us to continue our habitus without interruptions, we feel 'normal' because we do things as normal. Everyday simple materialities such as a plastic bowl or a coffee pot here provide deeper meanings whilst being used as everyday items. But it is their everyday functionality that is at the heart of why those objects are important.

What is important is that whilst it is tempting to give deeper meanings to everyday materialities, in the end, many objects in a migrant home are functioning instruments as they would be in any home setting. Scholliers (2001, p. 4) questions whether stuff like food would be the 'sole factor of identification of a group or an entire nation?' The answer is no. We cannot add symbolic meanings all the time to objects that are simply instrumental in facilitating everyday life. But at the same time, we need to remember that instrumentality is not a shallow reading of objects; it is central to an everyday sense of belongingness to our material worlds. We need to 'use' the objects we buy or are given as gifts. And thinking about the different functions of an object will help to understand and categorise objects better if we want to realise the true potential of material possessions and the worlds of migrants.

Personalising the interiors of the domestic spaces, as well as its practical and utilitarian elements, has the function of increasing or improving the wellbeing of migrants. The use of personalised items helps migrants to feel attached to the surroundings, even if they have instrumental usage (Hadjiyanni, 2019). A noteworthy example of this is Seo and Mazumdar's (2011) research with Korean older migrants in California, who changed the interiors of their housing structures to increase their wellbeing through heating a part of the room's floor to symbolise a traditional Korean concept of 'hearth'. The participants narrated how a warm spot in the room would evoke feelings of childhood as they had a spot in Korean houses that is meant to be taken as a hearth. A similar point is made in Fathi's (2023) research with an Iranian asylum seeker in Germany who used plants, photos and traditional Iranian artefacts to change the 'empty' dynamics of his room in a *heim* (refugee camp). He (the participant), who actively wrote and thought about the materiality of his surroundings, argued that the emptiness or lack of material details in the housing structure in heims where he lived in Germany was intended (in his opinion) to detach the resident from the structures of the house to increase their sense of alienation from where they live (Fathi, 2023). That is, the life of a resident asylum seeker is inseparable from the empty structures of their surroundings whilst at the same time it becomes estranged and detached from this architecture. This duality turns home-making into an impossible act in this setting. In order to tackle this, he used some plants, which he called 'his children' in the room, to have another live element in the room. These small and everyday efforts by the asylum seeker in an attempt to replace what he has lost in the process of displacement and constant movement is reminiscent of Taylor's (2013) argument about 'botanica', belongings of Cypriot refugees in London. As Taylor (2013, p. 150) argues in relation to trees, crops and plants in the lives of Cypriot refugees in London, 'the loss of the material home is rarely recognized in the assessment of the financial, cultural and social impoverishment of the refugee'.

What is evident here is that deeper meanings of everyday objects and their uses are reflected upon when actors are asked about them. In fact, meanings become enmeshed in the everyday contexts of lives and form part of the life and being but are often not articulated as such. This shows the importance of interdisciplinary projects within sociological, psychological and anthropological research and the need for further research in the field of material culture studies to reflect upon the importance but also the instrumentality of objects in displacement.

## 4.5 Objects and Social Identity

Home can be an important symbol of individuals' identities; particularly the interior of a house or a dwelling can be very telling about people's occupations, personalities and lifestyles. Després (1991) writes about the link between the interior of a home (here a dwelling) and identity from a social psychology perspective. She says the home:

[...] plays a crucial role in people's definition of their self-identity, acting as a dialogue between them and the larger community. As a container for the material possessions that are meaningful to each household member, the home provides the information necessary to the development of their self-identity, these objects being concrete embodiments of different aspects of their personality (Csikszentmihalyi & Rochberg-Halton, 1981). Acting up and modifying their dwelling through the selection and manipulation of its external appearance, individuals acquire and communicate knowledge about their personal self. (Després, 1991, p. 101)

Fox (2016) argues that since we are born, we accumulate objects that we use to selfidentify. These objects usually mark a certain zone, or an 'intimate domain of our being' (p. 71). In this last section, the focus is on research on those objects that convey a form of identity in migration, a sense of self in terms of who we are and who we are not, both on an individual and collective level. Amongst these studies, there are scholars who point to different forms of identification processes. The focus here is on those relating to class, gender and national identity as these identity categories emerge as the most important ones in research examining migrants' lives from an intersectional perspective.

Savaş (2014) in research with Turkish migrants in Vienna discusses how a collective taste about materialities of home is achieved through consumption of the same or similar objects imported from Turkey such as tea glasses and saucers. Savaş (2014) argues that this contributes to a form of belonging to Vienna as a Turkish person. The mass migration of objects from Turkey to Austria produces a form of collective national identity and taste among Turkish migrants who plan to permanently make Vienna their home. This combination of national and classed identity is an interesting take by Savas (2014) as it reveals how class can be reproduced in migration through the consumption of certain material stuff from a collective source (homeland). Similarly, Fathi's (2017) research with Iranian women migrants in the UK shows how classed and national identities can be represented through objects that migrants take with them. For example, certain objects have class and status connotations such as Persian carpets, which are transported by Iranian migrants as a common practice (Fathi, 2020). Fathi (2020) argues that carpets have a deep and historical symbolic meaning in the Persian culture. As they are given as an essential item of a girl's dowry in Iran at the time of marriage, they seem to serve a particularly important function in circumstances of migration. They are objects with both national identity and classed status due to their monetary value, rarity and unaffordability for many families. These characteristics make the carpet a valuable identity object and a narrative around its significance is used effectively by migrants to showcase the 'superior' interiors of their homes to other Iranian migrants as well as non-Iranian visitors (Fathi, 2020).

In this way objects reproduce various meanings of home in a complex way through value systems and status compositions (Noble, 2012). Noble (2012, p. 435) argues that 'through consumption, people participate in the production of cultural meaning, because they re-signify these objects in ways which appropriate, personalise, and individualise them'. He suggests that by reappropriating objects from the realm of the public domain to the domestic space, people actively assign individualised meanings to objects and make them their own, both 'symbolically and physically' (Noble, 2012, p. 435). As such, objects can offer pathways into individuality. They distinguish their owners from others, an important aspect of class and consumption that has been taken up by sociologists of consumption (Warde, 2015). As concepts of social class, prestige and status are wider than the meanings associated only with socio-economic position (Bourdieu, 1984), the role of objects and patterns of consumption remains highly subjective and complex (Fathi, 2017) and can tell us a lot about how materiality, which is bound up with intersectional power relations, can have symbolic meanings in the lives and homing projects of migrants.

Furthermore, objects that are used to convey a particular form of identity (national, classed, gendered) maintain individuality and a sense of integration within the social arena whilst they are also important for improving mental wellbeing. For example, Smith (2007, p. 417) states that the 'handling of familiar or nostalgic objects whose worth is enhanced for the individual through idiosyncratic memory' is how migrants maintain personal identity and mental wellbeing. Keeping, using and consuming objects at home is a way to demonstrate who they are/not.

In consumption and class scholarship, many consumption practices take place within the private spheres of home. Indeed personalisation of interiors of a house or a dwelling can be a visual expression of the resident's identity and lifestyle as has been found in different contexts and among various age groups (Rechavi, 2009; Shin, 2014). Shin (2014) in studying older residents of care homes, argues that room

personalisation by using ornaments, objects made in arts and crafts activities or lifelong collections of china sets transforms their rooms of residence into their 'homes'. The use of objects in relation to identity is also a topic that has been taken up by scholars writing about refugee camps and shelters. Nabil et al. (2018) argue that in Za'atari refugee camp, the residents use art works and private and public areas of the camp to help them retain their lost identity and heritage. They argue that practices of 'crafting' and 'making', learning DIY techniques and tools can enhance their living conditions, bringing them closer to what can be called or felt like home.

## 4.6 Conclusion

In an era in which we are more connected (physically and virtually) and people move faster and perhaps more easily, the role of objects in the process of homemaking along migration routes needs fresh attention. Movement across borders is a physical act that includes movements of subjects as well as certain objects. The material objects, even those that are discarded and left behind, as mundane as a water bottle, or ragged clothes, tell a lot about people's migration and settlement journeys (De León, 2013, 2015). People's possessions remain an important aspect of migrant homing, but to describe, discuss and narrate them, requires reflective action. Studies in the different fields of anthropology, psychology and sociology have delved into how home-migration journeys are affected by what is taken, left, kept dearly or discarded. Migrants continually adapt to their new spaces and to the structural im/possibilities of home, whilst at the same time they change the same space to fit their new lifestyles (Alt, 2006). These residential changes are facilitated in large part by objects that they bring from their homeland, the ones they purchase and acquire and the ones that they discard. Through objects, people on the move reproduce their new lives.

In the first section of this chapter, which focuses on memory and nostalgia, we drew on studies on the importance of personal artifacts and family objects such as pieces of houses in homelands and photographs to show how they can reveal personal attachments. These studies attest to an important fact, which is that although there is much overlap between the three themes relating to objects of migrant home, that is, the memories associated with them, their everyday use and the sense of identity they give to the owner, they are an inseparable part of home in migration and their contribution to the formation of home needs to be considered through the role they play in the personal and wider worlds of migrants. The second part discusses objects that may carry deeper levels of meanings, but their functionality remains with their purpose of acting as an everyday instrument to facilitate life. Some refer to how objects of belonging can facilitate ways of being (Povraznović Frykman, 2011). The final part focuses on objects that would bestow a form of identity that connects migrants to the outside world, placing them as actors that could tell others who they are through consumption of objects. This last group, directly related to intersectionality, then provides avenues through which migrants can reproduce national, classed and gendered identity affiliations. This offers a worthy route of investigation into new avenues within migration studies.

To conclude, we argue that material components of migration trajectories are important elements in reinstating memories of past homes, the present life and identities that often seem to be lost in the process of forced displacement, asylum seeking, in-between time zones, transit contexts and all other unsettling situations. Material home is perhaps seen as the last issue that migrants who are dealing with unstable situations need to be concerned about. This kind of thinking is reflected in how accommodations and shelters are soulless and empty spaces, but a look at the current state of the art in this area shows how vital objects are in structurally making it possible to call a space home or not.

## References

- Ahmed, S. (1999). Home and away: Narratives of migration and estrangement. *International Journal of Cultural Studies*, 2(3), 329–347.
- Alonso Rey, N. (2016). Memory in motion: Photographs in suitcases. In M. Palmberger & J. Tošić (Eds.), *Memories on the move: Experiencing mobility, rethinking the past* (pp. 101–125). Palgrave Macmillan.
- Alt, S. M. (2006). The power of diversity: The roles of migration and hybridity in culture change. In B. M. Butler & P. D. Welch (Eds.), *Leadership and polity in Mississippian society* (Occasional paper no. 33. Center for Archaeological Investigations) (pp. 289–308). Southern Illinois University.
- Arcidiacono, F., & Pontecorvo, C. (2019). On materiality: Home spaces and objects as expanding elements of everyday experiences. *Forum Qualitative Sozialforschung/Forum: Qualitative Social Research*, 20(3), no page number.
- Ballinger, P. (2003). *History in exile: Memory and identity at the borders of the Balkans*. Princeton University Press.
- Basu, P., & Coleman, S. (2008). Introduction. Migrant worlds, material cultures. *Mobilities*, *3*(3), 313–330.
- Becker, F. D. (1977). Housing messages. Dowden, Hutchinson and Ross.
- Boccagni, P. (2013). Migration and the family transformations it "leaves behind": A critical view from Ecuador. *The Latin Americanist*, *57*(4), 3–24.
- Boccagni, P. (2017). Migration and the search for home: Mapping domestic space in migrants' everyday lives. Palgrave Macmillan.
- Boccagni, P., & Pérez Murcia, L. E. (2021). Fixed places, shifting distances: Remittance houses and migrants' negotiation of home in Ecuador. *Migration Studies*, 9(1), 47–64.
- Bourdieu, P. (1984). A social critique of the judgement of taste (Traducido del francés por R. Nice). Routledge.
- Brown, L., & Paszkiewicz, I. (2017). The role of food in the Polish migrant adjustment journey. *Appetite*, *109*, 57–65.
- Christou, A., & Janta, H. (2019). The significance of things: Objects, emotions and cultural production in migrant women's return visits home. *The Sociological Review*, 67(3), 654–671.
- Chuck, A. W., Milke, D. L., & Beck, C. H. M. (2005). Degree of bedroom personalization in institutional and homelike settings for persons with dementia: A quantitative investigation. *Canadian Journal on Aging*, 24, 329–337.
- De León, J. (2013). Undocumented migration, use wear, and the materiality of habitual suffering in the Sonoran Desert. *Journal of Material Culture*, *18*, 321–345.

- De León, J. (2015). *The land of open graves: Living and dying on the migrant trail*. University of California Press.
- Després, C. (1991). The meaning of home: Literature review and directions for future research and theoretical development. *Journal of Architectural and Planning Research*, 8(2, Special Issue: The Meaning and the use of home), 96–115.
- Dinç, P. (2009). Gender (in)difference in private offices: A holistic approach for assessing satisfaction and personalization. *Journal of Environmental Psychology*, 29, 53–62.
- Edwards, E. (2010). Photographs and history. Emotion and materiality. In S. Dudley (Ed.), *Museum* materialities: Objects, engagements, interpretations (pp. 21–38). Routledge.
- Edwards, E. (2012). Objects of affect: Photography beyond the image. Annual Review of Anthropology, 41, 221–234.
- Fathi, M. (2017). Intersectionality, class and migration: Narratives of Iranian women migrants in the UK. Springer.
- Fathi, M. (2020). Displaced home-objects in homing experiences. In P. Adey et al. (Eds.), *The handbook of displacement* (pp. 613–628). Palgrave Macmillan.
- Fathi, M. (2023). Constructing home through unhome: Narratives of resistance by an Iranian asylum seeker in Germany. *Social Sciences*, 12(1), 16.
- Fedyuk, O. (2012). Images of transnational motherhood: The role of photographs in measuring time and maintaining connections between Ukraine and Italy. *Journal of Ethnic and Migration Studies*, 38(2), 279–300.
- Fox, M. A. (2016). Home: A very short introduction. Oxford University Press.
- Giorgi, S., & Fasulo, A. (2013). Transformative homes: Squatting and furnishing as sociocultural projects. *Home Cultures*, 10(2), 111–133.
- Golovina, K. (2018). Material culture and bricolage: Russian-speaking migrants in Japan who make and procure objects. *Forum for Anthropology & Culture.*, 14, 204–229.
- Gregory, S. (1999). Gender roles and food in families. In L. McKie, S. Bowlby, & S. Gregory (Eds.), *Gender, power and the household* (pp. 60–75). Macmillan.
- Hadjiyanni, T. (2019). The right to home: Exploring how space, culture, and identity intersect with disparities. Springer Nature.
- Kershen, A. (2017). Food in the migrant experience. Routledge.
- Kuhn, A. (2007). Photography and cultural memory: A methodological exploration. *Visual Studies*, 22(3), 283–292.
- Law, L. (2001). Home cooking: Filipino women and geographies of the senses in Hong Kong. *Ecumene*, 8, 264–283.
- Ley, D. (2004). Transnational spaces and everyday lives. *Transactions of the Institute of British Geographers*, 29(2), 151–164.
- Lopez, S. L. (2010). The remittance house: Architecture of migration in rural Mexico: Buildings and landscapes. *Journal of the Vernacular Architecture Forum*, 17(2), 33–52.
- Mata-Codesal, D., & Abranches, M. (Eds.). (2018). Food parcels in international migration: Intimate connections (pp. 1–21). Palgrave Macmillan.
- Mata-Codesal, D., & Abranches, M. (2019). Sending, bringing, consuming and researching food parcels. In D. Mata-Codesal & M. Abranches (Eds.), *Food parcels in international migration: Intimate connections* (pp. 1–21). Palgrave Macmillan.
- Miller, D. (1987). Material culture and mass consumption. Basil Blackwell.
- Miller, D. (1998). Material cultures: Why some things matter. University of Chicago Press.
- Nabil, S., Talhouk, R., Trueman, J., Kirk, D. S., Bowen, S., & Wright, P. (2018). Decorating public and private spaces: Identity and pride in a refugee camp. In *Extended abstracts of the 2018 CHI* conference on human factors in computing systems (pp. 1–6).
- Navaro-Yashin, Y. (2009). Affective spaces, melancholic objects: Ruination and the production of anthropological knowledge. *Journal of the Royal Anthropological Institute*, 15(1), 1–18.
- Neumark, D. (2013). Drawn to beauty: The practice of house-beautification as homemaking amongst the forcibly displaced. *Housing, Theory and Society, 30*(3), 237–261.
- Noble, G. (2012). Home objects. In S. Smith (Ed.), International encyclopaedia of housing and home (pp. 434–438). Elsevier.

- Pérez Murcia, L. E., & Boccagni, P. (2022). Do objects (re) produce home among international migrants? Unveiling the social functions of domestic possessions in Peruvian and Ecuadorian migration. *Journal of Intercultural Studies*, 43(5), 589–605.
- Pink, S. (2001). Doing visual ethnography: Images, media and representation in research. Sage.
- Povrzanović Frykman, M. (2011). Connecting three homelands: Transnational practices of Bosnian Croats living in Sweden. In M. Valenta & S. P. Ramet (Eds.), *The Bosnian diaspora: Integration in transnational communities* (pp. 241–259). Ashgate.
- Povrzanović Frykman, M. (2016). Conceptualising continuity: A material culture perspective on transnational social fields. *Ethnologia Fennica*, 43, 43–56.
- Povrzanović Frykman, M. (2018). Food as a matter of being: Experiential continuity in transnational lives. In *Food parcels in international migration: Intimate connections* (pp. 25–46). Palgrave Macmillan.
- Povrzanović Frykman, M. (2019). Transnational dwelling and objects of connection: An ethnological contribution to critical studies of migration. *Journal of European Ethnology and Cultural Analysis (JEECA)*, 1, 29–45.
- Povrzanovic Frykman, M., & Humbracht, M. (2013). Making palpable connections: Objects in migrants' transnational lives. *Ethnologia Scandinavica*, 43, 47–67.
- Rabikowska, M. (2010). The ritualisation of food, home and national identity among Polish migrants in London. Social Identities, 16, 377–398.
- Rechavi, T. B. (2009). A room for living: Private and public aspects in the experience of the living room. *Journal of Environmental Psychology*, 29(1), 133–143.
- Rose, G. (2003). Family photographs and domestic spacings: A case study. *Transactions of the Institute of British Geographers*, 28(1), 5–18.
- Savaş, Ö. (2014). Taste diaspora: The aesthetic and material practice of belonging. Journal of Material Culture, 19(2), 185–208.
- Scholliers, P. (2001). Meals, food narratives, and sentiments of belonging in past and present. In P. Scholliers (Ed.), *Food, drink and identity: Cooking, eating and drinking in Europe since the Middle Ages* (pp. 1–22). Berg.
- Seo, Y. K., & Mazumdar, S. (2011). Feeling at home: Korean Americans in senior public housing. *Journal of Aging Studies*, 25(3), 233–242.
- Shin, J. H. (2014). Making home in the age of globalization: A comparative analysis of elderly homes in the U.S. and Korea. *Journal of Environmental Psychology*, *37*, 80–93.
- Smith, M. L. (2007). Inconspicuous consumption: Non-display goods and identity formation. Journal of Archaeology Method and Theory, 14, 412–438.
- Smith, A., & Pitt, M. (2009). Sustainable workplaces: Improving staff health and well-being using plants. *Journal of Corporate Real Estate*, 11(1), 52–63.
- Svašek, M. (2008). Postsocialist ownership: Emotions, power and morality in a Czech village. In M. Svašek (Ed.), *Postsocialism: Politics and emotions in Central and Eastern Europe* (1st ed., pp. 95–114). Berghahn Books.
- Svašek, M. (2012). Introduction: Affective moves: Transit, transition and transformation. In M. Svašek (Ed.), *Moving subjects, moving objects: Transnationalism, cultural production and emotions* (1st ed., pp. 1–40). Berghahn Books.
- Taylor, H. (2013). Refugees, the state and the concept of home. *Refugee Survey Quarterly*, 32(2), 130–152.
- Tolia-Kelly, D. (2004). Materializing post-colonial geographies: Examining the textural landscapes of migration in the South Asian home. *Geoforum*, 35, 675–688.
- Trabert, S. (2020). Understanding the significance of migrants' material culture. *Journal of Social Archaeology*, 20(1), 95–115.
- Turan, Z. (2004). Personal objects from the homeland: Re constructing cultural and personal identities. *International Journal of the Humanities*, 1, 465–480.
- Turan, Z. (2010). Material objects as facilitating environments: The Palestinian diaspora. *Home Cultures*, 7(1), 43–56.

- Walsh, K. (2006). British expatriate belongings: Mobile homes and transnational homing. *Home Cultures*, 3(2), 123–144.
- Warde, A. (2015). The sociology of consumption: Its recent development. Annual Review of Sociology, 41, 117–134.
- Wells, M. (2000). Office clutter or meaningful personal displays: The role of office personalization in employee and organizational well-being. *Journal of Environmental Psychology*, 20, 239–255.
- White, C. L., & Beaudry, M. C. (2009). Artifacts and personal identity. In T. Majewski & D. Gaimster (Eds.), *International handbook of historical archaeology* (pp. 209–225). Springer Publishing.
- Winnicott, D. W. (2001 [1971]). Playing and reality. Routledge.
- Woodward, I. (2001). Domestic objects and the taste epiphany. A resource for consumption methodology. *Journal of Material Culture*, 6(2), 115–136.
- Yefimova, K., Neils, M., Newell, B. C., & Gomez, R. (2015). Fotohistorias: Participatory photography as a methodology to elicit the life experiences of migrants. In 2015 48th Hawaii international conference on system sciences (pp. 3672–3681). IEEE.

**Open Access** This chapter is licensed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license and indicate if changes were made.

The images or other third party material in this chapter are included in the chapter's Creative Commons license, unless indicated otherwise in a credit line to the material. If material is not included in the chapter's Creative Commons license and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder.



# Chapter 5 Migration Status, Class, Race and Home



# 5.1 Introduction

For many, the concept of home is very simple. It's a place where no one questions your right to be; a place of belonging that points to your history, your past, an archive of sorts that metaphorically documents a lineage that marks you as nonalien. (Silva, 2009, p. 694)

Silva's (2009) concept of home is one that denotes the privilege of being able to claim a place as home in an unproblematic way, a privilege that is historically deeply rooted. At the same time, it suggests that for many others, home is not so straightforward – it is elusive, it is a place of not-belonging, or is a place either left-behind or imagined as being elsewhere. What Silva (2009) points to in these observations is the deeply political nature of home as an expression of the bordering processes that separate the ones who belong, without question, from those whose presence is constantly questioned.

These bordering processes must be understood in the context of historical legacies of possession, dispossession, colonization and exclusion. Complex geopolitical and socio-spatial power relations underlie the processes by which one's visible and invisible lineages (racialised and classed) mark one out as alien or not. Contemporary migration regimes, moreover, reproduce these power relations, producing migrant statuses that are bound up with economic and racial inequalities that structure the possibilities and impossibilities of home in migration.

Therefore, this chapter places migrant status in the foreground, focusing on the role of migrant status and relatedly, of race and class, as powerful differentiating forces shaping the structural im/possibilities of home in migration. In other words, we explore the role of (racializing) migration regimes as powerful forces that allow or deny individuals, in different ways, the possibilities of feeling at home, imagining home or doing home. It is critical also to acknowledge how migrant status, class and race intersect not only with each other but with gender, age, and other power relations in structuring these im/possibilities.

In the next section, the focus is placed on 'privileged' positioning in migration, exploring the ways in which privilege, whiteness and coloniality overlap in notions of home. Privilege is intersectionally related to classed and racialised systems. It produces and is produced through migration status and the politics of class, race and ethnicity within migration and settlement regimes. In this regard we review some studies that focus on privileged forms of migration, such as lifestyle migration and so-called high-skill and 'expat' migration. We explore how migrant groups who have positions of privilege can draw on their available capital (and privilege) to reproduce comfortable notions of home in migration, and the ways in which different migration regimes enable this.

The subsequent section focuses on the phenomenon of enforcement, precarity and stuckness in migration, as a lens through which to explore how migration regimes intersect with social class and race to limit the possibilities of feeling at home for other migrant groups, and importantly, how they navigate these limitations and borders in seeking to find and make home in the world. Precarity is particularly related to temporal migration regimes so this section draws on the notion of temporality as an important component of home-making in migration processes. The final section presents a critical discussion of how the concepts of privilege and precarity help in understanding im/possibilities of home and migration in relation to migration status, class and race.

## 5.2 Privilege and Home in Migration

In this section, the concept of 'privilege' is deployed to shed further light on the structural im/possibilities of home in migration. Including privileged forms of migration in the discussion is essential if we are to understand the dynamics of structural possibilities of home. Privilege is understood here as the oft invisible benefits conferred on those who are advantaged by systems of inequality and we argue that it is crucial to focus on privilege in seeking to understand the workings of uneven power relations and structures in society. The concept of privilege has become particularly valuable in migration studies in recent years, being utilised in analysing forms of migration associated with a high degree of control over one's own mobility and the reproduction of high social status, such as those labelled lifestyle migration, high-skill labour mobility and student migration (Benson, 2019; Croucher, 2012; Duplan & Cranston, 2023). Many agree that privileged migration is that which is undertaken by relatively privileged individuals, usually from relatively rich countries and characterised by ease of crossing international borders (Benson & O'Reilly, 2018; Croucher, 2012). Some consider student migration as a form of temporary migration that is privileged, even sometimes being labelled as 'academic tourism', in particular when students originate from a wealthy country (Breen, 2012).

Robertson and Roberts (2022) argue that middle-class cultures of migration are expanding, with mobility aspiration becoming increasingly normative in the life-trajectories of middle-class migrants. Benson and O'Reilly (2018) argue further that privileged migration is integrally bound up with the global power dynamics of both neoliberalism and postcolonialism, both of which perpetuate relations of economic inequality that enable highly voluntary migration for some while imposing enforced im/mobilities on others. Thus, although privilege in migration can be related to a range of social distinctions (including gender), we focus here in particular on class privilege, whiteness and western-ness as indicators of privilege in global migration dynamics, particularly in relation to the role of neoliberalism and postcolonialism.

As argued in Chap. 1, home is always socially situated, and the ways in which it is lived, experienced and understood are shaped in myriad ways by the intersecting power relations that underpin one's social positioning and situated imagination. Home in this way is also bound up with global power structures that make it more or less possible for people to feel at home in the world. Traditional understandings of home have tended to construct home in terms of an unproblematic, even natural, connection between a person, place and community, a safe and still place to leave and return (Dawson & Rapport, 2021) connecting to discourses of romantic nationalism, which construct the nation as home. However, critics of this idea argue that the ability to feel a close, secure, safe and unproblematic emotional connection to one's dwelling-place, or place of origin, is a privilege afforded to the few (Ahmed et al., 2003). The traditional discourse of home also denies the reality that very often the ability of one group to feel at home in place is only possible because of the displacement, dispossession or oppression of others. Whether historically through the role of colonialism, or more recently through far-right claims to exclusive placebased rights to belong, the ability to claim a home and to feel at home cannot be separated from wider power relations of privilege, marginalisation and politics of belonging.

Literature focusing on forms of migration such as lifestyle migration, student migration, high-skill migration and so-called 'expat'<sup>1</sup> migrants is insightful in this respect. Some of this literature is reviewed here, exploring how migrant groups with positions of privilege can draw on their available capital to (re)produce a comfortable sense of home in migration, which in turn itself can contribute to reproducing privilege. The literature points to three specific ways in which this process manifests: first, the ability to feel at home in a host society; second, the idea of temporary home/s as a privilege; and third, the ability to maintain an ideal of the homeland as home. Cutting across these three aspects, we argue here, is one central unifying feature of much (though not all) privileged migration, which is the constant possibility of returning 'home' or moving onwards to re-make home. This is

<sup>&</sup>lt;sup>1</sup>The term 'expat' is highly contested as it tends to be used to describe white Western people, often elites, who live outside their country of origin, with the result of placing them in a different category to all other migrants, thus reinforcing the Western othering of those deemed to be 'migrants'. We use the term here only when discussing studies that use it.

something that makes the lived experience of home-making in migration very different to that of less privileged and more precarious migrants who do not see this as a feasible option. These processes are also contextualised within the wider global and postcolonial power dynamics that produce, and are reproduced through, the imagination and making of homes in migration.

## 5.2.1 Feeling at Home in the Host Society as Privilege

Much literature attests to the structural difficulties of making a home in a new society, referring to barriers such as those related to one's migrant or citizenship status, age, gender, income level and family status (as outlined in previous chapters). This implies that privileged social positioning, on the other hand, can ease the process of settlement, familiarisation and inhabitance that is central to feeling at home in a new place, whether through the right to stay, the right to work, the ability to secure decent housing, the ability to be accompanied by family members, to forge new meaningful social connections, or to maintain transnational ties, or having the cultural capital to express one's identity in a place without fear of censure. In other words, a sense of spatial security is possible.

For example, Green (2023) reveals how white and western privilege enables later-life lifestyle migrants in Bali to comfortably make their homes in the host society, not least because of their purchasing power in terms of securing good quality housing and being able to maintain a very good quality of life, including the ability to hire domestic servants. Green (2023) elaborates on how these migrants construct a strong sense of home in the new place through the materiality of their houses, through which they nurture a sense of belonging and emotional attachment to the locality in Bali. They mobilise notions of authenticity and localness to construct and express identities through material aspects of their physical homes, but as Green (2023) points out, do so in ways that deny the unequal power relations, both locally and globally, that make it possible for them to have access to the type of housing and lifestyle with which to do so. His study finds echoes in Zhang and Su's (2020) research with lifestyle migrants in China who, they find, express strong place attachments to their places of residence, which they tend to construct as 'ideal homes', but simultaneously distance themselves from locals and often feel alienated.

Prazeres (2018), in her research with Canadian international students in Global South cities, suggests that her participants' privileged status as white western international students in these cities is a key factor in their ability to feel at home there. She relates this to their status as temporary and voluntary migrants for whom processes of local embedding are much less fraught than for other migrants. Moreover, Prazeres (2018) argues that the ease with which the students claim feelings of belonging and at-homeness in these cities is in itself a source of social distinction and cultural capital which they mobilise to enhance their social status as global citizens after they leave. Her study illuminates how they 'collect' homes throughout

their lives as they travel from place to place, as a way of accumulating cultural capital. In this way, the ability to comfortably feel at home in different places further enhances the privileged status of those who have access to this lifestyle, and further reinforcing uneven power relations shaping home in migration.

We cannot fully understand the forms of privilege enjoyed by the participants in Green's (2023) and Prazeres' (2018) studies without situating them, as they both do, in the context of global power dynamics, and acknowledging that constructions of home in migration are variously tied up with the legacies of colonialism and the uneven articulations of globalisation (see also Ahmed et al., 2003 and for an example of African migration, Ndlovu, 2010). For example, Green (2023) refers to the continuing salience of colonial 'place-myths', associated with the Dutch colonial legacy in Bali, which are mobilised by western lifestyle immigrants in their constructions of local belonging. Elsewhere, Moreton-Robinson (2003) provides a powerful critique of white migrants' constructs of home in Australia, linking these constructs to the history of colonization and dispossession of Indigenous people. She argues that the ability to claim that one 'belongs' in Australia, especially as a white person, is built on the historical racialized dispossession of Indigenous people and the continued hegemony of whiteness. Ahmed (1999, p. 335) intriguingly suggests that for the privileged migrant, 'the world is already constituted as their home', pointing perhaps to a certain quality of ease, comfort and sense of entitlement that accompanies the privileged (white, western, socio-economically advantaged) migrant on their travels and allows a sense of home to be both claimed and felt anywhere. Thus, Ahmed (1999) criticises scholarship that neglects the power inequalities that impact on how migrants come to feel and to claim a sense of belonging and home in their host societies.

#### 5.2.2 Temporary/Mobile Home in Privileged Migration

The privilege of claiming to feel at home 'anywhere' implies that home for the privileged migrant may also have a mobile and somewhat free-floating character. A number of studies point to the idea of 'temporary' or 'mobile' home as a feature of the hypermobile lifestyles of privileged transnational migrants (for example, Butcher, 2010; Walsh, 2011). In this construct, home is not tied to one place but is constructed as fluid or portable, located in transnational space, or in multiple places. The ability to construct home in these terms could be understood as a characteristic of privilege. Inherent in the migration regimes that enable privileged forms of migration is the constant possibility of returning to a stable home or choosing to move on to re-make home elsewhere – in other words, a high degree of choice/ control and global-movement possibilities because of factors such as citizenship status, income levels and valuable social and cultural capital. In these contexts, one can surmise that it may be possible to invest in home-making in different places, or to form fleeting attachments to different places, while maintaining one or more primary homes, or to construct a sense of home that is more tied to a type of lifestyle

than a concrete place. This is possible because the risks of these 'investments' (emotional, financial, social) are mitigated by the safety-net of the possibility of home being elsewhere.

For example, Cai and Su (2021), based on their research with 'expat' migrants in Guangzhou, China (their term, referring here to business and professional elite migrants) adopt the concept of 'temporary home'. They argue that 'the making of temporary home exemplifies the ways in which Western 'expats' undertake and experience a privileged mode of mobility' (p. 2815). They draw on Clifford's (1997) concept of 'dwelling-in-travelling' to articulate the idea of home as existing between mobility and immobility, not tied to one place but neither detached from place. Their study participants expressed a strong attachment to Guangzhou and their lives there, where they felt at home, but generally viewed themselves as visitors who would one day return 'home' or move on elsewhere (Cai & Su, 2021). Importantly, Cai and Su (2021) do not imply that 'temporary home' in mobility is a universal possibility for migrants. They characterise their participants' ability to feel at home in Guangzhou, while also remaining somewhat detached and planning to leave, as a form of privileged mobility, exemplified in the notion of temporary home, but made possible through favourable residency regulations and the global career opportunities of business and professional migrants in China. These circumstances are of course bound up with global neoliberal economic power relations that facilitate the easy mobility of some while imposing enforced im/mobilities on others.

However, it is important to point out that studies of constructions of home among privileged migrants do not usually extend to claiming that privileged migrants are completely free to create home on-the-move in accordance with their personal desires. Zhang and Su (2020) show how lifestyle migrants in China find themselves 'unmaking' home as they struggle with local social and economic circumstances that limit their ability to fulfil their image of their idealised home. Butcher (2010) draws on her research with Australian professionals in Singapore to emphasise that while 'expats' possess the cultural capital to re-make home on the move, they often feel destabilised or uncomfortable in unfamiliar environments and thus turn to more secure and familiar notions of home (such as 'expat bubbles' or nostalgia for a familiar place). Ullah et al. (2021) also find that privileged migrants originating from wealthy countries in Asia, Europe and North America who reside in Singapore, Malaysia, Thailand and Brunei often find themselves positioned as 'outsiders' in their adopted homes. Very often, although the host environment and culture are celebrated by 'expat' migrants as part of the communication of a cosmopolitan sense of home, this is done in a way that others and exoticises through an external gaze, while maintaining their strongest social ties to other (class- and raceprivileged) expats and transnational networks (see for example, Benson & O'Reilly, 2018). In other words, home is by its nature bound up with emotion/affect, with place, and with (classed and racialised) habitus. As Butcher (2010) argues, despite claims of the mobility and flexibility of home for highly mobile professionals, home still needs to feel like a space of culture fit and familiarity.

#### 5.2.3 Privileged Migration and Homeland

This quest for 'cultural fit and familiarity' (Butcher, 2010) means that the idea of a 'homeland' can continue to hold symbolic and material significance within privileged forms of migration. Firstly, the continuing connection to a homeland can be evident through home-making practices which involve transporting, displaying and using material belongings in the home that represent an emotional connection to the place of origin. This is particularly evident in Walsh's (2006) research with British 'expat' migrants in Dubai, in which she details how the emotional connection to the past home in Britain is part and parcel of creating a new home in Dubai. This occurs through the re-creation of a sense of continuity between the materiality of the current dwelling and past dwellings in Britain, and also through using British cultural products as part of building networks with other British (and Western English-speaking middle-class) migrants in Dubai. Similarly, Benson and O'Reilly's (2018) research with privileged British migrants in Malaysia and Americans in Panama reveals the importance of co-ethnic 'expat' networks and organisations as familiar spaces of belonging.

The ability to transport goods such as furniture and paintings between places of residence around the world, in order to re-create one's home, is of course a privilege, one that is not available to most transnational migrants. Furthermore, it must be acknowledged that Western 'expat' communities in host societies are 'classed and racialised formations' (Benson & O'Reilly, 2018, p. 217) that are made possible through deep social and cultural inequalities within host societies and globally. Thus the performance of national, racial or class belonging must be seen as a key element of processes of home-making that work to maintain valuable cultural and material connections to a globally powerful homeland identity.

Secondly, the place of origin, or homeland, can become the object of feelings of nostalgia and homesickness especially when migrants struggle to feel a sense of comfortable belonging in the host society. This is, of course, a common feature of transnational migration, but we argue that privileged migrant status is particularly associated with the construction of place of origin as home, or homeland, because the possibility of returning there exists (see the literature on remittance houses such as Boccagni & Erdal, 2021; Mand, 2010). This is evident in Wiles' (2008) research with New Zealanders in London. For her participants, return is a constant possibility; most see their migration as temporary and view New Zealand as home. They are relatively privileged migrants who choose to migrate and are facilitated in doing so by the preferential visa arrangements in the UK for young Commonwealth citizens (Wiles, 2008). In this context, a collective imaginary of New Zealand as home is constructed and reproduced among New Zealanders living in London. This is maintained through the ease of travelling back and forth between the two. The feasibility of holding a stable sense of one's homeland and staying connected to it through transnational connections points to the role of privilege (migrant status and economic means) in facilitating a comfortable sense of home in migration.

This contrasts with the relationship to country of origin among some refugees or other forced migrants for whom a return to homeland is not possible (Korac, 2009). As Pérez Murcia (2019) shows, for some the place left behind no longer constitutes a home. However, the idea of homeland as home is often maintained in a symbolic sense among migrant groups – more as a collective meaning rather than a private sense of home, a useful distinction made by Povrzanović Frykman (2002), denoting that while the homeland may be idealised as the symbolic/sacred/true home, it is necessarily not considered as a possible real home in which to live. Its symbolic importance is great, and it may even be imagined as a future home, but there are no moves to actually return. This is evident among Povrzanović Frykman's (2002) Croatian diaspora in Sweden, for whom Croatia is revered as the sacred homeland even though they may never have lived there, but its importance is central to their identities as exiled Croats. It contrasts with Wiles' (2008) New Zealand migrants, for whom the sense of New Zealand as home is very real, because it is highly possible to return there. The circumstances of the migration shape the relationship to the homeland, reflecting these differential structural im/possibilities of home in migration.

## 5.2.4 Summary

This review of existing literature reveals the many ways in which classed, racialised and western privilege eases the path to finding a comfortable sense of home in migration, whether through making a new home in the host society, creating temporary or mobile home, or maintaining a sense of homeland as home. Feeling at home in migration is made possible through the economic, social and political resources that a privileged migrant can deploy in controlling their own im/mobility, of which one key resource is the constant possibility of returning 'home' or moving on elsewhere. Returning 'home' is possible due to factors such as economic and political stability and personal resources in the place of origin, while moving on is facilitated by the mobility capital associated with Western citizenship and economic resources. In other words, the structural possibilities of home are bound up with the structural possibilities of privileged mobility. Understanding privilege in migration in this way provides a different perspective on the concept of 'spatial security' (see Chap. 1), that is, the security associated with feeling a sense of attachment to place and having the legal and formal documentation allowing one to live and stay in a place, which is what enables home in migration (Fathi & Ní Laoire, 2023). As we argue in Fathi and Ní Laoire (2023), place attachment can be to a non-local sense of place, an imagined past or future place, that allows one to construct home in migration (even if that home is not precisely in the current place of residence). However, for spatial security to exist, place attachment needs to be accompanied by legal rights. Ironically, as much literature attests, having the legal right to stay or settle in a place is what enables one to leave that place, because of the possibility of returning to it (Lundström, 2014). Privileged migration is associated with the legal right to stay (even if not permanently) in the host society, alongside the ability to return to one's place of origin. This, we argue, produces what could be called *transnational spatial security*, understood as the feeling and opportunity of secure inhabitance in multiple places at once. This concept helps us to understand how the structural possibilities of home in migration are integrally bound up with a complex range of processes from differential citizenship/residency rights to economic inequalities and the accumulation of cultural and social capital. As understanding privilege necessitates understanding oppression (Pease, 2010), and also precarity in migration status, the next section brings the focus to a different perspective on the intersection of migration status, class, race and home.

#### 5.3 Precarity and Home

Having discussed how privilege can enhance the possibilities of home in migration, this section shows that migrants who do not enjoy the same privileges that are afforded to others, as a result of their social positionality in terms of migrant status, social class and racialised identities, face more *im*possibilities in relation to feeling at home in migration. Many migrants such as those who are in irregular situations, seasonal, undocumented, asylum seeking and those who live in temporary situations in transit and border zones, deal with precarity of work, income, housing, residence status and citizenship on a daily basis and find the possibilities of making a home in migration very limited.

In this section the focus is placed on the concept of precarity as a lens through which to explore structural impossibilities of home in migration. As Lewis et al. (2015) argue, precarity is a condition that is increasingly associated with contemporary global migration, referring to lack of security in relation to livelihoods, along with the vulnerability to exploitation, violence and exclusion that comes with that. Precarity is closely connected to the idea of spatial security that is essential to feeling at home in place; insecure and uncertain migrant and residency status, insecurity of income or vulnerability to practices such as eviction or job loss, inevitably prevent a sense of spatial security. In particular, inadequate housing intersects with temporal limitations for migrants in circumstances of precarity. One aspect of this is that multifaceted systems of marginalisation (where migration regimes and other axes of difference intersect) often make sedentariness, through enforced immobility in migration, the only option for such migrants, adding to their sense of non-belonging and dissatisfaction with their surroundings.

# 5.3.1 Refugee Accommodation and the Structural Impossibilities of Home

Over 60% of the world's refugee population (19.5 million) resides in urban areas (UNHCR, 2017) and Brun and Fábos (2015) argue that nearly two-thirds of the world's refugees are in a seemingly never-ending exile with an average length of

waiting reaching 20 years. In the EU, cities offer opportunities in terms of the labour market (Van Liempt et al., 2015) and networking with other migrants, but at the same time, urban space also offers harsh conditions of settlement due to the lack of competency of cities to offer basic services (Van Liempt & Miellet, 2021). States also employ various policies to spatially segregate the communal housing arrangements often put in place for migrants or asylum-seekers to designated areas within or outside the spaces of towns and cities across Europe (Van Liempt & Miellet, 2021). These temporary accommodation structures, such as shelters, hostels and 'direct provision' (DP) accommodation centres, are normally devoid of any signs of what would suggest the notion of home (Thorshaug & Brun, 2019). In fact, what these 'empty' spaces (Fathi, 2023) show is displacement more than home (Van Liempt, 2023).

Beeckmans et al. (2022) in a recent edited collection, identify four types of socio-spatial contexts for home in displacement - camp, shelter, city and house. They show that the ways in which camps and shelters, as institutionalised forms of housing, are regulated by states means that they prove to be ineffective in giving a sense of home. Brun and Fábos (2015, p. 11), in another edited collection, discuss similarly how encampments and detention centres, or indeed any other precarious and temporary living conditions, emerge from 'policies of limbo', that is, a limbo 'created materially, discursively, and politically by the refugee regime' and which has profound implications for migrants' home-making. Fathi (2023) draws on the concept of unhome to refer to refugee accommodation centres, which are marked by solitude and exclusion unlike a home that is set within a community and is entangled with a social life. She argues that the soulless and empty housing structures of refugee accommodation or homeless shelters produce a sense of non-belonging as a way of reducing the time spent there by the residents. A considerable body of research demonstrates how reception centres cannot provide the basic elements of home, despite the fact that they are places where many people live for extended periods of time (Thorshaug & Brun, 2019) where their experiences of forced migration can be considered to become experiences of 'forced arrival', that is, of enforced conditions of living after arrival (Kreichauf, 2018). Studies point to the role of constant surveillance, lack of autonomy, cramped living conditions, lack of privacy, inability to prepare their own food and boredom as factors that militate against enabling a sense of home (Grønseth & Thorshaug, 2022; Howlett-Southgate, 2021; Van der Horst, 2004). As O'Reilly (2020) shows, these accommodation policies, by producing insecurity, fear and lack of autonomy, deny residents the basic attributes of home and function as mechanisms of social and spatial exclusion.

However, it is important to recognise the different social positionalities of residents of refugee accommodation. For example, research with children living in these kinds of arrangements suggests that they can experience them differently to adults. White (2012) illustrates the many ways in which children living in Direct Provision (DP) centres in Ireland, who participated in his research, were able to develop meaningful place attachments and social connections, despite the apparently isolated and unhomely nature of DP. He relates this to the ways in which spaces and daily interactions of communal living offered the children opportunities for cultivating peer friendships with other children living in the centre and also offered access to informal spaces for playing. This insight points to the importance of incorporating children-centred perspectives in discussions of home in migration.

The development of informal makeshift accommodation for migrants in transit shows how home-like spaces can emerge in the gaps left by the inadequacies of state provision of migrant accommodation, though these are highly vulnerable to removal or destruction by authorities. A well-known example of such temporary makeshift housing compounds was the Jungle, in Calais, France, located at the final stop for many migrants before reaching the UK. The Jungle was destroyed in October 2016. Migrants in Calais' Jungle resisted several attempts of expulsion and destruction of the camp by the French authorities. In February 2016, they were given a week to evacuate the premises, to which they appealed basing their argument on the right to respect for private and family life as well as on the right to respect for a home (Slingenburg & Bonneau, 2017). The Jungle was eventually destroyed only to push the residents who were stuck at the borders of France and the UK to other parts of France. As Van Liempt (2023, p. 991) argues, the destruction of these kinds of housing structures and social spaces is not limited to the destruction of materialities; it is as much about destroying 'a notion of rights and recognition, or even existential legitimacy'. Similarly, Massa (2022) writes about an incident of forced eviction, by law enforcement operators, of African migrants from a squat in Rome as not just the removal of their physical dwelling but of the feelings and values bound up with a sense of home.

#### 5.3.2 Protracted Displacement and Precarity

Precarious livelihoods, lack of access to formal or legal representation or protection of those transnationally displaced or of millions of IDPs (Internally Displaced People) results in long waiting times, at times more than two decades, for settlement. The majority of the world's protracted refugees are settled (or better to say are stationary) in developing countries (for example sub-Saharans in Tunisia and Algeria, South Americans in Mexico, Rohingya in Bangladesh and Afghans in Iran). This approach to managing the resettlement of displaced people has been termed 'warehousing' of human beings (Smith, 2004); rather than offering a 'durable solution' to refugee displacement, many of these policies that still continue to this date are based on the unrealistic and inhumane premise of simply ensuring migrants' survival until they return home.

Brun (2015), referring to the case of IDPs from Abkhazia who resided for more than 20 years in temporary housing near Tbilisi in the Western side of Georgia, argues that the only durable solution that is perceived by various stakeholders is their return. In this case, the next durable solution to the thousands of displaced people was a scheme to accommodate the displaced residents of the camps in housing cottages built outside cities. This strategy in effect did not change the precarity of their living conditions. Brun (2015) proposes the two concepts of immanence and

transcendence to shed further light on the effects of this. According to Brun (2015) transcendence refers to 'a mode of temporality in that the living subject is future oriented' whilst immanence refers to the act of sustaining life, as repetitive and cyclical. She argues that when transcendence is not available, individuals cannot fulfil their subjectivity. They have feelings of stuckness, being trapped in time with no foreseeable future oriented life, because according to Brun (2015, p. 47), 'future lies too far ahead'. What is important and relevant to our discussion in Brun's (2015) argument is that when we are talking about precarity of home-making, we deal with regimes of temporality that go hand-in-hand with policies and political processes that hold individuals in the present time and impose on them immanence without transcendence.

The translation of Brun's (2015) argument can be witnessed in the precarity of home for millions of marginalised forced migrants, which is embedded within their lack of legal rights, and leads to such protracted and long-lasting in-between situations. In other words, the temporality regimes have profound implications for how migrants can envisage a future life that includes everyday practices of education, work and leisure in making home. Their situations also remind us that perhaps the territorial aspect of a home is not as important as the emotional and relational, as according to Brun and Fábos (2015), drawing on Eastmond (2006), after so many years of being displaced, home as a fixed location, either in the place of origin or location of current residence, loses its meaning. Home, as a place where 'a normal life can be lived [...] with economic security, social context and [...] a sense of belonging', instead, they argue, lies in multiple places, or a 'trans-local' space where each locality becomes part of a new home (Brun & Fábos, 2015, p. 8; Eastmond, 2006). In fact, perhaps it is out of necessity, in order to feel ontologically secure, that many migrants relate home to mobile elements such as practices, materialities and relationships, as opposed to geographies and placed-based attachments. Massa (2022) suggests that it is more productive to talk about home as a set of practices through which migrants in situations of extreme precarity and vulnerability deal with the situations in which they find themselves and work at making and re-making home, than as a static concept.

# 5.3.3 Intersectionality, Selectivity and Everyday Bordering of Home

Precarity has become embedded in the living situations of migrants in many different types of circumstances and is bound up with the proliferation of different types of migrant status including forms of temporary migrant status and those falling within what Menjívar (2006, p. 999) calls 'liminal legality' as well as with precarious labour regimes and gendered vulnerabilities. Inequalities, especially in relation to housing for migrants, are produced not just by government policies but also by the actions of key players in housing markets, such as the financial sector and the labour market. Housing inequalities are also integrally bound up with less tangible social, cultural and legal factors that differentiate between who can make a home and who cannot.

Within this complex picture, there are policies that determine which migrants 'deserve' to be housed (in destination countries) or relocated (from transit contexts) based on constant assessments in terms of their eligibility for housing or refugee status (Dhaliwal & Forkert, 2015). Similarly, increasingly neoliberal approaches to migration are based on the benefits migration brings to host societies, a model that sifts through migrants and applies a selectivity based on factors such as age, social class or education level. For example, Theresa May (Former British Prime Minister and Home Secretary) said in 2010:<sup>2</sup>

Let me deal with a myth that has arisen in recent months. We can reduce net migration without damaging our economy. We can increase the number of high value migrants: the entrepreneurs, the investors, the research scientists – at the same time as we reduce the total number of people coming to Britain through the economic routes. We can attract more of the brightest and the best at the same time as we reduce the overall number.

Such policies are not limited to the UK. Across many countries in the Global North, we are witnessing how structurally it is impossible for certain groups of migrants to make a home, as preferential treatments, and systems of everyday bordering that treat migrants differentially, are in place before they arrive in destination countries. These preferential arrangements, both before and after arrival, reveal how migration policy regimes are integrally bound up with intersectional power relations, such as those relating to social class, age and race, and in this way, shaping the structural im/possibilities of home in migration for different social groups.

Situations of precarity that limit the possibilities of feeling at home manifest in different ways for different migrant groups. For example, Wilkins' (2019) research with Vietnamese migrants on temporary visas in London shows how practices of border regulation impact on their ability to feel a sense of home and belonging in London. She relates this to the geopolitics of immigration in the UK, which has shifted towards a selective points-based system which gives preferential treatment based on factors such as skill levels, language fluency, age or health. As a result of the uncertainties of their temporary status, together with their vulnerability to poor housing and working conditions, many migrants in Wilkins' (2019) research live with a sense of being unable to create a long-term home.

What Wilkins' (2019) research demonstrates is that precarity and related feelings of insecurity and un-belonging are not just about legal status. Even in situations of legal status, the uncertainties attached to some forms of status have a profound effect on the possibilities of home in migration. Furthermore, precarity as a condition can have effects on a person's life that persist even when legal status becomes more secure. For example, Howlett-Southgate (2021) demonstrates the complexities of home-making for people who have achieved refugee status in Ireland, because

<sup>&</sup>lt;sup>2</sup> https://www.gov.uk/government/speeches/immigration-home-secretarys-speech-of-5-november-2010.

of the ongoing effects of having spent long periods of time in the asylum system along with broader processes of exclusion experienced by refugees. Kox and Van Liempt (2022) make a similar argument based on their research with refugees in the Netherlands. Howlett-Southgate (2021) argues that problems in accessing housing and work, together with experiences of racism, and separation from family members, mean that creating a new sense of home for refugees in Ireland is extremely challenging.

Wimark's (2021) research with queer refugees in the Swedish countryside reinforces this point. He finds that even when queer refugees move out of state-controlled spaces, they find it difficult to feel at home in 'normative' spaces. He suggests that this 'blurs the border of a specific timeframe of leaving/losing a home (country) and creating a new home since many queer refugees find it difficult to find belonging that fully embraces them' (p. 649) and he points to the importance of moving beyond state-controlled spaces to the norms that govern home-making (Wimark, 2021).

#### 5.3.4 Making Home in Displacement and Precarity

Despite living in situations of precarity and displacement, existing research shows how migrants in such circumstances continue to work at making home, finding home or re-creating home. Strategies to find a sense of home can include making current living conditions more home-like, finding home-like everyday spaces, maintaining ties to former homes or imagining future homes. Brun and Fábos (2015) provide a detailed framework of the process of 'making homes in limbo' among individuals in protracted displacement. Their concept of 'constellations of home' refers to the diverse and complex ways in which migrants living in protracted displacement continue to 'make home' through their present practices and hopes for the future, despite the many ways in which their current circumstances are marked by hardship and uncertainty (Brun & Fábos, 2015). They caution, however, that even when home-making practices occur in a location like a refugee camp, this does not necessarily mean that the place becomes a home.

Research has explored how residents of asylum-seeker accommodation centres manage to make home even in such unhomely spaces. For example, Boccagni (2022) points to four different types of home-making practices through which residents of asylum reception centres seek to shape their environments to make them more familiar, normal and personal. These are: ways of improving space, ways of enabling cultural reproduction and biographical continuity, ways of privatising space and ways of beautification. Others point to strategies of inhabitance such as restoring short-term temporal predictability through establishing daily routines (Gil Everaert, 2021). Going outside the institutional accommodation centres to find home-like spaces elsewhere, for example, outdoor green spaces (Van Liempt & Staring, 2021), is a strategy that is important for some. However, Gil Everaert (2021) argues that even as residents of such centres enact home-making practices, they do so as a way of 'inhabiting the meanwhile', that is, of building temporary homes without letting go of their future plans for a more permanent home elsewhere.

Some scholars have looked at how precarious housing is intersected with materiality, how objects of significance are used for personalisation and how small-scale acts attempt to give new meanings to spaces of non-home (Boer, 2015; Neumark, 2013; Trapp, 2015). Motasim and Heynen (2011) show that IDPs in Sudan design their own space, establishing it as a mode of resistance against the forceful urban environment of Khartoum (the capital city) which is different to the lands they have had to flee from. By doing this, they make home into a mobile form of habitat that allows them to take it with them where they go. In other contexts, migrants living in precarity create a sense of home through practices that build a diasporic identity or create 'intimate bubbles of homeliness', for example, through sharing food (Vandevoordt, 2017, p. 606) or participating in social, cultural or religious rituals with other migrants (Nititham, 2017).

#### 5.3.5 Summary

The multiple ways in which precarity limits the structural possibilities of home are outlined in this section. We pay particular attention to migrants in situations of extreme vulnerability and protracted displacement, highlighting the impossibilities of home in such circumstances. However, the literature also reveals the complexities of home for migrants in different types of precarious situations, and reveals that circumstances that reduce spatial security (both in terms of legal status as well as wider possibilities of place belonging) provide an ongoing challenge to efforts to making home in migration. Despite these challenges, migrants continue to work to make home in different ways.

In light of this, we point to Brun and Fábos' (2015) proposal for a dynamic political understanding of home in migration which recognises that even in conditions of liminality (as opposed to 'limbo'), people continue to work to make home, improve their current living circumstances and imagine a better future home. They argue that binary 'here-there' or 'past-future' understandings of home fail to capture the realities of what home comes to mean in such conditions and instead it is important to recognise home as existing between places and in the active doing of home. This point is reinforced by others, such as Pérez-Murcia (2019) and Boccagni (2022), who argue for the need to conceptualise home in displacement in nonessentialised ways.

## 5.4 Conclusion

Understanding home and migration in relation to migrants' status draws the attention to a politics of home and migration that informs one's conditions of mobility and settlement. The underlying forces, including policies and political acts, that enable some groups and disable others from entering, remaining, leaving or even fleeing a context mean that governments and nation-states play an important role in the structural im/possibilities of home-making. Immigration status impacts migrants' access to housing, labour market, and ability to create a sense of belonging. Bloch et al. (2014) argue that young undocumented migrants are very likely to stay in their countries of residence for good or a long time. This means that a form of 'legal precariousness', if they are children or young migrants, can add to their limbo situation and defer their feeling at home and belonging. Others transition into precarity from some form of 'legal' status such as visitor visas and student visas and overstay and breach the terms of their entry (Koser, 2005, cited in Bloch et al., 2014). What is noteworthy though is that conditions of privilege and precarity can change, as life circumstances change.

A relational understanding of privilege and precarity will help then to address these nuances. For example, how do migrants move from one position to the other? How do their privileged positions turn into precarity and vice versa? Drawing boundaries between the two is difficult as social locations of migrants are not static, as policies change, locations and onward mobilities happen and aspirations and life situations alter all the time and, perhaps most importantly, due to factors such as changing technologies and labour markets and climate change, new forms of migratory journeys and settlement strategies take place.

As such, people who, at some point in time, experience privilege in one aspects of their lives but not in others, might see their position as precarious. Fathi (2017) in her research with Iranian migrant doctors and dentists in the UK, shows that before registering as medics with the British medical and dental councils, these skilled migrants experience extreme precariousness and they may resort to work in nonrelated positions. However, once they pass that hurdle, become more privileged migrants, with many spatial rights and security, they use consumption practices to place themselves in their 'correct' places as middle-class citizens rather than migrants. In her research with Vietnamese migrants in East London, Wilkins (2019) notes that housing precarity for these migrants led to some form of more established conditions after decades. Some long-established migrants managed to buy accommodation that had been allocated to them through housing associations, whilst renting was the only viable option for more recent migrants (in the last decade) (Wilkins, 2019). We see here that the duration of residence in a country but also the era of entry are important factors that can lead to more security or precarity. Sometimes, renting is considered to provide more flexibility but as Wilkins (2019) argues, drawing on Parutis (2011), renting can also be an acceptable option for those who see their time in the country as temporary and who plan to make their future homes elsewhere.

Research by Soudy (2017) with middle-class Egyptian migrants in Qatar and with 1.5 and 2nd generation Egyptian-Americans in the US sheds further light on the complexities of structural im/possibilities of home. According to Soudy (2017), for Egyptian-Americans in the US, their permanent residency status was a key factor in their seeing the US as their home due to the sense of security it bestowed. In comparison, Egyptians in Qatar, though they felt an everyday sense of home there, did not feel it was their long-term home, which Soudy (2017) attributes to their second-class citizenship status there. She argues that citizenship regimes shape

migrants' relationships to their host society in terms of their willingness to invest in that society and their feelings of belonging. However, the research also reveals that Egyptian-Americans experience feelings of outsider-ness in the US due to racism and Islamophobia. In other words, citizenship status intersects with racism to complicate privileged status in migration. Even though Soudy's (2017) participants are middle-class and largely professional, with permanent residency status in the US, their spatial security in the US is undermined by the prevalence of racism and Islamophobia there.

This chapter has considered a very important player in possibilities of home making. Migration status is the category that is defined by states, large institutions and socio-economic systems such as labour markets, which control what kind of capital migrants have access to and whether these forms of capital are transnationally accepted. Lewis et al. (2015) argue that precarity, which is an effect of not having access to those resources and capitals to which privileged migrants do, is associated with neoliberal globalised systems that value certain capitals over others. When considering privilege and precarity together we realise that states and their tools such as migration policies, borders and time limitations placed on entries and settlements make them the key players in structural possibilities of home. Furthermore, it is important to note that focusing only on policies on migration and settlement in relation to privilege and precarity also increases the risk of assuming methodological nationalism (Wimmer & Glick-Schiller, 2002). Although methodological nationalism, or the role of nation states in understanding migration and home-making experiences, is important, privilege and precarity are two concepts whose meanings are not bound only to this but capture the interrelationality of structural and personal elements that is so important in understanding home in migration.

#### References

- Ahmed, S. (1999). Home and away: Narratives of migration and estrangement. *International Journal of Cultural Studies*, 2(3), 329–347.
- Ahmed, S., Castañeda, C., Fortier, A. M., & Sheller, M. (Eds.). (2020 [2003]). Uprootings/ regroundings: Questions of home and migration. Routledge.
- Beeckmans, L., Gola, A., Singh, A., & Heynen, H. (Eds.). (2022). Making home (s) in displacement: Critical reflections on a spatial practice. Leuven University Press.
- Benson, M. (2019). Brexit and the classed politics of bordering: The British in France and European belongings. *Sociology*, *54*(3), 501–517.
- Benson, M., & O'Reilly, K. (2018). Lifestyle migration and colonial traces in Malaysia and Panama. Springer.
- Bloch, A., Sigona, N., & Zetter, R. (2014). Sans papiers: The social and economic lives of young undocumented migrants. Pluto Press.
- Boccagni, P. (2022). At home in the centre? Spatial appropriation and horizons of homemaking in reception facilities for asylum seekers. In *Making home(s) in displacement* (pp. 139–154). Leuven University Press.

- Boccagni, P., & Erdal, M. B. (2021). On the theoretical potential of 'remittance houses': Toward a research agenda across emigration contexts. *Journal of Ethnic and Migration Studies*, 47(5), 1066–1083.
- Boer, R. D. (2015). Liminal space in protracted exile: The meaning of place in Congolese refugees' narratives of home and belonging in Kampala. *Journal of Refugee Studies*, 28(4), 486–504.
- Breen, M. (2012). Privileged migration: American undergraduates, study abroad, academic tourism. Critical Arts, 26(1), 82–102.
- Brun, C. (2015). Home as a critical value: From shelter to home in Georgia. Refuge, 31(1), 43-54.
- Brun, C., & Fábos, A. (2015). Making homes in limbo? A conceptual framework. *Refuge*, 31(1), 5–17.
- Butcher, M. (2010). From 'fish out of water' to 'fitting in': The challenge of re-placing home in a mobile world. *Population, Space and Place, 16*(1), 23–36.
- Cai, X., & Su, X. (2021). Dwelling-in-travelling: Western expats and the making of temporary home in Guangzhou, China. Journal of Ethnic and Migration Studies, 47(12), 2815–2832.
- Clifford, J. (1997). *Routes: Travel and translation in the late twentieth century*. Harvard University Press.
- Croucher, S. (2012). Privileged mobility in an age of globality. Societies, 2(1), 1-13.
- Dawson, A., & Rapport, N. (Eds.). (2021). Migrants of identity: Perceptions of 'home' in a world of movement. Routledge.
- Dhaliwal, S., & Forkert, K. (2015). Deserving and undeserving migrants. Soundings, 61(13), 49-61.
- Duplan, K., & Cranston, S. (2023). Towards geographies of privileged migration: An intersectional perspective. *Progress in Human Geography*, 47(2), 333–347.
- Eastmond, M. (2006). Transnational returns and reconstruction in post-war Bosnia and Herzegovina. *International Migration*, 44, 141–166.
- Fathi, M. (2017). Intersectionality, class and migration: Narratives of Iranian women migrants in the UK. Palgrave Macmillan.
- Fathi, M. (2023). Constructing home through unhome: Narratives of resistance by an Iranian asylum seeker in Germany. *Social Sciences*, 12(1), 16.
- Fathi, M., & Ní Laoire, C. (2023). Urban home: Young male migrants constructing home in the city. *Journal of Ethnic and Migration Studies*, 49(3), 813–831.
- Gil Everaert, I. (2021). Inhabiting the meanwhile: Rebuilding home and restoring predictability in a space of waiting. *Journal of Ethnic and Migration Studies*, 47(19), 4327–4343.
- Green, P. (2023). Homes, home lives and precarious privilege: Older western residents and differing lifestyle mobilities in Ubud, Bali. Asian Studies Review, 47(2), 374–391.
- Grønseth, A. S., & Thorshaug, R. Ø. (2022). Struggling for home where home is not meant to be: A study of asylum seekers in reception centers in Norway. *Focaal*, 2022(92), 15–30.
- Howlett-Southgate, R. (2021). Life after asylum: Investigating the precarious geographies of home with refugees leaving direct provision in Cork, Ireland. Unpublished PhD thesis, University College Cork.
- Korac, M. (2009). *Remaking home: Reconstructing life, place and identity in Rome and Amsterdam.* Berghahn Books.
- Kox, M., & van Liempt, I. (2022). "I Have to start all over again". The role of institutional and personal arrival infrastructures in refugees' home-making processes in Amsterdam. *Comparative Population Studies*, 47, 165–184.
- Kreichauf, R. (2018). From forced migration to forced arrival: The campization of refugee accommodation in European cities. *Comparative Migration Studies*, 6(1), 1–22.
- Lewis, H., Dwyer, P., Hodkinson, S., & Waite, L. (2015). Hyper-precarious lives: Migrants, work and forced labour in the Global North. *Progress in Human Geography*, 39(5), 580–600.
- Lundström, C. (2014). White migrations: Gender, whiteness and privilege in transnational migration. Springer.
- Mand, K. (2010). 'I've got two houses. One in Bangladesh and one in London... Everybody has': Home, locality and belonging (s). *Childhood*, *17*(2), 273–287.

- Massa, A. (2022). "All we need is a home": Eviction, vulnerability, and the struggle for a home by migrants from the Horn of Africa in Rome. *Focaal*, 2022(92), 31–47.
- Menjívar, C. (2006). Liminal legality: Salvadoran and Guatemalan immigrants' lives in the United States. American Journal of Sociology, 111(4), 999–1037.
- Moreton-Robinson, A. (2003). I still call Australia home: Indigenous belonging and place in a white postcolonizing society. In Ahmed, et al. (Eds.), Uprootings/regroundings (pp. 23–40). Berg.
- Motasim, H., & Heynen, H. (2011). At home with displacement? Material culture as a site of resistance in Sudan. *Home Cultures*, 8(1), 43–69.
- Ndlovu, T. (2010). Where is my home? Rethinking person, family, ethnicity and home under increased transnational migration by Zimbabweans. *African Identities*, 8(2), 117–130.
- Neumark, D. (2013). Drawn to beauty: The practice of house-beautification as homemaking amongst the forcibly displaced. *Housing, Theory and Society, 30*(3), 237–261.
- Nititham, D. S. (2017). Making home in diasporic communities: Transnational belonging among Filipina migrants. Routledge.
- O'Reilly, Z. (2020). The in-between spaces of asylum and migration: A participatory visual approach. Palgrave Macmillan.
- Parutis, V. (2011). "Home" for now or "home" for good? East European migrants' experiences of accommodation in London. *Home Cultures*, 8(3), 265–296.
- Pease, B. (2010). Undoing privilege: Unearned advantage in a divided world. Zed Books.
- Pérez Murcia, L. E. (2019). 'The sweet memories of home have gone': Displaced people searching for home in a liminal space. *Journal of Ethnic and Migration Studies*, 45(9), 1515–1531.
- Povrzanović Frykman, M. (2002). Homeland lost and gained: Croatian diaspora and refugees in Sweden. In N. Al-Ali & K. Koser (Eds.), New approaches to migration? Transnational communities and the transformation of home (pp. 118–137). Routledge.
- Prazeres, L. (2018). At home in the city: Everyday practices and distinction in international student mobility. Social & Cultural Geography, 19(7), 914–934.
- Robertson, S., & Roberts, R. (Eds.). (2022). *Rethinking privilege and social mobility in middleclass migration: Migrants 'in-between'*. Routledge.
- Silva, K. (2009). Oh, give me a home: Diasporic longings of home and belonging. *Social Identities*, 15(5), 693–706.
- Slingenberg, L., & Bonneau, L. (2017). (In)formal migrant settlements and right to respect for a home. European Journal of Migration and Law, 19(4), 335–369.
- Smith, M. (2004). Warehousing refugees. World Refugee Survey, 38(1), 38-56.
- Soudy, N. (2017). Home and belonging: A comparative study of 1.5 and second-generation Egyptian 'expatriates' in Qatar and 'immigrants' in the US. *Journal of Ethnic and Migration Studies*, *43*(9), 1544–1561.
- Thorshaug, R. Ø., & Brun, C. (2019). Temporal injustice and re-orientations in asylum reception centres in Norway: Towards critical geographies of architecture in the institution. *Fennia*, 197(2), 232–248.
- Trapp, M. (2015). Already in America: Transnational homemaking among Liberian refugees. *Refuge*, *31*(1), 31–41.
- Ullah, A. A., Hasan, N. H., Mohamad, S. M., & Chattoraj, D. (2021). Privileged migrants and their sense of belonging: Insider or outsider? *Asian Journal of Social Science*, 49(3), 161–169.
- UNHCR. (2017). Urban refugees. Retrieved January 7, 2017, from http://www.unhcr.org/urban-refugees.html
- Van der Horst, H. (2004). Living in a reception Centre: The search for home in an institutional setting. *Housing, Theory and Society, 21*(1), 36–46.
- Van Liempt, I. (2023). Book review: Beeckmans, L., Gola, A., Singh, A., & Heynen, H. Making home(s) in displacement: Critical reflections on a spatial oractice. *Urban Studies*, 60(5), 990–992.
- Van Liempt, I., & Miellet, S. (2021). Being far away from what you need: The impact of dispersal on resettled refugees' homemaking and place attachment in small to medium-sized towns in the Netherlands. *Journal of Ethnic and Migration Studies*, 47(11), 2377–2395.

- Van Liempt, I., & Staring, R. (2021). Homemaking and places of restoration: Belonging within and beyond places assigned to Syrian refugees in the Netherlands. *Geographical Review*, 111(2), 308–326.
- Van Liempt, I., Van Aalst, I., & Schwanen, T. (2015). Introduction: Geographies of the urban night. Urban Studies, 52(3), 407–421.
- Vandevoordt, R. (2017). The politics of food and hospitality: How Syrian refugees in Belgium create a home in hostile environments. *Journal of Refugee Studies*, 30(4), 605–621.
- Walsh, K. (2006) British expatriate belongings: Mobile homes and transnational homing. *Home Cultures*, 3(2), 123–144.
- Walsh, K. (2011). Migrant masculinities and domestic space: British home-making practices in Dubai. *Transactions*, 36(4), 516–529.
- White, A. (2012). 'Every Wednesday I am happy': Childhoods in an Irish asylum centre. *Population, Space and Place, 18*(3), 314–326.
- Wiles, J. (2008). Sense of home in a transnational social space: New Zealanders in London. *Global Networks*, 8(1), 116–137.
- Wilkins, A. (2019). *Migration, work and homemaking in the city: Dwelling and belonging among Vietnamese communities in London.* Routledge.
- Wimark, T. (2021). Homemaking and perpetual liminality among queer refugees. Social & Cultural Geography, 22(5), 647–665.
- Wimmer, A., & Glick Schiller, N. (2002). Methodological nationalism and beyond: Nation–state building, migration and the social sciences. *Global Networks*, 2(4), 301–334.
- Zhang, H., & Su, X. (2020). Lifestyle migration and the (un) making of ideal home. *Geoforum*, *115*, 111–119.

**Open Access** This chapter is licensed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license and indicate if changes were made.

The images or other third party material in this chapter are included in the chapter's Creative Commons license, unless indicated otherwise in a credit line to the material. If material is not included in the chapter's Creative Commons license and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder.



# Chapter 6 Conclusion



# 6.1 Introduction

At some point in life, people call a place, or multiple places, home. In fact, we may inhabit many types of spaces, but they are not home until we ascribe meanings of home to them. What makes a space a home, or home-like, is the emotional meaning it carries, which is bound up with the social relations and material things that comprise it. For migrants who experience departure from one home and the process of making another in a different country, having crossed borders, home becomes a complicated matter. In this book we have tried to offer an opening into what home in migration means, considering the different identity categories and positions one may have and the different types of spaces that can become home/s, or like home/s, in migration. We discuss how gender, age, generation, objects, class, race and migration status can have long-lasting impacts on how home is understood and experienced.

The aim of this book was to offer firstly an understanding of home in migration from the perspective of a micro and macro level of analysis, by looking at how 'structural im/possibilities' impede or facilitate the processes of homing and homemaking. In other words, we look at home from a structural perspective, taking account of the role of migration regimes, gendered politics of home, age and the life-course, material cultures and migration status, in how people can think of a place as a home or not. Secondly, the book seeks to do this from an intersectional perspective by not simply looking at these questions as if structural aspects of home were experienced in the same way by all migrants, but by organising the analysis around different social positionalities in relation to structural im/possibilities of home. Lastly, this small volume aimed to further expand the ways in which each area of research on home in migration could be expanded and how boundaries could be pushed further in order to provide a more sophisticated perspective on home in migration, home-making and homing.

## 6.2 Home in Migration at a Glance

The first chapter outlines a theoretical framework of structural im/possibilities of home in migration, referring to the social and political structures that frame homemaking by individuals. These structures incorporate migration regimes and intersectional social relations. Migration policies, political decisions, systems of bordering and population control and politics of belonging, all of which bestow settlement rights on some and push out those who are deemed outsiders, frame the structural possibilities of home through different types of migration regimes. These increasingly volatile and changeable migration regimes together provide unstable and temporal larger infrastructures that make it possible for some migrants to make a home or to feel at home while making it impossible for others. We argue, throughout this book, that what makes having a home, making a home and feeling at home possible is not simply an individual or personal matter. It is always contextual and embedded within a set of power relations that inform and facilitate such possibilities and impossibilities. For example, the deportation of hundreds of asylum-seekers from European countries by force each month to a variety of Global South countries, where their return poses serious threats to their lives, is an example of how structurally making a home becomes impossible. Another example is the extent to which a migrant student can make a decision to remain and make a home after their study is completed, navigating their residency journey through a myriad of bureaucratic processes to facilitate the possibilities of making home. We argue that home as a place of safety, security and control needs to be about both legal rights and feelings of belonging, the combination of which can be called 'spatial security', an essential element of feeling at home in migration. The intention here is to emphasise how certain groups of migrants have the capital in its different forms, and to different degrees, that makes home-making in migration possible, whilst others do not. Further, different groups of migrants may have access to different types of homemaking capital and these capitals are not just contingent on migration status but are racialised, gendered, classed, abled and aged (among others). This framework as such emphasises how the intersectionality of home combines with transnational migration regimes to shape im/possibilities of home. The relationship between these, although manifest in most migration scholarship, needs to be foregrounded more in analysis of home in migration.

Chapter 2 addresses the importance of gender in understanding home in migration. Drawing on feminist scholarship on home, the chapter shows how the gendered construction of space, based on economic, racial and patriarchal power relations, is essential to understanding the meanings of home in migration. The chapter shows how the association of femininity with private spaces of home can be both challenged and reinforced by women's mobilities and by migrant bordercrossings. Whilst much of the existing literature looks at experiences of women migrants in relation to domestic spaces of home, the chapter also shines a light on emerging research on migrant masculinities and home. However, although studies point to new and alternative femininities and masculinities, there is much evidence that traditional gender roles at home before, during and after migration, still persist. The next chapter, Chap. 3, highlights the significance of an age and life-course perspective in understanding of migrant home. In particular more attention needs to be paid to children's perspectives on home in migration. Through a focus on three distinct life-course stages of childhood, young adulthood and older age, the chapter provides an in-depth discussion of the ways in which the structural im/possibilities of home in migration present different types of challenges and opportunities for migrants at different life-course stages. This age-related focus also illuminates the ways in which home is integrally about family and other social relations, even as these can be ruptured and created through migration. The role of age in migrant homing then is a rather complex matter, being bound up with temporalities, biographies and relationalities, and as such merits greater attention.

Chapter 4 turns to materialities in migrant home by analysing objects in migrants' lives as not only facilitators of their home-making practices but also as tools for invigorating their memories and reinstating their identities. In order to better understand the role of objects in migration, the chapter proposes a framework for analysing the scholarship on materialities of migration by distinguishing between objects of memory, objects of use and objects of identity. This framework is useful in capturing the nuances in the growing scholarship in this area, particularly in terms of avoiding the temptation to ascribe meaning 'beyond' their immediate use to all kinds of objects in migrants' homes. In this way it is important not to construct migrants' lives as extraordinary and unusual in relation to home-making. After all, home-making is about living what one views as a normal life, where one is happy and feels belonging, but objects can play a key role in this process and can help in creating a nest where people feel comfortable with their surroundings.

Finally, Chap. 5 focuses on migration status, class and race as interconnected factors shaping the structural im/possibilities of home in migration. Nation-states are still key players in creating unequal positions for migrants by encouraging, facilitating, limiting and banning certain groups, nationalities, and communities. Systems of bordering (such as immigration and citizenship policies, and how these are classed and racialised) position some who move as privileged while others who move do so in precarious conditions where making a home becomes impossible. For example, shelters, refugee camps, and communal living arrangements for temporary migrants for short or long periods become places of unhome that add to the feelings of being in limbo. Examining how migrants make home in circumstances of privilege and precarity illuminates the complex ways in which these circumstances are bound up with migrant status, class and race.

The field of migration studies has expanded rapidly in the last two decades (Levy et al., 2020) and it is important to critically analyse how knowledge has been, and continues to be, produced within this field. The sub-field of 'home and migration' is a rich one, with potential for further development in the future, and hence it is important to take stock of where we are now in terms of what is known and how it is known. We hope this Short Reader helps to achieve this goal by taking a journey through the current social-scientific knowledge on home in migration. We look forward to seeing how this field emerges as one of the key sub-disciplines of migration studies.

# Reference

108

Levy, N., Pisarevskaya, A., & Scholten, P. (2020). Between fragmentation and institutionalisation: The rise of migration studies as a research field. *Comparative Migration Studies*, 8(1), 1–24.

**Open Access** This chapter is licensed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license and indicate if changes were made.

The images or other third party material in this chapter are included in the chapter's Creative Commons license, unless indicated otherwise in a credit line to the material. If material is not included in the chapter's Creative Commons license and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder.

